A Study of the Relationship between Parental Involvement and Mental Health of College Students

by

Ruthanna Blake Payne

A dissertation submitted to the Graduate Faculty of
Auburn University
in partial fulfillment of the
requirements for the Degree of
Doctor of Philosophy

Auburn, Alabama May 14, 2010

Keywords: parental involvement, mental health, college students

Approved by

James E. Witte, Chair, Associate Professor of Educational Foundations, Leadership and Technology

David C. DiRamio, Assistant Professor of Educational Foundations, Leadership and Technology

David Shannon, Professor of Educational Foundations, Leadership and Technology

Abstract

The purpose of this study was to describe the characteristics of parental involvement and mental health in a sample of traditionally aged college students and investigate the variance parental involvement predicts in mental health. Five hundred and eighty-eight freshmen at a large research university responded to a 97 question survey. Parental involvement was broken into parental involvement in college choice, social involvement, academic involvement, student satisfaction with parental involvement, frequency of contact, and frequency of visits. Mental health was constructed of emotional well-being, psychological well-being, and social well-being. Demographic information was also collected. Independent sample t-tests, one-way analyses of variance, multiple analyses of variance, and backward multiple regression analyses were performed to analyze the results.

The average parental involvement score for the entire sample was 59.96 on a 100 point scale. The most parental involvement occurred in students' social involvement followed by college choice then academic involvement. The average satisfaction score was 70.16 on a 100 point scale. Participants communicated with their parents an average of 9.99 times per week and visited an average of 7.06 times per semester. The average mental health score was 71.10 on a 100 point scale. In the sample, 59.7% were moderately mentally healthy followed by 30.4% who were flourishing and 9.9% who were languishing.

A multiple regression analysis revealed that 9% of the variance in mental health was accounted for by parental involvement for the entire sample and 14% for females. Implications of the study include: parental involvement does impact mental health status, especially in regards to students' satisfaction with their parents' involvement. Higher education institutions need to continue to find more ways to purposefully include parents in students' social and academic lives at college and create opportunities for parents and their students to communicate openly about their changing relationship.

Acknowledgements

This dissertation is dedicated to my daughter Annie who is my inspiration to always strive for success and my mother Barbara who encouraged me throughout all the stages of my education.

I would like to thank Dr. Jim Witte, my advisor and friend, who stood by me from start to finish in this process. Additionally, I thank Dr. David DiRamio, Dr. David Shannon and Dr. Maria Witte for their support and feedback in completing this research. I also express my appreciation to Dr. Ginger Ambler, Dr. Corey Keyes, and Bryan Oliver whose original research and support provided the foundation for this research.

I would like to thank all my friends and colleagues in the Educational Support

Services Department at Auburn University. I am especially grateful to Sarah Fitch and

Jane Turnbull Humphries for their continuous encouragement without which I could not have completed this adventure.

Table of Contents

| Abstra | ict | ii |
|---------|--|------|
| Ackno | owledgements | iv |
| List of | Tables | viii |
| I. | Introduction | 1 |
| | Background | 1 |
| | Purpose | 2 |
| | Statement of the Problem | 4 |
| | Significance of the Study | 4 |
| | Research Questions | 6 |
| | Limitations/Delimitations of the Study | 6 |
| | Definition of Terms. | 7 |
| | Organization of Study | 8 |
| II. | Review of the Literature | 9 |
| | Introduction | 9 |
| | Parental Influence throughout the Stages of Development | 9 |
| | Parental Influence in Education | 16 |
| | Parental Involvement in Elementary and Secondary Education | 20 |
| | Parental Involvement in Higher Education | 22 |
| | Parental Influence on College Student Mental Health | 33 |

| | Addressing Mental Health Concerns on the College Campus | 39 |
|--|---|-----|
| | Summary | 46 |
| III. | Methods | 47 |
| | Introduction | 47 |
| | Research Design | 49 |
| | Population | 50 |
| | Procedures | 53 |
| | Measures | 54 |
| | Analyses | 63 |
| | Summary | 64 |
| IV. | Results | 65 |
| | Introduction | 65 |
| | Descriptive Statistics | 65 |
| | Research Question 1 | 68 |
| | Research Question 2 | 79 |
| | Research Question 3 | 83 |
| | Research Question 4 | 86 |
| | Summary | 91 |
| V. | Conclusions and Recommendations | 95 |
| | Conclusions | 95 |
| | Recommendations | 99 |
| References | | 102 |
| Appendix A: Institutional Review Board Approval Letter | | 136 |

| Appendix B: | Email Request to Course Instructors | 138 |
|-------------|---------------------------------------|-----|
| Appendix C: | Administrator Script | 140 |
| Appendix D: | Alternative Questions | 142 |
| Appendix E: | Information Letter | 144 |
| Appendix F: | Parental Permission/Child Assent Form | 146 |
| Appendix G: | Survey | 149 |
| Appendix H: | Referral List | 156 |
| Appendix I: | Scoring Information for Keyes' Scales | 157 |
| Appendix J: | Letters of Approval | 160 |

List of Tables

| Table 1 | Frequencies and Percentages for Demographic Variables | 52 |
|----------|---|----|
| Table 2 | Psychological Well-Being Items | 56 |
| Table 3 | Social Well-Being Items | 57 |
| Table 4 | Student Satisfaction with Parental Involvement Items | 61 |
| Table 5 | Frequencies and Percentages for Levels of Parental Involvement | 66 |
| Table 6 | Means and Standard Deviations for Parental Involvement Variables | 66 |
| Table 7 | Frequencies and Percentages for Mental Health Categories | 67 |
| Table 8 | Means and Standard Deviations for Mental Health Variables | 67 |
| Table 9 | Mental Health and Parental Involvement Continuous Variables | 69 |
| Table 10 | Mental Health and Parental Involvement Continuous Variables by Gender | 71 |
| Table 11 | Mental Health and Parental Involvement Continuous Variables by Ethnicity | 72 |
| Table 12 | Mental Health and Parental Involvement Continuous Variables by Distance from Home | 74 |
| Table 13 | Parental Involvement Continuous Variables by First Generation Experience | 75 |
| Table 14 | Mental Health and Parental Involvement Continuous Variables by Mental Health Category | 77 |
| Table 15 | Multiple Analysis of Variance for Parental Involvement Continuous Variables by Mental Health Category | 78 |

| Table 16 | Cross Tabulations of Mental Health Category and Independent Variables | 80 |
|----------|--|----|
| Table 17 | Cross Tabulations of Level of Parental Involvement and Independent Variables | 83 |
| Table 18 | Summary of Backward Multiple Regression Analysis | 87 |
| Table 19 | Model Summary for Backward Multiple Regression Analysis | 88 |
| Table 20 | Summary of Backward Multiple Regression Analysis in Females | 89 |
| Table 21 | Model Summary for Backward Multiple Regression Analysis in Females | 90 |
| Table 22 | Summary of Backward Multiple Regression Analysis in Males | 91 |
| Table 23 | Model Summary for Backward Multiple Regression Analysis in Males | 92 |

I. Introduction

Background

Student mental health is a current priority in higher education (Ambler, 2007; Astin, 1993; Kadison & DiGeronimo, 2004; Snyder, 2007), as there has been an increase in the number of students coming to college with mental health concerns, the number of students reporting to student counseling centers, and the severity of mental illness students are facing (American Psychiatric Association, [APA], 2000; Arehart-Treichel, 2002; Benton, Robertson, Tseng, Newton, & Benton, 2003; Riba, 2004). Previous research addressing student mental health has focused on poor or incomplete mental health, referred to by Keyes (2002, 2003, 2005, 2007) as languishing (Ambler, 2007). While this information is critical, Keyes, Amber, and others have called attention to an additional need for more research on positive or complete student mental health, Keyes' flourishing, to build a comprehensive understanding of contributing environmental variables (Ambler, 2007; Keyes, 2002, 2003, 2005, 2007; Keyes & Haidt, 2003; Keyes & Lopez, 2002).

An increase in the amount of involvement parents have in their college students' lives has also recently been observed on college campuses (Savage, 2007). Parents are advising their students on where to attend, helping them with applications, providing financial support, and engaging in continuous communication with their students once they begin school (Coburn, 2003, 2006; Colavecchio-Van Sickler, 2006; Daniel, Evans,

& Scott, 2001; Lange & Stone, 2001; Lum, 2006; McGinty, 2002; Merriman, 2007; Scott & Daniel, 2001, 2007; Strage & Brandt, 1999). A secure relationship with parents has been suggested as beneficial for students going through the difficult transition from high school to college (Baxter Magolda, 1999; Conneely, Good, & Perryman, 1999; Daniel et al., 2001; Elkind, 1994; Savage, 2007), but parental over-involvement, including parents engaging in constant contact with student and administrators, making academic decisions for students, and taking student failure personally, can have lasting negative effects for students that are not at this time fully understood (Astin, Tsui, & Avalos, 1996; College Board, 2005; Daniel et al., 2001; Epstein & Sanders, 2002; Hill & Taylor, 2004; Padilla, Trevino, Gonzalez, & Trevison, 1997; Pascerelli & Terenzini, 2005; Steinberg & Silk, 2002; Stone, 2006; Strage & Brandt, 1999). Previous research on parental involvement has focused predominately on kindergarten through high school and hence there is a lack of research on parental involvement in college.

Purpose

Positive parental involvement in kindergarten through 12th grade education has been linked with a number of outcomes when compared to students with too much or too little parental involvement. These include students making higher grades (Christenson, 1999; Desimore, 1999; Dornbusch, Ritter, Liederman, Roberts, & Fraleigh, 1985; Fehrmann, Keith, & Reimer, 1987; Lee, 1993; Martini, 1995; Muller, 1993; Perna & Titus, 2005; Steinberg, Lamborn, Dornbusch, & Darling, 1992; Sui-Chu & Willms, 1996; Zick, Bryant, & Osterbacka, 2001), choosing positive peer groups (Brown, Mounts, Lamborn, & Steinberg, 1993; Falbo, Lein, & Amador, 2001), experiencing higher levels

of extracurricular achievement (Linver & Silverberg, 1997; Paulson, 1994), being less likely to drop out (Horn & West, 1992; McNeal, 1999; Rumberger, Ghatak, Poulos, Ritter, & Dornbusch, 1990; Sy, Rowley, & Schlenberg, 2007), and adjusting more successfully to transition (Lee, 1993; Reynolds, 1992; Wong & Hughes, 2006; Zick et al., 2001).

While appropriate levels of parental involvement may contribute to these positive outcomes, parental over-involvement in kindergarten through 12th grade education has been linked with delayed identity development, poor academic performance, increased conflict between parents and students, and increased experiences of internalizing disorder symptoms (Barber, Maughen, & Olsen, 2005; Nucci, 2001; Smetana, Crean, & Campion-Barr, 2005; Steinberg, Dornbusch, & Brown, 1992). Previous studies also show that over-involvement is most likely to occur during times of academic transition, such as elementary to middle school and middle school to junior high (Barber et al., 2005; Nucci, 2001; Smetana, Crean, & Campione-Barr, 2005; Steinberg, Lamborn, Dornbusch et al., 1992). The transition from high school to college is an especially challenging transition for many students, and the purpose of this study is to investigate if parental involvement again increases during the college transition and if parental involvement is therefore contributing to mental health concerns on college campuses.

This study focused on describing parental involvement that contributes to college student mental health, as defined by Keyes' (2002, 2003, 2005, 2007) continuum of mental health, which ranges from flourishing to moderate mental health to languishing. Traditionally aged college freshmen were the focus of the study because parental involvement has been found to have the strongest influence on younger students and the

impact of the transition is the strongest during the first year (Strage & Brandt, 1999; Dyson & Renk, 2006). Using Ambler's (2007) study of student engagement and mental health as a model, the purpose of the current study was to use Keyes' (2002) Mental Health Scales of Subjective Well-being and Oliver's (n.d.) Survey of Parental Involvement to explore students' perceived status of well-being and levels of parental involvement in their lives. Impact of gender, ethnicity, distance from home, and whether or not the student was a first generation college student was also investigated.

Statement of the Problem

Ambler (2007) revealed that only 17% of college students in her study were flourishing. Her study also revealed that a supportive campus environment was the most significant factor contributing to student well-being. Parents have become an integral part of the support system on college campuses. Colleges have an obligation to create an environment that promotes holistic student development and well-being (American College Personnel Association [ACPA], 1994, 1996), and understanding the relationship between parental involvement and student mental health is critical to fulfilling this obligation (Coburn, 2006; Daniel et al., 2001; Merriman, 2007; Scott & Daniel, 2001).

Significance of the Study

Higher numbers of students are attending college today yet fewer students are graduating (United States Department of Education [USDE], 1995; Strage & Brandt, 1999). A survey conducted by the American College Health Association [ACHA] (2006) reported 92% of students felt overwhelmed, 44% indicated experiencing debilitating

depression, and approximately 1% attempted suicide during the previous year (Kennedy, 2007). While some students flourish on their own, others need administrators to step in and help them make necessary changes in their environments, including their human aggregate environment, which parents dominate along with peers (Pascarella & Terenzni, 1980; Strange & Banning, 2001).

Parents have been told by the media and educators since their children were in kindergarten that it is important for them to take an active role in their children's education (Bird, 2006; Coleman, 1998; Lin, 2001 a, 2001 b; Perna & Titus, 2005; Sy et al., 2007; Steinberg, Lamborn, et al., 1992; Sui-Chu & Willms, 1996; Wong & Hughes, 2006; Zick et al., 2001). Technology makes this possible, even from a distance (Lange & Stone, 2000; Merriman, 2007; Savage, 2007; Strage & Brandt, 1999). Also, parents are concerned about campus safety in light of recent tragedies and about health risks such as smoking, alcohol, sleeping habits, and nutrition (Bylund, Imes, & Baxter, 2005; Mansfield & Warwick, 2005; Shellenbarger, 2006). Finally, higher education is a significant investment, and in today's society of consumerism, parents want to get their money's worth (Coburn, 2006; Conneely et al., 2001; Daniel et al., 2001; Lange & Stone, 2001; Merriman, 2007; Savage, 2007; Shellenbarger, 2006). Parents have pertinent reason to be involved in their students' lives. However, more research is needed to define the point at which parents become over involved and the impact this over involvement has on college students' mental well being (Daniel et al., 2001; Forbes, 2001; Padilla et al., 1997; Pascarella & Terenzini, 1998; Tinto, 1993; Strage & Brandt, 1999).

Research Questions

The following research questions were addressed in this study: 1) What are the characteristics of parental involvement and mental health experienced by a sample of traditionally aged undergraduate students, 2) What relationship does mental health category have with gender, ethnicity, distance from home, experience of being a first generation college student, and level of parental involvement in a sample of traditionally aged undergraduate students, 3) What relationship does level of parental involvement have with gender, ethnicity, distance from home, and experience of being a first generation college student, and 4) How does parental involvement predict variability in mental health of traditionally aged undergraduate students?

Limitations/Delimitations of the Study

Limitations of the research were:

- Students self reported information and may have wished to represent themselves
 or their parents more positively.
- 2. The study was limited to freshmen at a large research university in Alabama, making it difficult to generalize to institutions in other states or of a different size.
- 3. In the state of Alabama, students must be 19 years of age or older to be considered an adult. Those older than 19 years of age must have parental consent to complete survey research. Only a minimal number of 18 year olds participated in the study.
- 4. The Survey of Parental Involvement is a new instrument in need of more validation.

Delimitations of the research were:

- 1. Parents were not surveyed in this study. Their personal perception of their involvement may be different from their students.
- Parental involvement was not explored for upperclassmen students. Only
 freshmen taking a freshman orientation class were invited to participate in the
 study.

Definition of Terms

The following terms were used as defined below:

- Student Mental Health: Keye's (2002) definition of the mental health continuum was used in this study, differentiating students as flourishing, experiencing high levels of emotional, psychological, and social well-being; languishing, experiencing low levels of the three domains of well-being; and moderately mentally healthy, falling somewhere in between.
- Parenting: Philosophies and strategies used in rearing children from infancy through adolescence in regards to the three domains defined by social domain theory which are conventional, moral, and psychological (Nucci, 2001).
- Parental Involvement: Parents' engagement in their students' education. At the kindergarten through 12th grade level, asking children about what they are learning; helping with homework; creating an environment that is conducive to learning in the home; introducing children to community resources; communicating with teachers; and participating in special events and organizations such as parent teacher associations and parents' day are considered

as parental involvement (Morrow, 1989; Stone, 2006; Sy et al., 2007). At the college level, assisting students in the college choice process; communicating with students via phone or email; advising students about social and academic involvement; visiting students at college; contacting faculty and administrators; and participating in special events such a parents' weekend all constitute as parental involvement (College Board, 2005; Conneely et al., 2001; Oliver, n.d.; Savage, 2007).

Organization of Study

Chapter I presents the background, purpose of the study, statement of the problem, significance of the study, research questions, limitations of the study, and definition of terms.

Chapter II reviews the literature related to this study. Parenting, parental involvement in elementary through secondary education, parental involvement in postsecondary education, and student mental health are addressed.

Chapter III addresses procedural details undertaken in this study including research design, population, data collection, and statistical analysis of data.

Chapter IV presents an interpretation of the data and findings.

Chapter V provides conclusions and recommendations for further research.

II. Review of Literature

Introduction

This study examined the relationship between parental involvement and mental health in traditionally aged college students. Chapter two provides a review of the literature regarding these constructs and previous research investigating the nature of their relationship. The following information is covered: (a) parental influence throughout the stages of development, (b) parental influence in education, (c) parental influence in elementary and secondary education, (d) parental influence in higher education, (e) parental influence on college student mental health, and (f) addressing mental health on the college campus.

Parental Influence throughout the Stages of Development

Students come to college at various stages of development and with varying dispositions, all of which are attributed in part to prior experiences during infancy through adolescence. While parents are the dominant authority during infancy and childhood, peers join the circle of influence in adolescence (Amato, 1994; Barnett, Kibria, Baruch & Pleck, 1991; Bumpass & Aquilino, 1994; Moore & Zaff, 2002; Shaw, Krause, Chatters, Connell & Ingersoll-Dayton, 2004; Umberson, 1992). Adolescents spend more time with peers, but parents still remain extremely influential (Astin, 1993;

Baumrind, 1991; Erikson, 1959; Moore & Zaff, 2002; Steinberg & Silverberg, 1986).

According to the National Longitudinal Survey of Youth 1997 cohort (Moore, et al., 2004), in a cohort of more than 9,000 adolescents 84% reported thinking highly of their mother and 81% reported thinking highly of their father (Moore et al., 2004, p. 1). Other studies have confirmed that positive parent-teen relationships lead to fewer problem behaviors (Bahr, Maughan, Marcos, & Li, 1998; Bearman et al., 1997; Blum & Rinehart, 1997; Coombs, Paulson, & Richardson, 1991; Hundleby & Mercer, 1987; Miller, 1998; Whitbeck, Hoyt, Miller, & Kao, 1992) and holistic well-being in adolescents (Borkowsky, Ramey, & Bristol-Power, 2002; Hair, Moore, & Garrett, 2004; Barber & Erickson, 2001; Hair, Jager, & Garrett, 2002; Engels, Finkenauer, Meeus, & Dekovic, 2001; Franz, McClelland, & Weinberger, 1991; Kerns & Stevens, 1996; Zahn-Waxler & Smith, 1992). Research confirms that parental influence is critical throughout the early stages of development.

A New Developmental Stage

Today's parents are staying more involved in their children's lives for longer during a new developmental stage referred to as emerging adulthood (Andom, 2007; Arnett, 2000, 2004) or extended adolescence (Rudolf, 1994). Some theorists now consider college to represent a new developmental stage, basing the stages on life accomplishments rather than age (Daniel et al., 2001; Elkind, 1994). According to Arnett(2000) approximately 60% of college students enter college straight from high school, transitioning directly into what Rudolf (1994) calls an extended moratorium. During this time they are free from parental control but still dependent on parents

financially and in other ways (Andom, 2007; Wetherill, & Fromme, 2007; Wetherill, Neal, & Fromme, 2008).

This period was previously thought of as the beginning stage of adulthood or young adulthood, but researchers have begun to believe that while students are still in the education system full time this period is qualitatively different (Andom, 2007; Daniel et al., 2001; Elkind, 1994). Andom (2007) reported that 392 participants in his study defined reaching adulthood as measured by achieving financial independence, holding a steady career, moving out of the family home, and demonstrating responsible behavior. Fewer than 20% of those participating in the study believed that college students met the qualifications of adulthood (Andom, 2007).

Instead, this time between the ages of 18 and 28 is viewed as a stepping stone into adulthood when individuals can experience freedom and take time to decide the direction they want to pursue in their lives (Rudolf, 1994). Changes in social norms including people getting married later, having children later, and the social acceptance of job experimentation during these years have played a role in the emergence of this stage. Additionally, parents' ability financially and willingness to continue supporting their children allow this stage to take place (Andom, 2007; Arnett, 2004; Rudolf, 1994). Parenting Theory and Practice

Regardless of the stage, it is necessary to be knowledgeable about parenting theory and practice in order to understand the parent-child relationship and its impact on children's development. Several different theories of parenting have emerged from different disciplines. This section outlines the prominent theories recognized across different areas of study and practice.

Precision Parenting

Several different theories have evolved to define effective parenting practices.

One of the most well-known is authoritative parenting or what Mason, Cauce, Gonzales, and Higaga (1996) refer to as precision parenting. Precision parenting requires the appropriate balance between responsiveness, defined as the "extent to which parents foster individuality and self-assertion by being attuned, supportive, and acquiescent to children's requests," and demandingness, defined as "claims parents make on children to become integrated into society by behavior regulation, direct confrontation, maturity demands, and supervision of children's activities" (Baumrind, 2005, p. 61). Successfully finding this balance leads to autonomy development and general well-being in adolescents (Buamrind, 1991; Steinberg & Silk, 2002).

Authoritative parenting was originally identified by Baumrind (1967) through a study conducted on parent attachment behaviors. In her research observing parent-child and parent-adolescent interaction, she discovered that parenting fell into one of three different styles: authoritative, authoritarian, or permissive (Baumrind, 1967, 1971, 1999, 2005). Authoritative parenting is characterized by a balance between responsiveness and demandingness, in other words implementing boundaries while also being supportive. Passive parenting on the other hand favors responsiveness while setting minimal to no boundaries. Finally, authoritarian parenting asserts too much demandingness with limited support. Later, Baumrind (1971, 1991, 2005) added a fourth parenting style, disengaged, to depict parenting that incorporates neither demandingness nor responsiveness. Baumrind (1967, 1971, 1999, 2005) discovered the type of parenting

style practiced correlated with children and adolescents' levels of adjustment, socialization, and academic ability (Barber, 1996; Steinberg & Silk, 2002).

Social Domain Theory

While parents tend to utilize one of Baumrind's parenting styles the majority of the time (Baumrind, 1967, 1971, 1999, 2005), parenting is a complex endeavor that requires different levels of parental control in different situations (Turiel, 2005).

According to the social domain theory, there are three different domains in which parents exercise control including the moral domain, conventional domain, and personal domain (Barber et al., 2005; Nucci, 2001; Nucci, Hasabe, & Lins-Dyer, 2005; Smetana, 1995; Smetana, Crean, & Campion-Barr, 2005; Turiel, 2005). Parenting in the moral domain includes parental influence on children's values and character development, while parenting in the conventional domain addresses teaching children social norms and expectations. Established norms in these domains are well recognized within families, communities, cultures, and society (Nucci, Hasabe et al., 2005; Turiel, 2005). The third domain, the personal domain, involves parental influence on children's identity development and is the most subjective domain (Barber et al., 2005; Baumrind, 2005; Nucci, Hasabe et al., 2005; Turiel, 2005).

Measuring parental involvement in the first two domains is accomplished through observing if children are able to behave and make decisions based on the norms they've been taught. Once they are able to consistently do so independently, parenting in these domains minimizes to only advising when needed (Barber, 1996; Barber et al., 2005). Parenting in the personal domain is more difficult to measure because identity development is constantly taking new directions during childhood and adolescence and

there is not an identity norm to strive for (Barber, 2002; Barber et al., 2005; Baumrind, 2005; Schaefer, 1965). In fact parents and adolescents often disagree on what falls within this domain and what does not (Barber et al., 2005; Baumrind, 2005; Turiel, 2005).

Tripartite Parenting Behaviors

Building on Schafer's (1965) parenting behavior classifications of acceptance, psychological control, and firm control along with Baumrind's (2005) parenting typologies, Steinberg, Dornbusch et al. (1992) developed an updated model of tripartite parenting behaviors. According to this model, parental control over the moral domain and conventional domain, behavior control, focuses on teaching children socially and morally appropriate behavior by monitoring activity and setting limits (Barber et al., 2005; Baumrind, 2005; Nucci, Hasabe et al., 2005). As children develop through childhood to adolescence, parents gradually decrease the amount of behavior control they exercise in order to promote self regulation and independence (Barber, 1996; Barber et al., 2005). Parental support, the next parenting behavior, involves being nurturing and showing affection towards children and adolescents to create a safe environment that allows for optimal autonomy development. Parental support is exercised throughout development in each domain (Barber et al., 2005; Schaffer, 1965; Steinberg & Dornbusch et al., 1992). Finally, parental control over the personal domain, psychological control, attempts to influence children's personal interests, style, and conduct. Excessive use of psychological control has been linked to negative outcomes including delayed identity development, poor academic performance, increased conflict between parents and children, and more internalizing disorder symptoms such as anxiety and depression

(Barber et al., 2005; Nucci, 2001; Schaffer, 1965; Smetana, Crean, & Campion-Barr, 2005; Steinberg, Dornbusch et al., 1992).

Steinberg, Dornbusch et al. (1992) discovered in their research that parents have a tendency to practice more psychological control during transitions such a moving to junior high or high school. Also, parents of lower socio-economic status have been found to exercise higher levels of psychological control due to the perception of increased risk in their environment (Lins-Dyer, 2005; Nucci, Camino, & Minitsky-Sapiro, 1996; Smetana, Crean, & Daddis, 2002; Smetana & Daddis, 2002; Turiel, 2005).

Psychological control plays the largest role in the parent child relationship post childhood, especially during times of transition.

Influences on Parenting Behaviors

According to Serbin and Karp (2003), parenting style is a combination of the parenting style one grew up with, individual personality, and situation. Other studies have identified a number of factors that influence parenting styles and behaviors including work hours, marital relationship, financial status, personal goals, social networks, and recreational activities amongst others (Barber et al., 2005; McNall, Eisenberg, & Harris, 1991; Roberts, Block, & Block, 1984; Turiel, 2005). The most significant factor, according to the developmental niche theory, is parents' cultural beliefs about development and their own experience growing up with their parents (Gallimore, Goldenberg, & Weisner, 1993; Super & Harkness, 1986; Sy et al., 2007). According to Chao's (2002) educational niche theory, this influence carries over into parental involvement in a child's education as well. Parents' beliefs about education and what constitutes appropriate educational attainment determine their level of involvement

parents have in their children's education (Chao, 2002; Coleman & Churchill, 1997; Kim & Rohner, 2002; Martini, 1995; Sy et al., 2007).

Parental Influence in Education

Parental influence also carries over into the educational experience, impacting students' performance in elementary, secondary and higher education (Chao, 2002; Hair, Moore et al., in press; Herman, Dornbusch, Herron & Herting, 1997; Moore et al, 2004). In education, no one is more invested in the operation of a school than the parents of the children who attend. Parents provide a source of quality assurance and social capital, as they encourage both the school and their children to perform to their potential (Coleman, 1998; Lin 2001a, 2001b; Perna & Titus, 2005). According to Perna and Titus (2005), "aspirations and family support foreshadow student success" (p. 12). Parental expectations for children to attend college is one of the strongest predictors of students enrolling in college and persisting (Association for the Study of Higher Education [ASHE], 2007; Perna & Titus, 2005). Persistence is further impacted by the attention parents gave to education and the study routines they taught their children during elementary and middle school, which are carried on to college (Schilling & Schilling, 1999). Involving parents in their children's education benefits the children themselves, the families, the school, and the community at large (Sliwka & Istance, 2006; Sy et al., 2007).

Successfully fostering healthy involvement in education is a challenging task, however. Families in today's society are diverse, representing various structures, dynamics, cultures, and ranges of socio-economic status (Levine & Curetan, 1998). It is

necessary to understand what constitutes appropriate involvement and availability for each family in order to assist them in being a positive, active influence in their students' education and development (Griffith, 1998; Hill, 2001; Hill et al., 2004; Jeynes, 2003; Wong & Hughes, 2006). Adding to the challenge, parental involvement is a difficult construct to study due to the subjective nature of parent and teacher self reporting, the tendency of parents to exaggerate their level of involvement, researchers using different definitions of parental involvement, and different perceptions of appropriate involvement (Baumrind, 2005; Fan & Chen, 2001; Griffith, 1998; Hill, 2001; Nord, Lennon, Liu, & Chandler, 1999; Reynolds, 1992; Smetana, 1995; Wong & Hughes, 2006). Also, the majority of research on parental involvement in education until recently has focused on elementary and secondary education. More research is needed to determine the impact of parental involvement at the college level.

Despite these barriers, researchers have come a long way in understanding parenting and the role of parents in their children's education. The parent-child relationship is a complicated phenomenon to navigate across the development of the child, but understanding the influence of, theory behind, and best practices for parenting arm parents and educators with some guidelines (Turiel, 2005). Additionally, understanding parents' approach to parenting in general offers insight into their approach to involvement in their child's education. Parental involvement has the potential to benefit students at all levels of education from pre-school through college, but only when it is exercised appropriately, allowing for autonomy and identity development (Hill et al., 2004; Jeynes, 2003; Wong & Hughes, 2006). It is imperative that educators understand how to define and foster appropriate involvement, and due to mandates of No Child Left

Behind and highly involved parents today, reaching this understanding is no longer optional (Savage, 2007; Sy et al., 2007; United States Department of Education [USDE], 2005).

Benefits of Parental Involvement in Education

The benefits of parental involvement in children's education are well established in the research. It has been linked with students being more motivated (Wong & Hughes, 2006), making higher grades (Christenson, 1999; Desimore, 1999; Dornbusch et al., 1987; Fehrmann et al., 1987; Lee, 1993; Martini, 1995; Muller, 1993; Perna & Titus, 2005; Steinberg, Lamborn, et al., 1992; Sui-Chu & Willms, 1996; Zick et al., 2001), choosing positive peer groups (Brown et al., 1993; Falbo et al., 2001), experiencing higher levels of extracurricular achievement (Linver & Silverberg, 1997; Paulson, 1994), adjusting more successfully to transition (Lee, 1993; Wong & Hughes, 2006; Zick et al., 2001; Reynolds, 1992), and being less likely to drop out (Horn & West, 1992; McNeal, 1999; Rumberger et al., 1990; Sy et al., 2007). Additionally, parents who are involved in their child's education are more satisfied with the education their child is getting (Bird, 2006).

Defining Parental Involvement in Education

Referring back to the Chao's (2002) educational niche theory, being "involved" means something different to each parent based on his or her experiences. While it is important to remember to not make assumptions about parents' preferred level of involvement, research has established some trends on this subject. Parents who have attained higher degrees of education themselves experienced more satisfaction with their own educational experiences, feel more confident in their ability to assist their child in

his or her education, are in a traditional family structure, have a larger social network, and are of higher socio-economic status all of which contribute to more involvement (Coleman & Churchill, 1997; Moles, 1993; Nord & West, 2001; Reynolds & Grill, 1994; Sheldon, 2002; Stone, 2006). At the other end of the continuum, research has shown that ethnic minority parents, immigrant parents, and parents of a lower socio-economic status tend to be less involved (Chavkin & Williams, 1993; Hill et al., 2004; Kohl, Weissberg, Reynolds, & Kasprow, 1994). Boethel (2003) pleads the case that this is not necessarily due to lack of interest but could be a result of lack of understanding of what is appropriate involvement, lack of transportation, and conflicting work schedules.

It is also important to remember that parental involvement can take many different forms. Sy et al. (2007) provide a broad definition of parental involvement as "parental behaviors aimed at promoting or enhancing a child's educational development" (p. 2). These behaviors can occur across multiple contexts including at home, in school, and in the community. At home, parents create an environment that is conducive to learning and assist their children with homework assignments (Baker, Kessler-Sklar, Piotrkowski, & Parker, 1999; Chao, 2002; Choi, Bemechat, & Pinsberg, 1994; Kim, 2002; Mau, 1997). Also, holding conversations about what the child is learning in school and showing interest is a way to be actively involved (Stone, 2006). At school, parents can take advantage of different opportunities for involvement and build relationships with teachers and administrators (Morrow, 1989). Finally, parents can utilize resources in the community that connect to the skills and material their children are learning in school, such as visiting a history museum (Grolnick 2003; Gutman & McLoyd, 2000; Sun,

1998). Just as parental involvement occurs differently in different families, different levels of education take different approaches to parental involvement.

Parental Involvement in Elementary and Secondary Education

Parents begin their involvement in their children's education at the elementary school level. The majority of research on parental involvement in education has focused on these years through the high school years. This section outlines the components of parental involvement during this stage.

Opportunities for Parental Involvement

Encouraging parental involvement is a regular for practice educators in elementary and secondary education, especially since No Child Left Behind made it a mandate (Dornbusch & Glasgow, 1996; Sliwka & Istance, 2006; Stone, 2060; USDE, 2005). Opportunities for parental involvement in elementary through high schools are abundant, especially at the lower levels, including parent associations, family day events, fundraising projects, sporting events, and volunteering to list some traditional venues. Some schools are also encouraging involvement through creative means of offering parent workshops, parent socials, and extra credit for students who are taking their education seriously at home, as seen through parent report cards submitted to the school (Benson & Martin, 2003; Carlise, Stanley, & Kemple, 2005; Sliwka & Istance, 2006). Finally, technology is connecting parents to schools through Student Information Systems, allowing parents to monitor their children's attendance, grades, and involvement via their computers at any time that is convenient to them (Bird, 2006). Parents have full access to all their students' information.

Barriers to Appropriate Parental Involvement

Despite well-established benefits of and ample opportunities for parental involvement, there are still some parents who are not involved in their children's education. Some possible reasons are not feeling welcome, fearing consequences for children, language barriers, and feeling unsure about appropriate levels of involvement (Al-Hassan & Gardner, 2002; Coleman & Churchill, 2007; Sheldon, 2002; Sliwka & Istance, 2006). Also, some teachers contribute to lack of involvement due to feeling threatened by it or not understanding how to cultivate it (Hilliard & Pelo, 2001; Hurt, 2000; Keyser, 2001; Sliwka & Istance, 2006). Barriers to involvement must be identified and removed in each school in order to create an environment that is welcoming to parents (Lee, 1993; Wong & Hughes, 2006; Zick et al., 2001; Reynolds, 1992).

Encouraging Appropriate Parental Involvement

Keeping in mind that parents are stakeholders in schools, it is important to make adjustments to meet their needs (Carlisle et al., 2006; Ferrera & Ferrera, 2005; Sliwka & Istance, 2006). This is a complex task considering the diverse group of parents educators are working with today, but research has uncovered some best practices for overcoming barriers and achieving increased levels of parental involvement for all families (Lee & Manning, 2001; Levine & Curetan, 1998; USDE, 2005; Swick, 2006; Sy et al., 2007). Personalized communication to parents, including phone calls and written correspondence, makes parents feel more welcomed into the school environment and as a result into their child's education (Benson & Martin, 2003; Halsey, 2004; Swick, 2006). Creating a welcoming environment physically also encourages parents to come. Signs should be in place to direct parents to the appropriate place to check in and staff should

assist parents with making their way to their destination (Benson & Martin, 2003; Swick, 2006). Another best practice is including parents as members of the classroom by recognizing them for their contributions and providing them with educational opportunities themselves (Benson & Martin, 2003; Comer, 2001; Halsey, 2004; Swick, 2006). Finally, when teachers schedule appointments and events, it is important to keep in mind that parents may have alternative work schedules or other children to take care of. Educators must be creative to make parental involvement possible, for example by welcoming extended family to events, providing babysitting, and holding events at alternate times (Benson & Martin, 2003; Boethel, 2003; Griffith, 1998; Hill, 2001; Hill et al., 2004; Jeynes, 2003; Swick, 2006; Wong & Hughes, 2006).

Parental Involvement in Higher Education

While educators in elementary and secondary schools are looking for ways to get parents more involved in their children's education, educators in higher education are trying to learn how to manage the increase in parental involvement on their campuses (Coburn, 2006; Colavecchio-Van Sickler, 2006; Daniel et al., 2001; Lange & Stone, 2001; Lum, 2006; McGinty, 2002; Merriman, 2007; Scott & Daniel, 2001). Popular media has recently been highlighting this phenomenon of parents of college students being hyper-involved in their children's education, referring to them as helicopter parents (American Broadcasting Company [ABC], 2006; Davidson, 2008; Ettinger, 2006; Freedman, 2004; Jacobson, 2003; Kadaba, 2004; Lane, 2004; Strauss, 2006). In determining what qualifies as appropriate parental involvement for each family, educators in higher education must help families understand their developing students

and changing relationship with their students (Gerdes, 2004; Hill, 2001; Hill et al., 2004; Griffith, 1998; Jeynes, 2003; Levine & Cureton, 1998; Wong & Hughes, 2006).

Compounding this challenge, more research is needed to understand and define appropriate parental involvement in postsecondary education (Daniel et al., 2001; Epstein & Sanders, 2002; Hill & Taylor, 2004; Stone, 2006).

Parental Over-Involvement

While there is no mandate requiring administrators in higher education to engage parents in their children's education, it is necessary to do so in order to meet parents' expectations and survive in today's competitive market (Coburn, 2006; Daniel et al., 2001; Lange & Stone, 2001; Lum, 2006; Merriman, 2007; Pattenaude, 2000; Savage, 2007; Scott & Daniel, 2001). College students' parents are more involved than ever before in their children's college experience (Coburn, 2006; Colavecchio-Van Sickler, 2006; Daniel et al., 2001; Lange & Stone, 2001; Lum, 2006; McGinty, 2002; Merriman, 2007; Scott & Daniel, 2001). They are assisting their children with choosing where to attend (Bers, 2005; Brooks, 2004; Coburn, 2006; David, Ball, Davies, & Reay, 2003; Mansfield & Warwick, 2005; McGinty, 2002), providing financial support (Daniel et al., 2001; Scott & Daniel, 2001; Hossler & Vesper, 1993; Lange & Stone, 2001), responding when their children experience issues (Conneely et al., 2001; Daniel et al., 2001), and communicating with their students via email, text messaging, and cell phones (Merriman, 2007; Savage, 2007; Strage & Brandt, 1999).

It is understandable why parents are so involved in their college students' education. They have been told by the media and educators since their children were in kindergarten that it is important for them to take an active role in their children's

education (Bird, 2006; Coleman, 1998; Lin, 2001 a, 2001 b; Perna & Titus, 2005; Sy et al., 2007; Steinberg & Lamborn et al., 1992; Sui-Chu & Willms, 1996; Wong & Hughes, 2006; Zick et al., 2001). Technology also makes parental involvement possible, even from a distance, and students are often the ones initiating the involvement (Lange & Stone, 2000; Merriman, 2007; Savage, 2007; Strage & Brandt, 1999). In addition, parents are concerned about campus safety and health risks such as smoking, alcohol, sleeping habits, and nutrition (Bylund et al., 2005; Mansfield & Warwick, 2005; Shellenbarger, 2006). Finally, a child's higher education is a significant investment, and in today's society of consumerism, parents want to get their money's worth (Coburn, 2006; Conneely et al., 2001; Daniel, Evans, & Scott, 2001; Lange & Stone, 2001; Merriman, 2007; Savage, 2007; Shellenbarger, 2006).

Parents can easily become over-involved in their college students' lives and often do due to being worried about their student, not having correct information, or not having an accurate understanding of how to appropriately support their student (Gerdes, 2004; Merriman, 2007; Savage, 2007). The transition from high school to college presents multiple challenges for students. They are living on their own for the first time, facing new academic challenges, deciding on a career, learning to manage their finances, and being exposed to diverse situations that are unfamiliar to them (Baxter-Magolda, 1999; Conneely et al., 1999; Dyson & Renk, 2006; Evans, Forney, & Guido-Dibrito, 1998; Smith & Renk, 2007). While these challenges result in development for college students, they can also result in anxiety for parents (Boyer, 1987; Hood, 1984; Kuh et al., 2005; Pascarella & Terrenzini, 1991, 2005). The college transition is a significant event in the

lives of parents too, as they must adjust and learn to appropriately support their students' independence, intimacy, identity, and intellectual development (Coburn, 2003).

Parental Involvement and College Student Outcomes

Parental involvement in college has received mostly negative attention recently, but some researchers are pointing out positive effects of parental influence on college students. Parental support can be extremely helpful to students going through the difficult transitions college presents by helping students feel comfortable taking on challenges and risking failure (Baxter-Magolda, 1999; Conneely et al., 1999; Daniel et al., 2001; Elkind, 1994; Holahan & Moos, 1981). According to The American Freshman: National Norms for Fall 2007 reported by the Higher Education Research Institute [HERI] (2007), college freshman were satisfied with the level of involvement their parents took in their college experience. Over 70% of students surveyed reported that the amount of involvement their parents had in assisting them with completing applications, choosing courses, and choosing activities was just right. Over 75% of these students reported their parents' involvement in encouraging them to attend college and dealing with college officials was just right (HERI, 2007). The percentage of students indicating their parents' involvement was too little was less than 10% in all areas (HERI, 2007).

While some students are content with their parents' involvement in their lives, more information is needed about what behaviors make up this parental involvement and the impact it is having on students. Several recent studies have looked at the relationship between parental influence and different student outcomes, and findings have been mixed.

Parental Support, Parental Involvement, and College Student Persistence Ratell, Larose, Guay, and Senecal (2005) looked at the relationship between perceived parental involvement and student persistence amongst college students in a science curriculum. In the study, 729 undergraduates in Quebec completed a survey adapted for the study that measured socioeconomic status, achievement in science, perceived parental autonomy support, perceived parental involvement, feelings of competence, feelings of autonomy, feelings of relatedness, and persistence in a science program. The study used Grolnick's (2003) definition of parental involvement which includes spending time with and emotionally supporting children and his definition of parental autonomy support which is recognizing children's individual strengths and allowing them to make their own choices. Findings support a relationship between parental autonomy support and parental involvement with student feelings of competence, autonomy, and relatedness. Researchers also found that parental autonomy support predicted student persistence but parental involvement did not directly. Ratell et al.'s (2005) study supports that parental autonomy support and involvement are beneficial to students in helping build competence and autonomy, which in turn are necessary for

Parenting Style, Parental Involvement, and College Student Goal Orientation

Gonzalez, Greenwood, and WenHsu (2001) researched perceived parenting style,
parental education, parental involvement, and college students' goal orientation. A

sample of 311 students completed a survey adapted from the Goals Inventory (Roedel,
Schaw, & Plake, 1994) and the Parental Authority Questionnaire (Buri, 1991). Results
indicated a relationship between a mastery goal orientation, where students are interested

students to persist.

in genuine learning and challenge and tend to be intrinsically motivated, and maternal authoritativeness, where mothers offer high levels of autonomy support and realistic demands (Baumrind, 1967; Ginsburg & Bronstein, 1993). An additional relationship was found between performance orientation, where students are interested in successfully completing tasks and tend to be extrinsically motivated and paternal authorianism, where fathers put high demands on their children and offer minimal support. Performance orientation was also related to parents helping with homework (Ginsburg & Bronstein, 1993). While parental involvement was found to be positively correlated with authoritative parenting, no direct relationship was found between parental involvement and goal orientation besides parental help with homework assignments. Overall, Gonzalez et al.'s (2001) study does not support that parental involvement is related to college student goal orientation.

Parenting Style and Self-Regulation in College Students

Numerous studies have supported a relationship between parenting style and favorable academic outcomes in pre-college aged students (Dornbusch et al., 1987; Steinberg, Lamborn et al., 1994; Steinberg, Lamborn et al., 1992). Strage (1998) conducted a study with 465 college students to determine if these benefits also exist at the college level. Each participant completed the Student Attitudes and Perceptions Survey. As predicted, students who perceived their parents as authoritative and their families as close also scored positively for confidence, positive goal orientation, concern about planning for their future, adjustment to college, self-regulated learning behaviors, and a sense of being in control when it comes to academics. On the other end of the continuum, students who perceived their parents as authoritarian felt out of control of

their academic lives. Strage's (1998) study provides additional support for parental influence impacting college students.

Parenting Style, Parental Support, and Academic-Related Stress in College Students

Smith and Renk (2007) investigated the relationship between college student coping strategies, social support, parenting style, and academic stress. A sample of 93 undergraduate students completed a survey measuring each of these constructs. Findings involving parents and student stress were mixed. Researchers did find a positive relationship between social support from parents and the use of problem focused coping, which is characterized by attempting to change a situation to alleviate stress (Lazarus & Folkman, 1984), but did not find a direct relationship between parenting style and stress. Only paternal use of authoritarian parenting during female students' childhood years showed a significant positive relationship to student stress. The researchers postulated that perhaps parental influence makes an impact during pre-college years and then begins to play less of a significant role once students transition to college.

Parenting Style and Perfectionism in College Students

Following up from a study conducted by Frost, Lahart, and Resnblate (1991) that uncovered a relationship between perfectionism in mothers and college-aged daughters, Flett, Hewitt, and Singer (1995) looked at perfectionism and parenting style in college students. One hundred college students participated by completing the Multideminsional Perfectionism Scale (Hewitt & Flett, 1991) and the Parental Authority Questionnaire (Buri, 1991). They found positive significant relationships between both maternal authorianism and paternal authoritarianism and socially prescribed perfectionism in

males. In the female portion of the sample, significant relationships emerged between both maternal authoritativeness and paternal authoritativeness and self-oriented perfectionism. Flett et al. (1995) attributed these findings to the trend of parents placing more emphasis on achievement with sons (Huston, 1983; Parsons, Adler, & Kaczala, 1982) and daughters setting higher goals for themselves when they have strong family support. Overall, this study provides support for a relationship between parenting style and perfectionism in college students.

Parenting Style and Self-Actualization in College Students

Dominguez and Carton (1997) investigated parenting style and self-actualization in college students. Abraham Maslow's (1954, 1970) theory of motivation, postulates that individuals must first meet lower level needs in order to achieve self-actualization and positive social support from others helps facilitate progress. In a previous study, Nystul (1984) found a positive relationship between positive parenting behaviors and characteristics of self-actualization in children. The purpose of this study was to find if the same relationship existed with college students. Two hundred college students participated, completing the Short Index of Self-Actualization (Jones & Crandall, 1986) and the Parental Authority Questionnaire (Buri, 1991). As predicted, findings did indicate a significant positive relationship between both maternal authoritative parenting and paternal authoritative parenting and self-actualization as well as a significant negative relationship between parental authoritarianism and self-actualization in college students. Parents exercising positive involvement in their student's lives through and appropriate balance of responsiveness and demandingness does contribute to their students reaching their full potential.

Parenting Style and Self-Perception in College Students

Multiple studies have found a relationship between parenting style and selfperception in adults (Gecas & Schwalbe, 1986; Graybill, 1978; Hopkins & Klein, 1993;
Thomas & Raj, 1985). Klein, O'Bryant, and Hopkins (1996) conducted a similar study to
find out if the same is true in a sample of college students. As predicted, in a sample of
207 college students results indicated a significant negative relationship between both
maternal and paternal authoritarianism and positive self-perception as well as a
significant positive relationship between both maternal and paternal authoritativeness and
positive self-perception. College students whose parents practiced authoritative parenting
possessed higher self-perception than their peer whose parents practice authoritarian
parenting techniques. These results support an influence of past parenting practices on
college students' experiences in college and resulting perceptions of themselves and their
abililities.

Parenting Style and College Academic Adjustment and Success

Strage and Brandt (1999) studied the role parenting style plays in the academic adjustment and success of college students. In their study, 236 students ranging from freshmen to seniors completed the Student Attitudes and Perceptions Survey, which gathers data regarding students' relationship with their parents, confidence, persistence, ability to remain focused, and perception of instructors as a source of support. The results showed several meaningful relationships. Parental use of autonomy granting was found to be positively correlated with overall grade point average, persistence, and a good relationship with instructors. Parent demandingness was positively correlated with student confidence and a good relationship with instructors. Finally parent

supportiveness was positively correlated with confidence, persistence, and teacher rapport. These findings provide well-rounded support that parents indeed have an impact on their college students. Strage and Brandt found that this impact is stronger during the initial college years and decreases as students near graduation.

Parenting Style and College Student Self-Esteem, Self-Efficacy, and Depression

While the majority of studies looking at the impact of parenting style on college
students assesses parenting style used during childhood, Oliver and Paull (1995) assessed
current parenting style used and its impact on college students' self-esteem, self-efficacy,
and depression. One hundred and eighty-six students participated by completing the Self
Esteem Inventory (Cooper Smith 1967, 1981), Self Efficacy Scale (Tipton- &
Washington, 1984), Child Report of Parental Behavior Inventory (Schaefer, 1965),
Family Environment Scale (Moos, Ingel, & Humphrey, 1974), and the Beck Inventory
(Center for Cognitive Therapy, 1978). Results of the study confirmed that parenting style
was responsible for 13% of the variance in the sample's reported levels of self-esteem,
self-efficacy, and depression. This study confirms that parenting does play a role in
children's sate of mental well-being and should be investigated further to determine how
parents can have a positive effect.

Parenting Style and College Student Decision Making

As individuals move through adolescence, their focus shifts significantly from parents to peers, impacting their decision making processes (Erikson, 1959; Larson, 1974, 1983). Bednar and Fisher (2003) conducted a study to determine how parenting style influences decision making in college students. The researchers found that students who grew up with authoritative parents tended to turn to their parents for guidance with

moral and informational decisions while students with authoritarian or neglectful parents turned to peers for this guidance. It can be assumed that the impact of these different decision making strategies can be extreme, especially when life-altering decisions are being made.

Parental Generativity and College Student Outcomes

According to Erickson's (1982) theory of psychosocial identify development, generativity versus stagnation is the conflict faced during midlife, when many parents experience their children leaving for college. Peterson (2006) conducted a four-year longitudinal study to research the impact of parental generativity, a focus on making a positive contribution to future generations (Erikson, 1982), on college student outcomes. Sixty-nine students participated, and the results confirmed a positive relationship between parental generativity and several student outcomes including positive affect, future orientation, and prosocial characteristics such conscientiousness and openness to new experiences. A negative relationship was identified between parental generativity and neuroticism in the college students. His study supports the notion that parental influence does impact the college experience, specifically in terms of mental health. The impact parents had was positive and can even help students with avoiding negative mental health experiences.

Overview of Studies

The majority of past studies looking at parental influence and college student outcomes have focused on parenting styles, especially parenting styles used during students' childhood and adolescence. Researchers have found that children internalize early relationships and experiences with parents, which then influence them for the

entirety of their lives (Bowlby, 1982). However, a limitation of these studies lies in the acknowledgement that retrospective reports are subject to bias. Nevertheless, the studies do provide support that parenting style positively influences goal orientation, self-regulation, perfectionism, self-actualization, self-perception, academic adjustment, self-efficacy, and decision making in college students (Bednar & Fisher, 2003; Dominguez & Carton, 1997; Frost et al., 1991; Gonzalez et al., 2001; Klein et al., 1996; Oliver & Paull, 1995; Smith & Renk, 2007; Strage, 1998; Strage & Brandt, 1999; Ratell et al., 2005).

Several studies looked specifically at parental involvement, time and emotional support provided to students (Grolnick, 2003). Findings provide evidence that parental involvement does impact feelings of competence and autonomy, but no direct relationships were found with goal orientation or persistence (Gonzalez et al., 2001; Ratell et al., 2005). Finally, Peterson (2006) looked at parental generativity, which was found to encourage pro-social behaviors and defend against neuroticism. Overall, parental influence does impact college students, especially in regards to parenting style used and especially during the earlier years of college (Weiss & Schwarz, 1996; Strage & Brandt, 1999). Some of the outcomes demonstrated parents can have a negative effect on student mental health while other confirmed a positive effect is possible. More research is needed to gain a clearer understanding of what leads to these positive effects.

Parental Influence on College Student Mental Health

The previous studies confirm that parents do impact their college students in multiple capacities. The majority of the findings support opportunities for parents to positively influence their students through their involvement. However, there is also

evidence supporting that parental involvement can be harmful to students when exercised inappropriately (Lapsley, Rice, & Shadid, 1989; Sax, Bryant, & Gilmartin, 2002). Multiple studies have confirmed that the majority of adolescents who struggle with internalizing disorder symptoms grew up having unhealthy relationships with their parents (Chorpita & Barlow, 1998; Davies & Windle, 1997; Kataine et al., 1999; O'Conner et al., 1998; Wagner, Cohen, & Brooks, 1996). In fact, parents are believed to have the strongest impact on adolescent emotional well-being, especially for females (Jackson, Bijstra, Oostra, & Bosma, 1998; Juang & Silbereisen, 1999; Kenny & Donaldson, 1991; Mayhew & Lempers, 1998; Rogers & Holmbeck, 1997).

Parenting during the college years is especially challenging due to the significant transitions experienced by both students and parents. Both parties find themselves living new routines that bring new responsibilities, roles, and assumptions. Relationships with parents during this time are determined by multiple factors including heredity (Bouchard & McGue, 1990; Elkins, McGue, & Iacono, 1997; Hur & Bouchard, 1995; McGue, Elkins, Walden, & Iacono, 2005; Plomin & Bergeman, 1991; Plomin, McClearn, Pedersen, Nesselroade, & Bergeman, 1988; Rowe, 1981, 1983), parenting style (Bates, Petit, Dodge, & Ridge, 1998; Rubin, Burgess, Dwyer, & Hastings, 2003; Stoolmiller, 2001), and adolescent personality (Paulussen-Hoogeboom, Stams, Hermanns, & Peetsma, 2007; South et al., 2008). Additionally, during emerging adulthood parents' willingness and ability to support their child plays a part. Considering these numerous influences, the parent-adolescent or parent-emerging adult relationship can take many different turns based on these factors. Parents are left to figure out how to best support their children in

making good decisions through these transitions while adapting themselves to a changing relationship with their child.

Understanding Mental Health on the College Campus

The mental health scene of a college campus is determined by the makeup of the individuals on campus and the definition of mental health adopted by the observer. A consistent operational definition of mental disorder does not exist; however, the most respected definition, offered by APA (2000), is "a clinically significant behavioral or psychological syndrome or pattern that occurs in an individual and that is associated with present distress or disability or with a significantly increased risk of suffering death, pain, disability, or an important loss of freedom." (p. xxxi) Mental disorders fall into one of the following categories: disorders first diagnosed in infancy childhood, or adolescence; cognitive disorders; mental disorders due to a general medical condition; substance-related disorders; psychotic disorders; mood disorders; anxiety disorders; factitious disorders; dissociative disorders; sexual and gender identity disorders; eating disorders; sleep disorders; impulse-control disorders; adjustment disorders; personality disorders; or not yet identified disorders (APA, 2000). Any of these disorders could be present on a college campus.

Challenges of the College Transition

Entering into college is a significant and stressful life transition. Chandler and Gallager (1996) categorize the challenges students face as social, academic, and psychological. Socially students are adjusting to changes in their social network that at times can leave them with a weak support system (Sax, 1996). Also, some students experience difficulties balancing different groups of friends such as high school friends

and college friends or on campus and off campus friends (Christie & Dinham, 1991).

Romantic relationships can contribute to stress as well, especially when long distance relationships or breakups are involved (Guldner, 1996; Kaczmark, Backlund, & Bremer, 1990; Mearns, 1991; Reisman, Whalen, Frost, & Marganthall, 1991; Sax, Bryant et al., 2002). Academically students must learn how to succeed under new academic demands. The typical college classroom with large enrollment and minimal opportunities for interaction does not provide the same supportive environment the high school classroom students are accustomed to does (Martinez Aleman, 1997; Sax, Bryant et al., 2002).

Also, students are faced with the challenge of choosing an academic discipline and beginning to define career goals while adjusting to all of these changes (Evans et al., 1998; Sax, 1996).

All of these stressors can add up and lead to psychological distress. Learning to live independently in itself creates stress (Baxter-Magolda, 1999; Conneely et al., 1999; Smith & Renk, 2007). Many students make poor health decisions including excessive drinking (O'Malley & Johnston, 2002; Slutsko et al., 2004; Wechsler, Dowdall, Davenport, & Castillo, 1995), drug use (Schulenberg et al., 2005), poor nutrition (Beerman, 1991; Hertzler & Frary, 1992; Watabe-Dawson & Sosak, 2000), sexual activity with multiple partners (Desiderato & Crawford, 1995), and lack of sleep (Gallagher, Gill, & Goldstrom, 1998). Financial stress also plays a role with current increases in tuition and decreases in availability of financial aid. More and more students have to work while in school, taking away from study and personal time (Andrews & Wilding, 2004; Sax et al., 2002). The end result of all these factors is

increased stress levels and higher mortality rates in college students (Fromme, Kruse, & Corbin, 2008; Hingson et al., 2005).

In their book *College of the Overwhelmed: The Campus Mental Health Crisis and What to Do about It*, Kadison and DiGeronimo (2004) explain that the college transition is especially challenging for today's student. Recent tragedies including September 11 and shootings on college campuses have created a state of uncertainly in our country. Also contributing to this uncertainty are the ongoing conflicts overseas. In a study carried out by the Associated Press and Music Television University (Lipka, 2008) out of over 2,000 college students surveyed 50% reported knowing someone who was currently serving or had served in a war. Out of those students 55% reported this adding to stress in their lives. The economy today is another contributor to the current state of uncertainty. Students today feel more pressure to not only attend college but to choose a career that will provide a significant income, even if that area of study does not match their skills (Kadison & DiGeronimo, 2004). Previous studies have found that incongruence such as this in career choice can easily lead to stress and depression (Hinkleman & Luzzo, 2001).

Reason for Concern

Astin (2003) identified in *What Matters in College: Four Critical Years Revisited* that levels of student psychological well-being are declining during the college years. More students are experiencing common mental health issues while in college including depression, anxiety, relationship issues, drug and alcohol abuse, eating disorders, learning disabilities, obsessive compulsive disorder, and grief issues in addition to sexual abuse, suicide, and more advanced disorders such as schizophrenia and bipolar personality

disorder (Arehart-Treichel, 2002; Caba, 2003; Benton, Robertson, Tseng, Newton, & Benton, 2003; Kadison & DiGeronimo, 2004; Ambler, 2007). Several possible explanations exist including that more students are coming to college with these disorders; onset of more advanced disorders does not typically occur until the early twenties; and students face a significant number of stress factors today (APA, 2000; Riba, 2004).

It is not uncommon for college freshmen to experience feelings of loneliness and depression (Sax, Bryant, et al., 2000). According to data from the 2000 College Institutional Research Program's [CIRP] Freshman Survey and 2001 Your Freshman College Year [YFCY] Survey, out of a sample of 3,680 freshmen 10.7% of females and 5.5% of males reported frequent feelings of depression. Fifty-nine percent of females and 48.2% of males experienced depressive symptoms occasionally (Sax, Bryant et al., 2002). The American College Health Association National College Health Assessment Spring 2007 Reference Group Data Report, which summarizes the findings from 71,860 college student respondents, further supports a reason for concern. Over the previous year 63.3% of respondents had felt helpless, 93.2% overwhelmed, 90.0% exhausted, 70.2% sad, and 45% felt so depressed it was difficult to function. Fifteen percent or 10,995 students had actually been diagnosed with depressions at some point in their life. Out of this group of 10,995, 32.5% had received a diagnosis within the past year, 24.8% were in therapy when they took the survey, and 34.9% were taking medication for their depression. Some of the students taking the survey had even attempted, 1.5%, or seriously considered, 9.8%, suicide. Due to the close proximity in which college students live and interact, when even a small percentage of students are dealing with mental health concerns such as these it can be disruptive for the entire living environment.

Addressing Mental Health Concerns on the College Campus

After World War I, returning veterans appeared on college campuses to pursue an education and brought with them the adjustment challenges and mental health challenges that accompany return from war. As a result, postsecondary institutions brought advisors and counselors onto campus to help their student population address these needs. Again after World War II, more veterans arrived on campus and the critical need arose for college counseling centers (Rudolf, 1994; Schneider, 1977). Counseling centers today have become standard on college campuses and play a leading role in addressing the mental health needs of students (Ellen, 2005). Over the past decades the majority of these centers' focus has been on developmental and transitional issues, and once again campuses are experiencing a change in the mental health needs of their students with an increase in the number of students dealing with mental illness and an increase in the severity of these issues (APA, 2000; Riba, 2004). Counseling centers are being expected to adapt to these needs while operating with the same resources. Dr. Michelle Riba (2004), a leader in the American Psychological Association, has called attention to this problem and the need to address it.

College Campus Mirroring Society

The field of psychology overall has experienced this shift to a focus on mental illness (Ambler 2007; Keyes & Lopez, 2002; Keyes & Haidt, 2003; Seligman, 1998a). In fact prior APA president Seligman (1998a) named mental health psychology's forgotten

mission (Ambler, 2007). College students are not the only ones struggling with the uncertainties of today and the impact is evident in the current mental health status of society at large (Keyes, 2007). Only 20% of the entire adult population in the United States is functioning at optimal psychological health (Keyes, 2002, 2003, 2004, 2005).

Until recently, physical health care had been the sole focus of national population health. The health care field has successfully increased the life expectancy in this country by approximately 30 years through improving care to reduce and in many instances eliminate infectious diseases and lower infant mortality rates (Gribble & Preston, 1993; Keyes, 2007). However, longer life does not guarantee improved quality of life; instead this country has seen increases in chronic diseases and also mental disorders (Epel et al., 2004; McEwen, 1998; Stiles, 2005). According to Blue Cross and Blue Shield (2006), health care accounts for the largest percentage of gross domestic product in the United States, which is more than housing costs and food costs.

Mental health care has now entered the focus of population health by moving into the position of being the third most costly condition to treat (Keyes & Lopez, 2002; Keyes, 2007). Mental illness is one of the top five illnesses that contribute to shortened life span across the world (Murray & Lopez, 1996, 1997). In the United States, 50% of adults will experience a serious mental illness before they reach 55, and 25% will experience one consistently every year (Robins & Regier, 1991; United States Public Health Service, 1999).

The Need for a Paradigm Shift

Keyes (2007) is calling for a paradigm shift in how mental health is studied and addressed. It is no longer sufficient to focus solely on mental illness; it is imperative that

mental health also be understood and advocated for (Ambler, 2007; Diener, 2003; Harvey & Pauwels, 2003; Keyes, 2002, 2003; Keyes & Haidt, 2003; Keyes & Lopez, 2002; King, 2003; Peterson & Seligman, 2004; Ryff, 2003; Seligman, 1998a, 1998b; Seligman & Pawelski, 2003; Snyder & Lopez, 2002). Keye's mental health continuum is a useful framework to use for making this shift.

The mental health continuum views mental health not as the presence or absence of mental illness but rather as a continuum of health that can be measured as a continuous or categorical variable (Keyes, 2002, 2005, 2007). Categorically, individuals are classified as flourishing, languishing, or moderately mentally healthy. The classification process is similar to diagnosing any mental disorder listed in the *Diagnostic Manual of* Mental Disorders (Keyes, 2007). Mental health, according to the continuum model, is made up of positive emotions, which includes the variables of emotional well-being, and positive psychological functioning, which includes variables of psychological well-being and social well-being. The positive emotions dimension is measured by the level of positive affect and avowed quality of life an individual expresses. The second dimension, positive psychological functioning, is measured by levels of six symptoms of psychological well-being including self-acceptance, personal growth, purpose in life, environmental mastery, autonomy, and positive relations with others and five symptoms of social which are social acceptance, social actualization, social contribution, social coherence, and social integration. These three areas of functioning mirror Seligman's (1998b) pillars of positive psychology which include positive subjective experience, positive individual characteristics, and positive institutions and communities (Ambler, 2007; Aspinwall & Staudinger, 2003; Peterson & Seligman, 2004)

Individuals who have high levels on one or more symptoms of positive emotions and on six or more symptoms of positive social functioning are considered to be flourishing. On the other end of the continuum, those who have low levels on one or more symptoms of positive emotions and on six or more symptoms of positive social functioning are considered to be languishing. Those who do not fit criteria for either of those are considered moderately mentally healthy (Keyes, 2007).

In his multiple studies testing the mental health continuum with middle aged adults, Keyes (2002, 2004, 2005a, 2005b) has established a qualitative difference exists between individuals who are diagnosed as flourishing and those diagnosed as languishing. He has repeatedly found that those who fit the definition of flourishing have had a lower prevalence of mental disorders throughout their lifetime while those who are languishing have shown the highest prevalence of having two or more mental disorders during the past year alone. Keyes (2007) also defines complete mental health as flourishing and having never experienced a mental disorder. Individuals in his studies who qualified as anything other than complete mental health missed more days at work, experienced limitations in carrying out daily activities, suffered chronic physical impairments, and visited their health care provider more often (Keyes, 2002, 2004, 2005a, 2005b). While more research is needed on Keyes' Mental Health Continuum, especially with different populations, it provides a sound framework for addressing the mental health crisis in this country today.

Using the Mental Health Continuum with College Students

According to the American College Personnel Association's [ACPA] *Principles* of *Good Practice* (1996), college administrators are called to assist students in reaching

their potential. Many theorists have now begun to consider college as a stage of emerging adulthood during which students can explore what they want to do with their lives (Andom, 2007; Elkind, 1994; & Rudolf, 1994). Considering that college campuses have multiple support resources available for students, the college years are a critical and logical time to encourage complete mental health (Ambler, 2007; Murano, 2002). Many mental disorders have an estimated onset during the early twenties (Arehart & Treichel, 2002; APA, 2003; Riba, 2004). Researchers have found that individuals who experience a single episode of mental illness are at a significantly higher risk of experiencing mental illness again (Angst, 1988; Gonzales, Lewinsohn, & Clarke, 1985; Lewinsohn, Hoberman, & Rosenbaum, 1988; Piccinelli & Wilkinson, 1994). If college students can be placed in an environment that promotes flourishing, perhaps the onset of these initial episodes and potential future episodes can be avoided, and students can enter adulthood in a state of flourishing.

In 2007, Virginia Ambler was the first to apply Keyes (2007) Mental Health
Continuum to traditionally aged college students. She conducted a study with 534
students to find out the mental health status of a representative sample of college students
and test the relationship between mental health and student involvement. The
participants in the study completed a single questionnaire that included questions from
the National Survey of Student Engagement to measure student involvement, Keyes'
(2002) mental health scales, and demographic questions. The student involvement
questions addressed Kuh's (2005) five measures of involvement including academic
challenge, active/collaborative learning, student/faculty interaction, enriching educational
experiences, and supportive campus environment (Ambler, 2007, p. 40). Keyes' (2002)

three scales of well-being including emotional well-being, psychological well-being, and social well-being were addressed by the mental health questions. Finally, gender, race, and parents' educational attainment were collected in the demographic questions.

Ambler (2007) found that the mental health status of her sample mirrored Keyes's +(2005) findings with his sample of 25 to 74 year olds. Approximately 15% of her sample was flourishing, 67% was moderately mentally healthy, and 17% was languishing. No significant differences were found in gender, parents' educational attainment, or grade point average amongst the three groups. However, there were significant differences in the student involvement scores between those who were flourishing and moderately mentally healthy and between those who were moderately mentally healthy and those who were languishing. A stepwise regression analysis revealed that supportive campus environment was the strongest predictor of student mental health. It accounted for 19% of the variance in mental health in males and 20% in females. More specifically, supportive relationships with faculty had the biggest impact on males and quality relationships with peers made the biggest difference for females. Ambler (2007) urges researchers to conduct more research on the mental health status of college students to learn more about what contributes to student flourishing.

Parents as Part of a Supportive Campus Environment

Ambler (2007) found a supportive campus environment to play a significant role in the mental health of college students. As parents of college students are becoming more involved, it is important to explore the role they play in a supportive campus environment and in students' mental health status. Barber et al.'s (2005) four year longitudinal study tracking parenting across development provides reason for concern

that relationships with parents may not have the same positive effect as those with faculty and peers. In their study, parents and adolescents reported on the use of each tripartite parenting behavior during the stages of adolescence. Results showed that levels of parental support remained stable across adolescence, with the exception of a steady decrease of physical affection as adolescents aged. In regards to parental behavior control, which includes limit setting and parental monitoring, parents and adolescents had differing opinions. Adolescents reported that levels of behavioral control remained stable while parents felt they gradually declined. Researchers speculated that adolescents adapt to parents' initial levels of control and maintain their behavior accordingly despite decreases in intentional control on the part of parents. This phenomenon has also been observed in other studies (Hair, Jager et al., 2002). Finally, researchers found that the use of psychological control fluctuated throughout adolescence, peaking every time the adolescent experienced a major transition such as moving from middle school to junior high school and then from junior high to high school. This raises the question of if the same occurs during the high school to college transition.

The findings concerning psychological control call for additional research.

Excessive use of psychological control has been connected with delayed identity development, poor academic performance, increases in parent-child conflict, adjustment difficulties, and internalizing disorder symptoms such as anxiety and symptoms of depression (Barber et al., 2005; Nucci, 2001; Schaffer, 1965; Smetana, Crean, & Campion-Barr, 2005; Steinberg & Dornbusch et al., 1992), especially in populations who are exposed to more risk such as children growing up in lower socio-economic communities (Lins-Dyer, 2005; Nucci & Camino et al., 1996; Smetana, Crean, & Daddis,

2002; Smetana & Daddis, 2002; Turiel, 2005). Barber et al.'s (2005) findings lead to the question of whether or not parental use of psychological control peaks again during the transition from high school to college, creating the same potential for negative outcomes and contributing to the current increase in mental health concerns on college campuses.

Summary

Today the college years are viewed as a period of emerging adulthood during which students enjoy freedom to explore and parents remain involved as their primary source of support. This trend is a natural progression for parents, who were encouraged to be involved throughout their children's elementary and secondary school experiences and who are now concerned about their college students making the most of this significant investment.

Overwhelming amounts of research support the benefits of parental involvement in education during the elementary and secondary years. More research is needed to confirm if these benefits are replicated during the college years. Some research has suggested that they are not, and in fact inappropriate parental involvement may contribute to negative outcomes for students. Parents have been found to exercise the most psychological control, which is linked to negative mental health outcomes, during times of transition such as the transition from high school to college. Ambler (2007) found that a supportive campus environment is the strongest predictor of college student mental health. Increased levels of parental involvement today make parents a component of this environment for many students, and more research is needed to understand the impact parents have on their students' mental health.

III. Methods

Introduction

This study was conducted to examine the mental health status and levels of parental involvement of traditionally aged undergraduate college students. This chapter addresses the (a) Research Design, (b) Population, (c) Procedures, and (d) Measures used in the study. Demographic data, student mental health data, and parental involvement data were collected to address the following research questions:

- 1. What are the characteristics of parental involvement and mental health experienced by a sample of traditionally aged undergraduate students?
 - 1a. What impact does gender have on sampled students' experienced levels of parental involvement and mental health?
 - 1b. What impact does ethnicity have on sampled students' experienced levels of parental involvement and mental health?
 - 1c. What impact does distance from home have on sampled students' experienced levels of parental involvement and mental health?
 - 1d. What impact does being a first generation student have on sampled students' experienced levels of parental involvement and mental health?
 - 1e. What impact does mental health category have on sampled students' experienced levels of parental involvement and mental health?

- 1f. What impact does level of parental involvement have on sampled students' experienced levels of parental involvement and mental health?
- 2. What relationship does mental health category have with each independent variable in a sample of traditionally aged undergraduate students?
 - 2a. Is there a relationship between mental health category and gender?
 - 2b. Is there a relationship between mental health category and ethnicity?
 - 2c. Is there a relationship between mental health category and distance from home?
 - 2d. Is there a relationship between mental health category and experience of being a first generation college student?
 - 2e. Is there a relationship between mental health and level of parental involvement?
- 3. What relationship does level of parental involvement have with each independent variable in a sample of traditionally aged undergraduate students?
 - 3a. Is there a relationship between level of parental involvement and gender?
 - 3b. Is there a relationship between level of parental involvement and ethnicity?
 - 3c. Is there a relationship between level of parental involvement and distance from home?
 - 3d. Is there a relationship between level of parental involvement h and experience of being a first generation college student?

- 4. How does parental involvement impact variability in mental health of traditionally aged undergraduate students?
 - 4a. When gender is included as a moderating variable is the relationship between parental involvement and mental health altered?
 - 4b. When ethnicity is included as a moderating variable is the relationship between parental involvement and mental health altered?
 - 4c. When distance from home is included as a moderating variable is the relationship between parental involvement and mental health altered?
 - 4d. When experience of being a first generation college student is included as a moderating variable is the relationship between parental involvement and mental health altered?

Research Design

An exploratory correlational design was utilized in carrying out this study. Every participant completed the same survey measuring their mental health status and parental involvement. The strengths of this design were the possibilities to study a large sample size, to measure the strength of the relationship, and to determine what aspects of parental involvement if any had the strongest relationship with student mental health by conducting a multiple regression analysis. The weaknesses of this approach were that causation could not be established because "observational independent variables will always be confounded with all other variables that are naturally related to them" (Jaccard & Becker, 1997, p. 248). The variables may only be related by chance and further study will be needed (Jaccard & Becker, 1997; Tuckman, 1999), but this weakness does not

detract from the importance of this exploratory study which provides information about potential contributors to student mental health concerns (Gall, Borg, & Gall 1996).

Parental involvement as measured by Oliver's (n.d.) Survey of Parental Involvement was the primary independent variable and a dependent variable in the study. Several moderating independent variables were also analyzed to determine if an interaction effect exists, altering the relationship between parental involvement and mental health for different populations. These included gender, ethnicity, distance from home, and experience of being a first generation college student. Group-focused research recognizes that different populations have different reactions and allows for analysis of these individualized experiences (Schulenberg & Maggs, 2005; Snyder, 2004). Parents have a tendency to be more involved when their children are going through major transitions where there is perceived risk (Barber et al., 2005). Gender, ethnicity, distance from home, and experience of being a first generation college student all have the potential to affect perceived levels of risk and were therefore included in this study (Barber et al., 2005; Nucci, 2001; Schaffer, 1965; Smetana, Crean, & Campion-Barr, 2005; Steinberg, Darnbusch, & Brown, 1992). Mental Health as measured by Keyes' (2002, 2005, 2007) Mental Health Scales of Subjective Well-Being was the dependent variable in this study.

Population

The target population for this study was traditionally aged undergraduate college students, specifically freshmen. Freshmen were selected as the target population because parental involvement has been found to have the strongest impact on younger students

(Strage & Brandt, 1999). The sample for this study included freshmen at a large research university. Students enrolled in a one hour freshman orientation course were given the opportunity to participate through modified cluster sampling. This was carried out through the researcher only visiting a portion of the courses each semester. All freshmen have the opportunity to take one or more of these classes in the fall and spring semesters. All sections of freshman orientation classes were invited to participate but only those students in classes with instructors who invited the researcher to visit had the opportunity to participate. Through taking this course, each participant received the same opportunities to connect to faculty and resources on campus to assist them in their transition, providing more control for confounding variables than found in the entire freshman class at large.

While random selection was accomplished through students randomly selecting certain sections of the freshmen orientation class and instructors randomly choosing to participate in the study, the sample does have limitations. Conducting the study at a large residential research university limits gerneralizability to students at other types of institutions such as commuter and smaller campuses. Also, students who selected to take these classes may have a disposition towards having more positive mental health or may have more involved parents, who encouraged them to enroll in the course. Limitations in generalizability are permissible due to the exploratory nature of the study aiming to establish if a relationship exists (Ambler, 2007; Gall, Borg, & Gall, 1996).

The participants included 617 undergraduate students enrolled in a freshman orientation class at a large research university. While the freshman orientation classes are reserved for freshman students, upperclassmen that are experiencing academic

Table 1 $Frequencies \ and \ Percentages \ for \ Demographic \ Variables \ (N=588)$

| Variable | f | % | |
|------------------------|-----|------|--|
| Gender | | | |
| Male | 248 | 42.2 | |
| Female | 338 | 57.5 | |
| NR | 2 | 0.3 | |
| Race/Ethnicity | | | |
| African American | 90 | 15.3 | |
| Asian/Pacific Islander | 6 | 1.0 | |
| Caucasian | 464 | 78.9 | |
| Hispanic | 14 | 2.4 | |
| Native American | 3 | 0.5 | |
| Other | 9 | 1.5 | |
| NR | 2 | 0.3 | |
| Distance from Home | | | |
| 0-199 miles | 338 | 57.5 | |
| 200-399 miles | 143 | 24.3 | |
| 400-599 miles | 46 | 7.8 | |
| 600+ miles | 53 | 9.0 | |
| NR | 8 | 1.4 | |
| First Generation | | | |
| Yes | 61 | 10.4 | |
| No | 495 | 84.2 | |
| NR | 32 | 5.4 | |
| Academic Term | | | |
| Fall | 254 | 43.2 | |
| Spring | 334 | 56.8 | |

Note. NR = not reported.

difficulties are also permitted to take these courses. The participants included 588 freshmen and 29 upperclassmen. Only data for those students classified as freshmen were included in the sample. Participants identified their gender, ethnicity, distance the

institution was from their home, and whether or not they were a first generation college student. This information can be found in Table 1.

Procedures

The researcher submitted the research protocol form to the Institutional Review Board of the participating institution and it was approved for this study (Appendix A). Upon receiving approval from the Institutional Review Board to proceed, the researcher contacted every instructor teaching a freshman orientation to gain permission to visit their class and invite their students to participate in this study (Appendix B). In the fall of 2007, eight instructors invited the researcher to visit their classes. Approximately 200 students were invited to participate in the study and 112 submitted completed surveys, providing a 56% response rate. In the Spring semester of 2008, the researcher visited 25 classes and approximately 625 students, 356 of whom participated in the study yielding a 57% response rate. Data was collected again in the Fall semester of 2008 to include more students who were brand new to college. The researcher visited 13 classes including 325 students, and 149 or 46% completed the survey.

For each class meeting the researcher was invited to attend, she arrived approximately 10 minutes early to organize her materials. When class started and the instructor had finished covering his or her agenda, the researcher introduced herself and her study by following the Institutional Review Board approved Survey Administrator Script (Appendix C). The researcher invited students to participate in the study explaining that completing the survey would take approximately 10 minutes. It was stressed that the survey was anonymous and was not a requirement for the class. Also,

students were given the option to take the survey at that time and submit it, take the survey at a later time and mail it in using the provided envelope, or not complete the survey at all. In the state of Alabama, individuals must be 19 years of age to be considered an adult and to participate in research studies without parental consent.

Students under the age of 19 only had the options to complete the survey at a later time once parental consent was attained or to not participate in the study. No surveys were received in the mail and this is included as a limitation in the study. Students who did not participate were provided with a list of questions (Appendix D) to consider in order to prepare them to apply the presentation to their personal lives. The presentation was provided after surveys were submitted and therefore did not have an impact on students' answers.

Once all questions were answered, each student was handed a packet including an Information Letter (Appendix E), Parental Permission/Child Assent Form (Appendix F), addressed envelope, copy of the survey (Appendix G), list of questions to consider, and referral list for the event the a student had mental health questions or concerns after completing the survey (Appendix H).

Measures

Each participant in the study completed a single hard copy survey including demographic questions, Keye's (2002) Mental Health Scales of Subjective Well-Being, and Oliver's (n.d.) Survey of Parental Involvement. Participants were given the option of completing the survey in class while the researcher was present or by completing the survey at a later date and mailing it in through campus mail.

Mental Health Scales of Subjective Well-Being

According to Keyes' (2002), mental health breaks down into three types of symptoms including emotional well-being, psychological well-being, and social well-being. The Mental Health Scales of Subjective Well-Being measure respondents' overall score with 40 items gathering data on each of these sub-scales.

Seven items making up the emotional well-being scale measure positive affect (Keyes, 2002, 2005, 2007). Respondents are asked to rank how much of the time they have felt (a) cheerful, (b) in good spirits, (c) extremely happy, (d) calm and peaceful, (e) satisfied, and (f) full of life over the past thirty days on a five-point Likert scale ranging from one (all the time) to five (none of the time). Each of these items is reverse coded and then summed to determine the score for one of two symptoms of emotional well-being. The other symptom is measured by an eighth item asking respondents to rate their life overall on a scale of 0 (worst possible life overall) to 10 (best possible life overall).

Eighteen items making up the psychological well-being scale measure respondents' level of thriving in their personal lives (Ambler, 2007; Keyes, 2002, 2005, 2007). Psychological well-being is constructed of self acceptance, positive relationships, personal growth, purpose, mastery of environment, and autonomy. Three questions are included to measure each of these dimensions to which respondents indicate the degree to which they agree with statements on a Likert scale ranging from one (strongly agree) to seven (strongly disagree). Ten of the items are reverse coded and then each sub-scale is summed to determine the score for 6 of the 13 symptoms of positive functioning. The questions for each subscale are listed in Table 2.

Table 2

Psychological Well-Being Items

| Sub-scale | Survey Item |
|--------------------------------|---|
| Self-Acceptance | I like most parts of my personality.* When I look at the story of my life, I am pleased with how things have turned out so far.* In many ways I feel disappointed about my achievement in life. |
| Purpose in Life | Some people wander aimlessly through life, but I am not one of them.* I live life one day at a time and don't really think about the future. I sometimes feel as if I've done all there is to do in life. |
| Environmental Mastery | The demands of everyday life often get me down. In general, I feel I am in charge of the situation in which I live.* I am good at managing the responsibilities of daily life.* |
| Positive Relations with others | Maintaining close relationships has been difficult and frustrating for me. People would describe me as a giving person, willing to share my time with others.* I have not experienced many warm and trusting relationships with others. |
| Personal Growth | For me, life has been a continuous process of learning, changing, and growth.* I think it is important to have new experiences that challenge how I think about myself and the world.* I gave up trying to make big improvements or changes in my life a long time ago. |
| Autonomy | I tend to be influenced by people with strong opinions. I have confidence in my own opinions, even if they are different from the way most other people think.* I judge myself by what I think is important, not by the values of what others think is important.* |

^{*}Items must be reverse coded.

Table 3
Social Well-Being Items

| Sub-scale | Survey Item |
|----------------------|--|
| Social Coherence | The world is too complex for me. I cannot make sense of what's going on in the world. I try to think about and understand what could happen next in our country.* |
| Social Integration | I don't feel I belong to anything I'd call a community. I feel close to other people in my community.* My community is a source of comfort.* |
| Social Acceptance | People who do a favor expect nothing in return.* People do not care about other people's problems. I believe that people are kind.* |
| Social Contribution | I have something valuable to give the world.* My daily activities do not create anything worthwhile for my community. I have nothing important to contribute to society. |
| Social Actualization | The world is becoming a better place for everyone.* Society has stopped making progress. Society isn't improving for people like me. |

^{*}Items must be reverse coded.

Fifteen items making up the social well-being scales measure respondents' level of satisfaction with their social lives and environment (Ambler, 2007; Keyes, 2002, 2005, 2007). The questions are related to social interactions and personal growth. Social acceptance, actualization, contribution, coherence, and integration are the sub-scales for this area of well-being and three items are included in the instrument to measure each of them. Again, items, seven for this scale, are reverse coded and then each sub-scale is

summed to determine the score for the remaining five symptoms of positive functioning. The questions for each subscale are listed in Table 3.

Keyes (2007) conducted a reliability analysis when he utilized these scales with middle aged adults to establish internal consistency. He found Cronbach alpha scores of .81 on the Psychological Well-Being and Social Well-Being scales and .91 on the Emotional Well-Being Scale (Keyes, 2007). Ambler (2007) repeated this analysis with her data because she was the first researcher to use the survey with traditionally aged college students. She found similar Cronbach alpha coefficients of .86 on the Emotional Well-Being Scale, .82 of the Social Well-Being Scale, and .80 on the Psychological Well-Being Scale. All forty items yielded a coefficient of .90. The same analysis run for this study yielded Cronbach alpha scores of .86 for the Emotional Well-Being Scale, .81 for the Psychological Well-Being Scale, .83 for the Social Well-Being Scale, and .90 overall. According to Jaccard and Becker (1997), scores of .60 or greater are acceptable for exploratory studies, and scores over .80 are desired for all studies. Following these guidelines, Ambler's (2007) results and the results of this study provide preliminary support for using this scale with college aged students.

Validity of the data generated through this survey is supported by Ambler's confirmatory factor analysis yielding an Adjusted Goodness of Fit Index of .81, which is just below the ideal .90, and Root Mean Square Error of Approximation of .058, which is greater than the ideal minimal value of .05 (Ambler, 2007; Keith, 2006). Content validity is supported through the theoretical foundation of the items on survey.

Following Keyes' (2007) recommendation, mental health was analyzed as a categorical variable using mental health categories (i.e. flourishing, moderate mental

health, and languishing) and as a continuous variable using a calculated mental health score in order to obtain more information and compare results. Each participant in the sample was classified by mental health category based on Keyes (2002) mental health continuum. According to the continuum guidelines, individuals scoring in the upper tertile on one or both of the two emotional well-being scales and a minimum of six of the total 11 functional well-being scales, which includes psychological well-being and social well-being, were classified as flourishing. At the other end of the continuum, individuals scoring in the lower tertile on one or both of the two emotional well-being scales and a minimum of 6 of the 11 functional well-being scales were classified as languishing. All remaining individuals were classified as moderate.

Using Keyes' (2002, 2005, 2007) scales to measure mental health allowed for replication of Ambler's (2007) analysis of experiences of mental health of traditionally aged undergraduate students and allowed for an in depth follow up to her findings of supportive campus environment being the strongest predictor of student flourishing. Parents today are an important component of the campus environment and this study investigated the role parents have on their students and the environment.

Survey of Parental Involvement

Oliver (n.d.) breaks down parental involvement into the following theoretically derived subscales: parental involvement in college choice, parental involvement in student social involvement, parental involvement in student academic involvement, student satisfaction with parental involvement, frequency of contact between students and parents, and frequency of visits with parents. A total of 35 quantitative questions and three qualitative questions make up the instrument to measure parental involvement. All

quantitative questions are answered on a Likert scale ranging from one (strongly disagree) to four (strongly agree) and including zero (not applicable) as an option.

The college choice scale is measured by six items addressing the following: parents assisting with filling out applications, parents writing college essays, parents helping in the college decisions process, parents giving reminders about application deadlines, parents putting pressure on students to attend their alma mater, and parents putting pressure on students to attend college period. Seven items make up the social involvement scale. The items address parents encouraging students to be involved on campus, parents encouraging students to join a fraternity or sorority, parents speaking to students about drinking, parents speaking to students about drugs, parents speaking to students about social pressures, parents encouraging students to live on campus, and parents forcing students to live on campus. Finally, the academic involvement scale consists of six items addressing parent involvement in scheduling classes, parents providing wake up calls, parents reminding students of assignments, parents being knowledgeable about when tests take place, parents attending orientation, and parents asking about grades.

Following the same structure used with Keye's (2002, 2005, 2007) Mental Health Scales of Subjective Well-Being, parental involvement was also analyzed as a categorical variable and a continuous variable. A composite parental involvement score was computed by summing the 19 items making up the college choice, social involvement, and academic involvement scales. Participants were classified by level of parental involvement based on their scores on the subscales. College choice, social involvement, and academic involvement scores were divided into tertiles. Individuals scoring in the

upper tertile on at least two of the three subscales were classified as experiencing high parental involvement. Those scoring in the lower tertile on at least two of the subscales were classified as experiencing low parental involvement. All remaining participants were classified as having medium levels of involvement. This allowed for further analysis of the data.

Table 4
Student Satisfaction with Parental Involvement Items

| Scale | Survey Item |
|-------------------------------|--|
| College Choice | Parent involvement in college choice was positive. Parents helped college choice process. |
| Social Involvement | Parents helped transition from high school activities to college. Parent involvement in college social life helped transition. Parent involvement in college social life hindered transition.* Parent involvement in college social life was positive. |
| Academic college. Involvement | Parents helped transition you from high school course load to Parent involvement in academics was a hinderence.* Parent involvement in academics was positive. Parent involvement in academics improved your GPA. |

^{*}Items must be reverse coded.

An additional 12 items are included on the survey to measure student satisfaction with parental involvement. Table 4 lists the satisfaction items for each scale. Also, one item asks about general satisfaction with parents' overall level of involvement and another about general satisfaction with the academic institution.

The survey also asks about frequency of contact between students and parents. Participants are asked to provide the number of times they email, instant message, talk on the phone, and exchange mail with parents per week. Another set of questions asks participants how many times they travel home, parents come to visit, and they meet in other locations per semester. For both forms of contact, regular communication and in person visits, participants indicate who initiates the contact most often.

Finally, three qualitative questions offer participants an opportunity to comment on their parents' involvement. The first question asks what students would change about their parents' involvement. A second question asks students to describe their relationship with their parents. A final question inquires about how relationships with parents have impacted their college experience. Responses to these questions were itemized by topic to identify common themes.

Oliver (n.d.) is currently using his survey for the first time and therefore, a reliability analysis was run on a sample of surveys (N = 48) collected for this study to estimate reliability of the data. Cronbach Alpha coefficients were calculated for each category of involvement utilizing the items assessing involvement. Moderate reliability scores were found for each category as follows: College Choice r = .650, Social Involvement r = .597, and Academic Involvement r = .493. The 19 total items assessing involvement showed a strong correlation (r = .708). The 12 items assessing satisfaction with parental involvement also showed strong internal consistency (r = .776). A follow up reliability analysis was run the full sample of 588 and the following was found: college choice r = .599, social involvement r = .707, academic involvement r = .590, satisfaction r = .795, and all involvement items r = .773. All reliability scores either

exceeded or were just below the acceptable level for explorative studies (.60) and therefore data for each scale and overall were included in analyses. Future studies of parental involvement may want to add more questions to enhance internal consistency. This is included as a limitation of the study. Content validity is supported again through theoretical foundation of items on the survey.

Parental involvement was analyzed as both a categorical variable using levels of parental involvement (i.e. high, medium, and low) and as a continuous variable using calculated parental involvement scores. Using Oliver's (n.d.) survey to measure parental involvement allowed for replication of his analysis of experiences of parental involvement for traditionally aged undergraduate students and provided valuable data to explore a possible relationship with student mental health.

Analyses

Results were analyzed using a statistical software program. One-way analyses of variance were conducted to examine differences in mental health scores between different subgroups. Multiple analyses of variance were performed to investigate differences in the six subscales of parental involvement between the groups. An alpha level of .05 was observed to determine significance. Scores for mental health, parental involvement, parental involvement in college choice, parental social involvement, parental academic involvement, and student satisfaction with parental involvement were converted to a 100 point scale to allow for comparison. The conversion was calculated by dividing the obtained score by the total possible score for each subscale and the overall scale. Frequency of communication was reported by week while frequency of

visits was reported by semester. Also, a backward multiple regression analysis was carried out to determine the impact parental involvement has on variance in mental health of college students.

Summary

This study examined parental involvement and mental health in traditionally aged undergraduate college students. Each participant in the study was classified as having low, moderate, or high levels of parental involvement and assigned a parental involvement score. They were also classified as languishing, moderately mentally healthy, or flourishing and computed a mental health score. This data was used to examine if a relationship exists between parent involvement and mental health in college students. Qualitative responses were also collected concerning students' perception of their parents' involvement to provide more in depth information about their experience. All data was gathered using a single hard copy survey administered directly to students over the course of three academic semesters.

IV. Results

Introduction

The purpose of this study was to explore levels of mental health and parental involvement in traditionally aged undergraduate college students and relationships between these variables. Gender, ethnicity, distance from home, and experience of being a first generation college student were all considered as independent variables. The study addressed the following research questions: (a) What are the characteristics of parental involvement and mental health experienced by a sample of traditionally aged undergraduate students?; (b) What relationship does mental health status have with each independent variable in a sample of traditionally aged undergraduate students?; (c) What relationship does level of parental involvement have with each independent variable in a sample of traditionally aged undergraduate students?; and (d) How does parental involvement impact variability in mental health of traditionally aged undergraduate students? This chapter presents the results of the study.

Descriptive Statistics

In addition to being assigned a mental health score and a parental involvement score, every participant was identified as having a mental health status (flourishing, moderately mentally healthy, or languishing) and a level of parental involvement (high,

Table 5 $Frequencies \ and \ Percentages \ for \ Levels \ of \ Parental \ Involvement \ (N=588)$

| Variable | f | % | |
|---------------------------------------|------------|--------------|--|
| | | | |
| Level of Parental Involvement | | | |
| Level of Parental Involvement High | 163 | 27.7 | |
| | 163 258 | 27.7 43.9 | |

Table 6
Means and Standard Deviations for Parental Involvement Variables

| Variable | Tot N = | | |
|----------------------|------------|-------|--|
| | Mean | SD | |
| Parental Involvement | 59.96 | 10.37 | |
| College Choice | 57.45 | 13.65 | |
| Social Involvement | 67.72 | 15.12 | |
| Academic Involvement | 45.70 | 10.28 | |
| Satisfaction | 70.16 | 13.38 | |
| Communication | 9.99 | 10.13 | |
| Visits | 7.06 | 4.52 | |

Note. Parental Involvement variables were measured by Oliver's (n.d.) Survey of Parental Involvement. All scores were placed on a 100-point scale.

medium, or low). In the sample, 43.9% (n = 258) experienced medium levels of parental involvement, 28.4% (n = 167) experienced low levels, and 27.7% (n = 163) experienced high levels.

Table 7

Frequencies and Percentages for Mental Health Categories (N = 588)

| Variable | f | % | |
|--|------------------|---------------------|--|
| Mental Health Category Flourishing Moderately Mentally Healthy Languishing | 179 351 58 | 30.4 59.7 9.9 | |

Table 8

Means and Standard Deviations for Mental Health Variables

| Total N = 588 | |
|------------------|--------------------------------|
| Mean | SD |
| 71.10 | 8.32 |
| 57.45 | 13.65 |
| 67.72 | 15.12 |
| 45.70 | 10.28 |
| | N = Mean 71.10 57.45 67.72 |

Note. Mental Health variables are measured by Keyes's (2002) Mental Health Scales of Subjective Well-Being. All scores were placed on a 100-point scale. Sub-scale items were summed to calculate composite mental health scores.

In regards to mental health status, 59.7% (n = 351) were moderately mentally healthy, 30.4% (n = 179) were flourishing, and 9.9% (n = 58) were languishing. Keyes' (2002) survey has been used with college students in one other study conducted by Amber (2007) who found that 67.2% (n = 359) of her sample were moderately mentally healthy, 17.4% (n = 93) were languishing, and 15.4% (n = 82) were flourishing.

Research Question 1

Research Question 1: What are the characteristics of parental involvement and mental health experienced by a sample of traditionally aged undergraduate students?

The results are listed in Table 9. Additional analyses were run to address the question taking moderating variables into account.

The mean mental health score for the entire sample was 71.10 (SD = 8.32), and the mean parental involvement score was 59.96 (SD = 10.37). Means and standard deviations were also calculated for parental involvement variables and include the following: parental involvement in college choice (M = 57.45, SD = 13.65), parental social Involvement (M = 67.72, SD = 15.12), and parental academic involvement (M = 45.70, SD = 10.28). The mean score for students' satisfaction with their parents' involvement was 70.16 (SD = 13.38). Students communicated with their parents via email, instant messenger, telephone, and/or mail an average of 9.99 (SD = 10.13) times per week with 38.4% initiating the communication the majority of the time themselves, 33.9% having the communication initiated by parents the majority of the time, and 27.7% having equal initiation as parents. Participants saw their parents either at school, at home, or in an alternate location as average of 7.06 (SD = 4.52) times per semester with

15.5% initiating the visits the majority of the time themselves, 77.8% having the visits initiated by parents the majority of the time, and 6.7% having equal initiation as parents. When responses from all 588 participants were included the mean number of visits was 8.85 with a standard deviation of 17.13. Due to the extremely high standard deviation, outliers were excluded from the analysis and only scores falling within 1 standard deviation from the mean (n = 552) were considered in each analyses of visits.

Table 9

Mental Health and Parental Involvement Continuous Variables

| Variable | Total N = 588 | 3 | | |
|-----------------------------------|------------------|-------|-------------|--|
| | Mean | SD | Reliability | |
| Mental Health | 71.10 | 8.32 | .90 | |
| Parental Involvement ₂ | 59.96 | 10.37 | .708 | |
| College Choice | 57.45 | 13.65 | .599 | |
| Social Involvemen | et 67.72 | 15.12 | .707 | |
| Academic Involve | ment 45.70 | 10.28 | .590 | |
| Satisfaction | 70.16 | 13.38 | .795 | |
| Communication | 9.99 | 10.13 | .191 | |
| Visits | 7.06 | 4.52 | .265 | |

¹ Mental Health Score is the sum of all items on the Subjective Well-Being Scales. Scores were weighted based on Keyes' (2002) factor analysis then calculated on a 100-point scale.

² Parental Involvement variables were measured by Oliver's (n.d.) Survey of Parental Involvement. Scores were placed on a 100-point scale.

Research Question 1a: What impact does gender have on sampled students' experienced levels of parental involvement and mental health?

Two hundred and forty-eight male and 338 female traditionally aged undergraduate freshmen completed the survey.

Mental Health - The average mental health score for males was 69.78 (SD = 9.33) and for females was 72.05 (SD = 7.38). A One Way Analysis of Variance (ANOVA) was run to determine if significant differences in mental health scores were present between male and females. An alpha level of .05 was utilized to determine significance. The test revealed females scored significantly higher than males (F(1,570) = 11.278, p = .001).

Parental Involvement - Males' overall average parental involvement score was 59.00 (SD = 9.79) while females' average overall score was 60.64 (SD = 10.72). Descriptive statistics revealed that females had higher means on five of the six parent involvement subscales. More specifically, females reported greater parental involvement in terms of social, academic, communication, visits, and satisfaction while males reported higher levels for college choice. Group means and standard deviations are summarized in Table 10.

A multivariate analysis of variance test was run to determine if males and females differenced significantly in regards to parental involvement. Again an alpha level of .05 was utilized to determine significance. The test revealed a Wilks' Lamba of .899 (p < .001) with females scoring significantly higher on parent academic involvement, parent social involvement, student satisfaction with parental involvement, frequency of

communication, and frequency of visits. Males had significantly more parental involvement in college choice.

Table 10

Parental Involvement Continuous Variables by Gender

| Variable | Male N = 2 | | Fema | | MANOVA I | Results |
|----------------------|---------------|-------|-------|-------|----------|---------|
| | Mean | SD | Mean | SD | F | p |
| College Choice | 58.62 | 12.90 | 56.55 | 14.17 | 4.215 | .041* |
| Social Involvement | 65.49 | 14.31 | 69.34 | 15.43 | 8.148 | .002* |
| Academic Involvement | nt 44.27 | 10.30 | 46.76 | 10.15 | 8.148 | .004* |
| Satisfaction | 67.42 | 13.21 | 72.10 | 13.20 | 17.565 | <.001** |
| Communication | 7.73 | 7.72 | 11.61 | 11.31 | 22.307 | <.001** |
| Visits | 6.06 | 3.81 | 7.84 | 4.85 | 17.597 | <.001** |

^{*} *p* < .05, ** *p* < .001

Parental Involvement variables are measured by Oliver's (n.d) Survey of Parental Involvement. Scores were placed on a 100-point scale.

Research Question 1b: What impact does race/ethnicity have on sampled students' experienced levels of parental involvement and mental health?

Four hundred and sixty-four Caucasian and 122 Non-Caucasian traditionally aged undergraduate freshmen completed the survey.

Mental Health - The average mental health score for Caucasian students was 71.18 (SD = 8.30) and for Non-Caucasian students was 70.73 (SD = 8.49). An ANOVA revealed no significant differences.

Table 11

Parental Involvement Continuous Variables by Ethnicity

| Mean S | SD | Mean | n SD | F | |
|----------|------------------------------------|--|--|---|--|
| | | | . 50 | I' | p |
| 58.44 | 13.26 | 53.44 | 14.46 | 13.642 | <i>p</i> < .001** |
| 68.57 | 14.44 | 64.17 | 16.99 | 5.539 | .019* |
| t 45.747 | 10.12 | 45.35 | 10.87 | .531 | .467 |
| 70.21 | 13.20 | 69.67 | 13.97 | .003 | .953 |
| 9.67 | 10.03 | 10.97 | 10.42 | 2.141 | .144 |
| 6.89 | 4.26 | 7.77 | 5.38 | 3.384 | .066 |
| | 68.57 t 45.747 70.21 9.67 | 68.57 14.44 45.747 10.12 70.21 13.20 9.67 10.03 | 68.57 14.44 64.17 45.747 10.12 45.35 70.21 13.20 69.67 9.67 10.03 10.97 | 68.57 14.44 64.17 16.99 45.747 10.12 45.35 10.87 70.21 13.20 69.67 13.97 9.67 10.03 10.97 10.42 | 68.57 14.44 64.17 16.99 5.539 445.747 10.12 45.35 10.87 .531 70.21 13.20 69.67 13.97 .003 9.67 10.03 10.97 10.42 2.141 |

^{*} *p* < .05, ** *p* < .001

Parental Involvement – Caucasian students' overall average parental involvement score was 60.57 (SD = 9.92) while Non-Caucasian students' average overall score was 57.32 (SD = 11.57). Caucasian students reported greater parental involvement in terms

Mental Health Score is the sum of all items on the Subjective Well-Being Scales. Scores were weighted based on Keyes' (2002, 2005) factor analysis then calculated on a 100-point scale.

² Parental Involvement variables are measured by Oliver's Survey of Parental Involvement. Scores were placed on a 100-point scale.

of college choice, social, academic, and satisfaction while Non-Caucasian students reported higher levels for communication and visits. Group means and standard deviations are summarized in Table 11.

A multivariate analysis of variance test was run to determine if Caucasian and Non-Caucasian students differenced significantly in regards to parental involvement. Again an alpha level of .05 was utilized to determine significance. The test revealed a Wilks' Lamba of .954 (p < .001) with Caucasian students scoring significantly higher on parental involvement in college choice and social involvement.

Research Question 1c: What impact does distance from home have on sampled students' experienced levels of parental involvement and mental health?

Amongst the students who completed the survey, 338 were from fewer than 200 miles and 242 were from 200 or more miles away from the location of campus.

Mental Health - The average mental health score for those closer to home was 71.32 (SD = 7.32) and for those further away was 70.71 (SD = 9.60). Results from an ANOVA indicated no significant differences in mental health scores based on distance from home.

Parental Involvement - Students from fewer than 200 miles away scored an average parental involvement score of 59.79 (SD = 11.28) while students living further than 200 miles had an average score of 60.13 (SD = 9.03). Students from less than 200 miles away reported greater parental involvement in terms of college choice, academic, satisfaction, communication, and visits while students from more than 200 miles away reported higher levels for social. Group means and standard deviations are summarized in Table 12.

A multivariate analysis of variance test was executed to determine if distance from home significantly impacted parental involvement. Again an alpha level of .05 was utilized to determine significance. The test revealed a Wilks' Lamba of .843 (p < .001) with students closer to home scoring significantly higher on parental involvement in college choice, academics, and visits.

Table 12

Parental Involvement Continuous Variables by Distance from Home

| Variable | < 200 N N = 33 | | > 200 Miles N = 242 | MANO' | VA Results |
|----------------------|-------------------|-------|------------------------|--------|------------|
| | Mean | SD | Mean SD | F | p |
| College Choice | 58.10 | 14.53 | 56.82 12.16 | 4.854 | .028* |
| Social Involvement | 66.65 | 15.72 | 68.98 14.08 | .855 | .355 |
| Academic Involvement | ent 45.90 | 10.99 | 45.37 9.32 | 4.161 | .042* |
| Satisfaction | 70.50 | 16.68 | 69.77 12.84 | 1.123 | .290 |
| Communication | 10.14 | 10.26 | 9.67 10.00 | .047 | .829 |
| Visits | 8.62 | 4.71 | 5.03 3.32 | 27.909 | <.001** |
| | | | | | |

^{*} *p* < .05, ** *p* < .001

Mental Health Score is the sum of all items on the Subjective Well-Being Scales. Scores were weighted based on Keyes' (2002, 2005) factor analysis then calculated on a 100-point scale.

² Parental Involvement variables are measured by Oliver's Survey of Parental Involvement. Scores were placed on a 100-point scale.

Research Question 1d: What impact does being a first generation college student have on sampled students' experienced levels of parental involvement and mental health?

Sixty-one participants identified themselves as first generation college students while 495 identified themselves as not.

Table 13

Parental Involvement Continuous Variables by First Generation Experience

| Variable | First Generation $N = 338$ | | Not First Generation $N = 242$ | MANOVA Results | | |
|--------------------|----------------------------|-------|--------------------------------|----------------|-------|--|
| | Mean | SD | Mean SD | F | p | |
| College Choice | 54.17 | 15.65 | 57.69 13.48 | 2.488 | .115 | |
| Social Involvement | 63.17 | 18.36 | 68.29 14.82 | 3.274 | .071 | |
| Academic Involveme | ent 44.15 | 9.67 | 46.01 10.35 | 2.047 | .153 | |
| Satisfaction | 69.20 | 15.25 | 70.37 13.02 | .047 | .828 | |
| Communication | 8.98 | 5.65 | 9.98 10.48 | .155 | .694 | |
| Visits | 9.11 | 5.94 | 6.93 4.30 | 11.880 | .001* | |

^{*} p < .05, ** p < .001

Mental Health - The average mental health score for first generation college students was 72.38 (SD = 6.86) and for students who are not first generation was 70.83

Mental Health Score is the sum of all items on the Subjective Well-Being Scales. Scores were weighted based on Keyes' (2002, 2005) factor analysis then calculated on a 100-point scale.

² Parental Involvement variables are measured by Oliver's Survey of Parental Involvement. Scores were placed on a 100-point scale.

(SD = 8.50). Results from an ANOVA indicated no significant differences in mental health between first generation and non-generation participants.

Parental Involvement - First generation students scored an average parental involvement score of 56.64 (SD = 12.07) while non-first generation students had an average score of 60.36 (SD = 10.23). Non-first generation students reported greater parental involvement in terms of college choice, social, academic, satisfaction, and communication while first generation students reported more frequent visits with parents. Group means and standard deviations are summarized in Table 13.

A multivariate analysis of variance test was run to determine if first generation and non-first generation students differenced significantly in regards to parental involvement. Again an alpha level of .05 was utilized to determine significance. The test revealed a Wilks' Lamba of .964 (p = .005) with non-first generation students scoring significantly higher on visits.

Research Question 1e: What impact does mental health status have on sampled students' experienced levels of parental involvement?

Amongst the students who completed the survey, 179 were identified as flourishing, 351 as having moderate mental health, and 58 as languishing.

Parental Involvement – Flourishing students scored an average parental involvement score of 59.78 (SD = 10.12). Moderately mentally healthy students scored an average score of 60.35 (SD = 10.40) and finally languishing students' average score was 58.06 (SD = 10.91). Flourishing students reported higher scores in terms of social involvement, satisfaction with involvement, communication, and visits. Moderate students reported more academic involvement and languishing students reported more

involvement in college choice. Group means and standard deviations are summarized in Table 14.

Table 14

Parental Involvement Continuous Variables by Mental Health Category

| Variable | Flouris N = 1 | • | Moder N = 3 | | Languish N = 58 | _ |
|----------------------|----------------------|---------|----------------|-------|--------------------|-------|
| | Mean | SD | Mean | SD | Mean | SD |
| Mental Health | 77.12 | 5.50 | 70.19 | 5.98 | 58.03 | 10.34 |
| Parental Involvement | t ₂ 59.78 | 10.13 | 60.35 | 10.40 | 58.06 | 10.91 |
| College Choice | 56.65 | 13.70 | 57.79 | 13.79 | 57.83 | 12.76 |
| Social Involvement | 68.96 | 14.34 | 67.95 | 15.17 | 62.56 | 16.27 |
| Academic Involveme | ent 44.53 | 3 10.22 | 46.32 | 10.38 | 45.61 | 9.73 |
| Satisfaction | 73.22 | 12.75 | 69.87 | 12.98 | 62.35 | 14.49 |
| Communication | 11.15 | 11.34 | 9.46 | 8.49 | 9.58 | 14.44 |
| Visits | 7.36 | 4.16 | 6.98 | 4.62 | 6.64 | 5.02 |

¹ Mental Health Score is the sum of all items on the Subjective Well-Being Scales. Scores were weighted based on Keyes' (2002, 2005) factor analysis then calculated on a 100-point scale.

A multivariate analysis of variance test was run to determine if parental involvement scores differenced significantly based on mental health status. Again an alpha level of .05 was utilized to determine significance. The test revealed a Wilks'

² Parental Involvement variables are measured by Oliver's Survey of Parental Involvement. Scores were placed on a 100-point scale.

Lamba of ..882 (p = < .001). Significant differences were found in social involvement (p = .018) and satisfaction (p < .001). MANOVA results are summarized in Table 15.

Table 15

Multiple Analysis of Variance for Parental Involvement Continuous Variables by Mental Health Category

| | MANOVA | A Results |
|----------------------|--------|-----------|
| | F | p |
| College Choice | 1.172 | .322 |
| Social Involvement | 2.749 | .018* |
| Academic Involvement | 1.682 | .137 |
| Satisfaction | 5.694 | <.001** |
| Communication | .116 | .989 |
| Visits | .759 | .579 |

* *p* < .05, ** *p* < .001

A Tukey post hoc analysis revealed that flourishing students (p = .032) and moderately mentally healthy students (p = .048) both had significantly higher amounts of social involvement than languishing students. Also, flourishing students were significantly (p = .015) more satisfied with their parents' involvement than moderately mentally healthy students, and moderately mentally healthy students were significantly (p = .001) more satisfied with their parents' involvement than students who were languishing.

Research Question 1f: What impact does level of parental involvement have on sampled students' experienced levels of mental health?

Amongst the students who completed the survey, 163 were identified as experiencing high levels of parental involvement, 258 as experiencing medium levels, and 167 experiencing low levels.

Mental Health - The average mental health score was 71.70 (SD = 7.82) for students with high involvement, 70.55 (SD = 8.81) for students with medium levels, and 71.37 (SD = 8.00) for those with low levels. A one-way ANOVA was executed to determine if significant differences were present amongst students with different levels of parental involvement. An alpha level of .05 was utilized to determine significance and no significant differences in mental health were found based on level of parental involvement the students were classified in.

Research Question 2: What relationship does mental health category have with gender, ethnicity, distance from home, experience of being a first generation college student, and level of parental involvement in a sample of traditionally aged undergraduate students?

In order to determine if observed frequencies differed from expected frequencies of flourishing, moderately mentally healthy, and languishing students, chi square tests for independence were executed to examine differences by gender, ethnicity, distance from home, experience of being a first generation student, and level of parental involvement. The results for each of these analyses are summarized in Table 16. No significant differences in the number of participants classified in each of the mental health categories were found based on any of the variables studied.

Research Question 2a: What relationship does mental health category have with gender in a sample of traditionally aged undergraduate students?

Table 16

Crosstabulation of Mental Health Category and Independent Variables

| Variable | X ² | p |
|----------------------------------|----------------|------|
| Gender | 2.30 | .317 |
| Ethnicity | 1.20 | .600 |
| Distance | 2.84 | .242 |
| Generation | .935 | .627 |
| Level of Parental Involvement | 2.493 | .646 |

In order to find out if the observed frequencies of flourishing, moderately mentally healthy, and languishing students differed significantly from the expected frequencies based on gender, a chi square test for independence was run. The test showed that 27.0% of the men were flourishing, 62.5% were moderately mentally healthy, and 10.5% were languishing. Out of the females, 32.8% were flourishing, 57.7% were moderately mentally health, and 9.5% were languishing. The proportion of males and females in each mental health category did not differ significantly; there is no relationship between mental health category and gender, X^2 (2, N = 586) = 2.30, p = .317. Research Question 2b: What relationship does mental health category have with ethnicity in a sample of traditionally aged undergraduate students?

In order to find out if the observed frequencies of flourishing, moderately mentally healthy, and languishing students differed significantly from the expected frequencies based on ethnicity, a chi square test for independence was run. The test showed that 30.6% of the Caucasian students were flourishing, 60.1% were moderately mentally healthy, and 9.3% were languishing. Out of the non-Caucasian students, 30.3% were flourishing, 57.4% were moderately mentally health, and 12.3% were languishing. The proportion of Caucasian and Non-Caucasian students in each mental health category did not differ significantly; there is no relationship between mental health category and ethnicity, X^2 (2, N = 586) = 1.20, p = .600.

Research Question 2c: What relationship does mental health category have with distance from home in a sample of traditionally aged undergraduate students?

In order to find out if the observed frequencies of flourishing, moderately mentally healthy, and languishing students differed significantly from the expected frequencies based on distance from home, a chi square test for independence was run. The test showed that 29.6% of the students who were from less than 200 miles away were flourishing, 62.1% were moderately mentally healthy, and 8.3% were languishing. Out of the students who were more than 200 miles away, 31.4% were flourishing, 56.6% were moderately mentally health, and 12.0% were languishing. The proportion of students from the two groups did not differ significantly; there is no relationship between mental health category and distance from home, X^2 (2, N = 580) = 2.84, p = .242. Research Question 2d: What relationship does mental health category have with experience of being a first generation college student in a sample of traditionally aged

undergraduate students?

In order to find out if the observed frequencies of flourishing, moderately mentally healthy, and languishing students differed significantly from the expected frequencies based on experience of being a first generation college student, a chi square test for independence was run. The test showed that 31.1% of first generation students were flourishing, 62.3% were moderately mentally healthy, and 6.6% were languishing. Out of the students who were not first generation, 29.7% were flourishing, 5.8% were moderately mentally health, and 10.5% were languishing. The proportion of first generation students and non-first generation students did not differ significantly; there is no relationship between mental health category and experience of being a first generation student, X^2 (2, N = 556) = .935, p = .627.

Research Question 2e: What relationship does mental health category have with level of parental involvement in a sample of traditionally aged undergraduate students?

In order to find out if the observed frequencies of flourishing, moderately mentally healthy, and languishing students differed significantly from the expected frequencies based on level of parental involvement, a chi square test for independence was run. The test showed that 30.5% of students with low parental involvement were flourishing, 58.7% were moderately mentally healthy, and 10.8% were languishing. Out of the students with medium parental involvement, 27.9% were flourishing, 61.6% were moderately mentally health, and 10.5% were languishing. Out of the students with high parental involvement, 34.3% were flourishing, 57.7% were moderately mentally health, and 8.0% were languishing. The proportion of students with each level of parental involvement did not differ significantly; there is no direct relationship between mental health category and level of parental involvement, X^2 (4, N = 588) = .2.493, p = .646.

Research Question 3: What relationship does level of parental involvement have with gender, ethnicity, distance from home, and experience of being a first generation college student in a sample of traditionally aged undergraduate students?

In order to determine is observed frequencies differ from expected frequencies for levels of high, medium, and low parental involvement, chi square tests for independence were run and the results are summarized in Table 17.

Table 17

Crosstabulation of Level of Parental Involvement and Independent Variables

| Variable | X ² | p |
|------------|----------------|-------|
| Gender | 8.92 | .012* |
| Ethnicity | 8.49 | .014 |
| Distance | 1.106 | .595 |
| Generation | 4.775 | .595 |

p < .05

Research Question 3a: What relationship does level of parental involvement have with gender in a sample of traditionally aged undergraduate students?

In order to find out if the observed frequencies of students with high, medium, and low levels of parental involvement differed significantly from the expected frequencies based on gender, a chi square test for independence was run. The test showed that 21.4% of the men experienced high levels of parental involvement, 49.2% experienced medium levels, and 29.4% experienced low levels. Out of the females,

32.3% experienced high levels of parental involvement, 40.2% experienced medium levels, and 27.5% experienced low levels. The proportion of males and females in each mental health category did differ significantly; there is a relationship between level of parental involvement and gender, X^2 (2, N = 586) = 8.92, p = .012. A higher percentage of females had high parental involvement. On the other end of the continuum a higher percentage of males had medium and low parental involvement. The analysis shows that parents are more involved in their female students' lives than their male students' lives. An additional backward regression analysis was run later to determine the amount of variance in mental health that parental involvement is responsible for in both males and females.

Research Question 3b: What relationship does level of parental involvement have with ethnicity in a sample of traditionally aged undergraduate students?

In order to find out if the observed frequencies of students with high, medium, low parental involvement differed significantly from the expected frequencies based on ethnicity, a chi square test for independence was run. The test showed that 30.6% of the Caucasian students experienced high levels of parental involvement, 60.1% experienced medium levels, and 9.3% experienced low levels. Out of the Non-Caucasian students, 32.3% experienced high levels of parental involvement, 40.2% experienced medium levels, and 27.5% experienced low levels. The proportion of Caucasian and Non-Caucasian students in each mental health category did differ significantly; there is a relationship between mental health category and ethnicity, X^2 (2, N = 586) = 8.49, p = .014. A higher percentage of Caucasian students had high and medium parental

involvement and a higher percentage of Non-Caucasian students had low parental involvement.

Research Question 3c: What relationship does level of parental involvement have with distance from home in a sample of traditionally aged undergraduate students?

In order to find out if the obse-rved frequencies of students with high, medium, and low parental involvement differed significantly from the expected frequencies based on distance from home, a chi square test for independence was run. The test showed that 27.8% of the students who were from less than 200 miles away experience high levels of parental involvement, 42.3% experienced medium levels, and 29.9% experienced low levels. Out of the students who were more than 200 miles away, 27.3% experienced high levels of parental involvement, 46.3% experienced medium levels, and 26.4% experienced low levels. The proportion of students from fewer than 200 miles away and students from more than 200 miles away did not differ significantly; there is no relationship between level of parental involvement and distance from home, X^2 (2, N = 580) = 1.106, p = .575. The advancement of technology and ease of transportation has contributed to parental involvement being possible regardless of physical location. Research Question 3d: What relationship does level of parental involvement have with experience of being a first generation college student in a sample of traditionally aged undergraduate students?

In order to find out if the observed frequencies of students with high, medium, and low levels of parental involvement differed significantly from the expected frequencies based on experience of being a first generation college student, a chi square test for independence was run. The test showed that 16.4% of first generation college

students were flourishing, 49.2% were moderately mentally healthy, and 34.4% were languishing. Out of the students who were not in the first generation in their families to attend college, 29.7% were flourishing, 42.2% were moderately mentally health, and 28.1% were languishing. The proportion of first generation college students and non-first generation college students did not differ significantly; there is no relationship between level of parental involvement and experience of being a first generation college student, X^2 (2, N = 556) = 4.775, p = .092. Parents will engage in involvement in their students' college experience as they see fit regardless of whether or not they attended college themselves.

Research Question 4: How does parental involvement predict variability in mental health of traditionally aged undergraduate students?

A backward multiple regression analysis was run to determine if parental involvement variables predict variance in mental health. The following independent variables were considered: (a) parental involvement in college choice score, (b) parental involvement in student social involvement score, (c) parental involvement in student academic involvement score, (d) student satisfaction with parental involvement, (e) frequency of communication with parents, and (f) frequency of visits with parents. Following Ambler's (2007) recommendation, Variance Inflation Factor values and magnitude of effect scores are also reported in Table 18. According to Keith (2006), VIF values over five signify an issue with multicollinearity. In terms of effect sizes, β 's under .05 are too small to consider while β 's over .25 signify a large effect and those falling in between have a moderate effect (Keith, 2006).

Table 18
Summary of Backward Multiple Regression Analysis

| Variable | В | SE B | β | t | Sig. | VIF |
|----------------------|------|------|------|--------|--------|-------|
| Full Model | | | | | | |
| run Woder | | | | | | |
| College Choice | 052 | .026 | 093 | -2.009 | .045 | 1.226 |
| Social Involvement | .018 | .024 | .036 | .751 | .453 | 1.317 |
| Academic Involvement | 110 | .031 | 170 | -3.573 | < .001 | 1.302 |
| Satisfaction | .160 | .028 | .277 | 5.799 | < .001 | 1.309 |
| Visits | 109 | .073 | 011 | 265 | .791 | 1.070 |
| Communication | .083 | .033 | .109 | 2.523 | .012 | 1.061 |
| Restricted Model | | | | | | |
| College Choice | 049 | .026 | 087 | -1.922 | .005 | 1.187 |
| Academic Involvement | 107 | .030 | 166 | -3.528 | < .001 | 1.265 |
| Satisfaction | .165 | .026 | .286 | 6.343 | <.001 | 1.164 |
| Communication | .086 | .032 | .112 | 2.637 | .009 | 1.033 |

The entire sample was analyzed first, and the results are displayed in Tables 18 and 19. The analysis concluded the best model for predicting mental health included four of the six variables studied: student satisfaction with parental involvement, parental involvement in student academic involvement, frequency of communication with parents, and parental involvement in college choice, $R^2 = .094$, F(4, 523) = 13.850, p < .001.

VIF scores were all under two, raising no concern of multicollinearity. Effect sizes varied with satisfaction having a large effect (β = .286), parental academic involvement (β = -.166) and communication with parents (β = .122) having a moderate effect, and parental involvement in college choice (β = .87) having a small but meaningful effect on students' mental health. All the variables together accounted for 9% of the variance in students' mental health score, indicated by an adjusted R² of .089.

Table 19

Model Summary for Backward Multiple Regression Analysis

| Model | R | R² | Adj. R² | $R^2\Delta$ | F | Sig. |
|------------|------|------|---------|-------------|--------|--------|
| Full | .312 | .098 | .087 | .098 | 9.322 | < .001 |
| Restricted | .311 | .096 | .089 | .001 | 13.850 | < .001 |

a. Predictors: (Constant), College Choice, Social, Academic, Satisfaction, Visits, Communication

The only variable that displayed significant differences in mental health score in previous analyses was gender (t (584) = -3.281, p = .001) with females (M = 72.05, SD = 7.38) having significantly higher mental health scores than males (M = 69.78, SD = 9.33). Separate backward multiple regression analyses were run for females and males to determine which parental involvement variables contributed to mental health for each subgroup and to calculate the impact parental involvement had on variance in mental health for each subgroup individually. Results for these analyses can be found in Table 20 and Table 21.

b. Predictors: (Constant), College Choice, Academic, Satisfaction, Communication

Table 20
Summary of Backward Multiple Regression Analysis in Females

| Variable | В | SE B | β | t | Sig. | VIF |
|----------------------|------|------|------|--------|------|-------|
| | | | | | | |
| Full Model | | | | | | |
| College Choice | 059 | .032 | 122 | -1.839 | .067 | 1.286 |
| Social Involvement | 025 | .030 | 054 | 840 | .402 | 1.426 |
| Academic Involvement | 144 | .039 | 231 | -3.710 | .000 | 1.354 |
| Satisfaction | .227 | .037 | .398 | 6.210 | .000 | 1.430 |
| Visits | 094 | .084 | 063 | -1.120 | .264 | 1.088 |
| Communication | .037 | .035 | .057 | 1.039 | .300 | 1.042 |
| Restricted Model | | | | | | |
| Restricted Model | | | | | | |
| College Choice | 064 | .031 | 123 | -2.060 | .040 | 1.235 |
| Academic Involvement | 148 | .038 | 238 | -3.881 | .000 | 1.316 |
| Satisfaction | .212 | .033 | .370 | 6.402 | .000 | 1.168 |

A linear combination of parental involvement in college choice, academic involvement, and student satisfaction with parental involvement were included in the restricted model for predicting mental health in females, $R^2 = .146$, F(3, 301) = 17.025, p < .001. VIF scores were all under 2.5, raising no concern of multicollinearity. Student satisfaction with parental involvement ($\beta = .370$) had a large effect on female students'

mental health while parental involvement in college choice (β = .031) and academic involvement (β = -.238) had moderate effects. The variables together accounted for 14% of the variance in female students' mental health score, indicated by an adjusted R² of .138.

Table 21

Model Summary for Backward Multiple Regression Analysis in Females

| Model | R | R² | Adj. R² | $R^2\Delta$ | F | Sig. |
|------------|------|------|---------|-------------|--------|--------|
| Full | .392 | .154 | .136 | .154 | 8.930 | < .001 |
| Restricted | .383 | .146 | .138 | .003 | 17.025 | < .001 |

a. Predictors: (Constant), College Choice, Social, Academic, Satisfaction, Visits, Communication

A separate backward regression analysis was run to determine the extent to which parental involvement impacted variance in mental health in males.

A linear combination of satisfaction with parental involvement and communication with parents were included in the restricted model for predicting mental health in males, $R^2 = .060$, F(2,219) = 6.979, p = .001. VIF scores were all under 2.5, raising no concern of multicollinearity. Both satisfaction with parental involvement ($\beta = .129$) and communication ($\beta = .199$) had moderate effects. The variables together accounted for 5% of the variance in male students' mental health score, indicated by an adjusted R^2 of .052.

b. Predictors: (Constant), College Choice, Academic, Satisfaction

Table 22
Summary of Backward Multiple Regression Analysis in Males

| Variable | В | SE B | β | t | Sig. | VIF |
|----------------------|------|------|------|--------|------|-------|
| Full Model | | | | | | |
| College Choice | 002 | .46 | 004 | 053 | .958 | 1.267 |
| Social Involvement | 043 | .039 | .079 | 1.094 | .275 | 1.215 |
| Academic Involvement | 077 | .050 | 113 | -1.560 | .120 | 1.211 |
| Satisfaction | .082 | .043 | .138 | 1.925 | .056 | 1.177 |
| Visits | .082 | .137 | .040 | .594 | .553 | 1.032 |
| Communication | .216 | .075 | .194 | 2.876 | .004 | 1.047 |
| Restricted Model | | | | | | |
| Satisfaction | .077 | .039 | .129 | 1.957 | .052 | 1.006 |
| Communication | .222 | .074 | .199 | 3.017 | .003 | 1.006 |

Summary

Descriptive statistics, independent sample t-tests, one-way analyses of variance, chi square tests for independence, multiple analyses of variance, and backward multiple regression analyses were all executed to analyze the characteristics of parental involvement and mental health in the sample. Both parental involvement and mental health were analyzed as continuous and categorical variables.

Table 23

Model Summary for Backward Multiple Regression Analysis in Males

| Model | R | R² | Adj. R² | $R^2\Delta$ | F | Sig. |
|------------|------|------|---------|-------------|-------|------|
| Full | .275 | .076 | .050 | .076 | 2.908 | .010 |
| Restricted | .246 | .060 | .052 | 009 | 6.979 | .001 |

- a. Predictors: (Constant), College Choice, Social, Academic, Satisfaction, Visits, Communication
- b. Predictors: (Constant), Satisfaction, Communication

Parental involvement was further broken down based on Oliver's (n.d.) Survey of Parental Involvement into parental involvement in college choice, academic involvement, social involvement, student satisfaction with parental involvement, frequency of communication with parents, and frequency of visits with parents. The average parental involvement score for the entire sample was 59.96 (SD = 10.37). The most parental involvement occurred in students' social lives (M = 67.72, SD = 15.12) followed by college choice (M = 57.45, SD = 13.65) and academic involvement (M = 45.7, SD = 13.65) 10.28) for the entire sample and each subgroup studied. Student satisfaction had the largest effect size (β = .286). Participants communicated with their parents an average of 9.99 (SD = 10.13) times per week with 38.4% initiating the communication the majority of the time themselves, 33.9% having the communication initiated by parents the majority of the time, and 27.7% having equal initiation as parents. Participants visited an average of 7.06 (SD = 4.52) times per semester with 15.5% initiating the visits the majority of the time themselves, 77.8% having the visits initiated by parents the majority of the time, and 6.7% having equal initiation as parents.

Significant differences in parental involvement were found based on gender, ethnicity, distance from home, experience of being a first generation college student, and mental health status. Females reported significantly greater amounts of parental involvement in their social (p = .002) and academic (p = .004) lives, had significantly more contact through communication (p = <.001) and visits (p = <.001) with their parents, and were significantly more satisfied ((p = <.001) with their parents' involvement. On the other hand males had significantly more parental involvement in their college choice process (p = .041). Caucasian students reported significantly higher amounts of parental involvement in their college choice process (p < .001) and their social lives (p = .019). Students from fewer than 200 miles away reported significantly higher scores on parental involvement in college choice (p = .028), academics (p = .042), and visits (p = <.001). First generation college students visited with parents significantly more often (p = .001) with 90% of the participants initiating the visits the majority of the time. Finally, students identified as flourishing (p = .032) and moderately mentally healthy (p = .048) experienced significantly more parental involvement in their social lives than students identified as languishing. Also, flourishing students were significantly more satisfied (p = .015) with parental involvement than moderately mentally healthy students who were significantly more satisfied (p = < .001) than languishing students.

Mental Health was further broken down based on Keyes' (2002) Scales of Subjective Well-Being into emotional, social, and psychological well-being. The average mental health score for the entire sample was 71.10 (SD = 8.32). Significant differences in mental health were found based on gender with females scoring significantly higher (p = .001). A backward multiple regression analysis revealed that 9% of the variance in

mental health for the entire sample was due to parental involvement in college choice and academics, student satisfaction with parental involvement, and communication with parents. Parental involvement in college choice and academics and satisfaction accounted for 14% of the variance in mental health scores for females. Finally, satisfaction and communication contributed to 5% of the variance in mental health scores for males.

V. Conclusions and Recommendations

This chapter provides an account of the conclusions and recommendations based on the findings for the following questions:

- 1. What are the characteristics of parental involvement and mental health experienced by a sample of traditionally aged undergraduate students?
- 2. What relationship does mental health category have with each independent variable in a sample of traditionally aged undergraduate students?
- 3. What relationship does level of parental involvement have with each independent variable in a sample of traditionally aged undergraduate students?
- 4. How does parental involvement impact variability in mental health of traditionally aged undergraduate students?

Conclusions

Parental involvement contributed to 9% of the variance in mental health in this study. More specifically it was parental involvement in the college choice process, involvement in academics, students' satisfaction with their parents' involvement, and communication that mattered. Parental involvement was especially impactful for females, contributing to 14% of the variance in their mental health scores. Only a third of the participants in the study were classified as flourishing and 10% were classified as

languishing. The findings of this study provide further evidence that a need still exists to find ways to address inadequate mental health on college campuses and implicates that parental involvement is one variable to consider. The quantitative results and qualitative comments from this study provide insight into positive parental involvement that can contribute to flourishing in college students. Additionally, results uncovered areas for further research. Significant findings are outlined in this section.

Parental involvement in the college experience begins with the college search process. This facet of involvement was actually one of the three areas that contributed to variance in mental health. In regards to college choice, participants made contradicting comments. Some wished their parents had been less involved and "made [them] take on the responsibility of filling out applications on time and remembering to write essays instead of doing it for them." Others wished their parents had been more involved and "known more about their school before [they] enrolled there." According to these comments, students want parents to be knowledgeable about the institutions their students are considering but hold the student accountable for the application process.

Academic involvement was another contributing variable to mental health. Again students made contradicting comments. The majority of students commented that they would like their parents to be more involved in their academics by "anticipat[ing] test dates and encourage[ing] them on those days" and "know[ing] more about the classes [they] are taking." Those who wanted less academic involvement wrote about parents logging into their email and picking their classes. A common concern that was mentioned was parents understanding the difference between high school and college and not expecting the same performance in college. One student wrote "I wish they could

understand the change in academic achievement from high school to college." Another wrote she wished her mom would "understand the fact that tests in college are harder than high school and that's why [she] doesn't score very high."

Social involvement did not directly contribute to variance in mental health, but every subgroup of students reported parents being most involved in this area. It was not surprising that social involvement scores were high considering parents' tendency to increase amounts of psychological control during times of transition such as going to college (Barber, Maughen & Olsen, 2005; Nucci, 2001; Nucci, Hasabe, & Lins-Dyer, 2005; Smetana, 1995; Smetana, Crean, & Campion-Barr, 2005; Steinbuerg et al., 1992; Turiel, 2005) and increased parent concerns about student safety and health risks (Bylund, Imes, & Baxter, 2005; Mansfield & Warwick, 2005; Shellenbarger, 2006). However, it is intriguing that social involvement did not significantly contribute to mental health variance. Social involvement is the facet of involvement most closely related to parental psychological control, which when exercised in excess has been linked with negative mental health outcomes including delayed identity development, increased parent-child conflict, and internalizing disorder symptoms such as anxiety and depression (Barber et al., 2005; Nucci, 2001; Schaffer, 1965; Smetana, Crean, & Campion-Barr, 2005; Steinberg, Dornbusch et al., 1992). Also, participants in this study who were classified as flourishing and moderately mentally healthy had significantly higher parental social involvement scores than those who were languishing.

Qualitative comments relating to social involvement provide some insight. Most students commented that they enjoyed sharing with their parents about their social life but when it was on their terms and absent of parent lectures. One student wrote that he

wished his parents wouldn't "be so pushy about asking about the love life." Another student wrote "They were crazy in college and they get mad at me for doing the same things they did." A third student commented "I would change some of the phone calls about certain organizations I should join." Perhaps social involvement is an area that does not change significantly from high school to college and remains a constant variable.

Student satisfaction with parental involvement was found to be the leading factor impacting variance in mental health. When asked on the survey what they would change about their parents' involvement the majority of participants commented that they were satisfied with their parents' involvement and would not change anything. One student wrote "They care about how I'm doing but don't force themselves into my life." Another student commented "They call just enough and aren't overly attentive and involved. They realize that I am independent." A third student wrote "They are great! A perfect balance between parent/friend/teacher."

The final variables were communication and visits with parents. Students commented that they actually would like more contact with their parents including more phone calls, more mail, and more visits. One student wrote "I would change the fact that my parents are four hours away... that way they would be closer if I needed them." Students take comfort in knowing that parents are there to support them and appreciate parents showing that support through phone calls to just listen, visits to say hello, and mail.

Based on the findings of this study, college students appreciate a balance between autonomy support and solicited guidance from parents in all areas of involvement, which are the same principles of precision parenting prescribed for parenting in childhood through adolescence. These findings suggest that parenting with today's college student is not qualitatively different from what parents are accustomed to, supporting the concept of college now representing an extended adolescence. The practical implications for parents are that new variables, especially distance and new academic demands, introduce the need for parents to be creative in finding new ways to communicate and be involved. Maintaining open lines of communication that allow students to express their needs and level of satisfaction with their parents' involvement will contribute to parents having a positive influence on their students' mental health and overall college experience. Implications for professionals are new challenges of preparing students for transitioning into the workforce where they are expected to function independently when they are accustomed to relying on parental involvement.

Recommendations

Based on the findings of this study, the following recommendations are presented:

- It is recommended that higher education admissions offices work with high school guidance offices to make parents partners in their students' college search process. They should ensure that the student is actively carrying out the application process.
- It is recommended that higher education admissions offices and high school guidance offices do more to explain the difference between high school and college academics to assist parents with establishing realistic academic expectations.

- 3. It is recommended that higher education institutions continue to search for ways to appropriately include parents in students' social and academic lives such as invitations to special parent social events or academic award events. Technology should also be utilized to keep parents informed and connected to their students' college experience as this has been a successful strategy in the kindergarten through high school system.
- 4. It is recommended that higher education administrators consider implementing required senior capstone courses that address strategies for establishing independence and practicing self reliance upon graduating from college and entering the workforce.
- 5. It is recommended that college counseling centers and referral coordinators on college campuses include questions to gather information about parental involvement to determine if parents are contributing to student concerns or could potentially be a valuable component of an intervention.
- 6. It is recommended that parents exercise precision parenting for extended adolescence by establishing appropriate boundaries. Boundaries should foster a balance between independence and support.
- 7. It is recommended that parents and students communicate openly about ways parents can best support their students to ensure student satisfaction with parents' involvement. Higher education institutions can assist by creating vehicles that empower students and encourage discussion of student and parental expectations, for example purposeful opportunities during orientations and guiding questions based on critical academic moments.

- 8. It is recommended that further research be carried out to determine other factors that impact mental health status in college students in order to decrease the number of languishing students.
- 9. It is recommended that further research be carried out to determine how student needs for parental involvement change over the college experience beyond the freshman year.
- 10. It is recommended that more survey instruments be developed and validated to specifically investigate parental involvement with college students, allowing the research to advance beyond exploratory studies.

References

- American Broadcasting Company [ABC] (2006). Do 'helicopter moms' do more harm Than good? Retrieved from http://abcnews.go.com/2020/print?id=1237868.
- American College Personnel Association. (1996). *Principles of good practice*.

 Alexandria, VA: Author.
- American Psychological Association. (2003). *APA commends Davis/Osborne bill to Improve mental health services on college campuses*. Retrieved November 7 from http://www.apa.org/releases/campuscare.html
- Al-Hassan, S. & Gardner, R. (2002). Involving immigrant parents of students with disabilities in the educational process. *Teaching Exceptional Children*, 34(5), 52-58.
- Amato, P. (1994). Father-child relations, mother-child relations, and offspring

 Psychological well-being in early adulthood. *Journal of Marriage and the Family*,

 56, 1031-1042.
- Ambler, V. M. (2007). Who flourishes in college? Using positive psychology and student involvement theory to explore mental health among traditionally aged undergraduates. Unpublished doctoral dissertation, College of William and Mary, Virginia.
- American College Health Association [ACHA]. (2008). American College Health

- Association: National college health assessment spring 2007 (abridged). *Journal* of American College Health, 56(5), 469-479.
- American College Personnel Association [ACPA]. (1994). *The student learning Imperative: Implications for student affairs*. Washington, D.C.: Author.
- American College Personnel Association [ACPA]. (1996). *Principles of good practice*. Alexandria, VA: Author.
- American Psychiatric Association [APA] (2000). Diagnostic and statistical manual of mental disorders (4th ed. TR). Washington DC: Author.
- Andom, M. (2007, December 6, 2007). Traditional-Age College Students Are Not Adults in Most Parents' Eyes, or Their Own, Study Finds. *Chronicle of Higher Education*. Retrieved December 10, 2007 from http://chronicle.com/cgibin/printable.cgi?article.
- Andrews, B., & Wilding, J. M. (2004). The relation of depression and anxiety to life stress and achievement in students. *British Journal of Psychology*, 95(4), 5-30.
- Angst, J. (1988). Clinical course of affective disorders. In T. Helgason & R. J. Daly (Eds.), *Depression illness: Prediction of course and outcome* (pp. 1-47). Berlin, Germany: Springer-Verlag.
- Arehart-Treichel, J. (2002). Mental illness on rise on college campuses. *Psychiatric News*, *37*(6), 6.
- Arnett, J. J. (2000). Emerging adulthood: A theory of development from the late teens Through the twenties. *American Psychologist*, *55*, 469–480.
- Arnett, J. J. (2004). The road to college: Twists and turns. In J. J. Arnett (Ed.), Emerging

- adulthood: The winding road from the late teens through the twenties (pp. 119–141). New York: Oxford University Press.
- Astin, A.W. (1993). What matters in college: four critical years revisited. San Francisco: Jossey-Bass.
- Astin, A. W., Tsui, L., & Avalos, J. (1996). Degree attainment rates at American college and universities: Effects of race, gender, and institutional type. Los Angeles: Higher Education Research Institute, University of California, Los Angeles.
- Aspinwall, L. G., & Staudinger, U. M. (Eds.), A psychology of human strengths:

 Fundamental questions and future directions for a positive psychology. (pp. 305-317). Washington, D.C.: American Psychological Association.
- Bahr, S. J., Maughan, S. L., Marcos, A. C., & Li, B. (1998). Family, religiosity, and the risk of adolescent drug use. *Journal of Marriage and the Family*, 60(4), 979-992.
- Baker, A. J., Kessler-Sklar, S., Piotrkowski, C. S., & Parker, F. L. (1999). Kindergarten and first-grade teachers' reported knowledge of parents' involvement in their children's education. *Elementary School Journal*, 99, 367-380.
- Barber, B. K. (1996). Parental psychological control: Revisiting a neglected construct. *Child Development*, 1996, 67(6), 3296–3319.
- Barber, B. K. (Ed.). (2002). *Intrusive Parenting: How Psychological Control Affects*Children and Adolescents. Washington, D.C.: American Psychological

 Association.
- Barber, B. K., & Erickson, L. D. (2001). Adolescent social initiative: Antecedents in the ecology of social connections. *Journal of Adolescent Research*, 16(4), 326-354.
- Barber, B. K., Maughan, S. L., & Olsen, J. A. (2005). Patterns of parenting across

- adolescence. New Directions for Child and Adolescent Development, 108, 5-16.
- Barnett, R. C., Kibria, N., Baruch, G. K., & Pleck, J. H. (1991). Adult daughter-parent relationships and their associations with daughters' subjective well-being and psychological distress. *Journal of Marriage and the Family*, *53*, 29-42
- Bates, J. E., Petit, G. S., Dodge, K. A., & Ridge, B. (1998). Interaction of temperamental resistance to control and restrictive parenting in the development of externalizing behavior. *Developmental Psychology*, *34*, 982–995.
- Baumrind, D. (1967). Child care practicies anteceding three patterns of preschool behavior. *Genetic Psychology Monographs*, 75, 43-88.
- Baumrind, D. (1971). Current patterns of parental authority. *APA Developmental Psychology Monograph*, 4(1).
- Baumrind, D. (1999). The influence of parenting style on adolescent competence and substance abuse. *Journal of Early Adolescence*, 11(1), 56-95.
- Baumrind, D. (2005). Patterns of parental authority and adolescent autonomy. *New Directions for Child and Adolescent Development*, 108, 61-69.
- Baxter Magolda, M. V. (1999). Constructing adult identities. *Journal of College Student Development*, 40(6), 629-644.
- Bearman, P. S., et al. (1997). Protecting adolescents from harm: Findings from the National Longitudinal Study for Adolescent Health. *Journal of the American Medical Association*, 278(10): 823-832.
- Bednar, D. E., & Fisher, T. D. (2003). Peer referencing in adolescent decision making as a function of perceived parenting style. *Adolescence*, 38(152), 607-621.
- Beerman, K. A. (1991). Variation in nutrient intake of college students: A comparison by

- students' residence. Journal of the American Dietary Association, 91, 343-344.
- Benson, F. & Martin, S. (2003). Organizing successful parent involvement in urban schools. *Child Study Journal*, *33*(3), 187-193.
- Benton, S.A., Robertson, J.M., Tseng, W.C., Newton, F.B., & Benton, S.L. (2003). Changes in counseling center client problems across 13 years. *Professional psychology: Research and practice*, *34*, 66-72.
- Bers, T. (2005). Parents of traditionally aged community college students:

 Communications and choice. *Research in Higher Education*, 46(4), 413-436.
- Bird, K. (2006). How do you spell parental involvement? S-I-S. *THE Journal*, 33(7), 38-42.
- BlueCross and BlueShielf. (2006). *National healthcare trends*. Retrieved November 15, 2007 from http://www.bcbs.com/mcrg/chap1/ch1_Slide_1.html
- Blum, R., & Rinehart, P. M. (1997). Reducing the risk: Connections that make a difference in the lives of youth. *Youth Studies Australia*, *16*(4), 37-50.
- Boethel, M. (2003). *Diversity: School, family, and community connections: Annual synthesis*. National Center for Family and Community Schools, Southwest Educational Development Laboratory, Austin, TX.
- Borkowsky, J., Ramey, S., & Bristol-Power, M. (Eds.). (2002). *Parenting and the child's world: Influences on academic, intellectual, and social-emotional development.*Mahwah, NJ: Erlbaum.
- Bouchard, T. J., Jr., & McGue, M. (1990). Genetic and rearing environmental influences on adult personality: An analysis of adopted twins reared apart. *Journal of Personality*, 58, 263–292.

- Bowlby, J. (1982). Attachment and loss. New York, NY: Basic Books.
- Boyer, E. L. (1990). Scholarship reconsidered: Priorities of the professoriate.

 Princeton, NJ: Carnegie Foundation for the Advancement of Teaching.
- Brooks, R. (2004). 'My mum would be as pleased as punch if I went, but dad seems a bit more particular about it': Paternal involvement in young people's higher education choices. *British Educational Research Journal*, *30*(4), 495-514.
- Brown, B., Mounts, N., Lamborn, S., & Steinberg, L. (1993). Parenting practices and peer group affiliation in adolescence, *Child Development*, *64*, 467-482.
- Bumpass, L. & Aquilino, W. (1994). A social map of midlife: *Family and work over the middle life course*. Madison, WI: Center for Demography and Ecology,

 University of Wisconsin-Madison.
- Buri, J.R. (1991). Parental authority questionnaire. *Journal of Personality Assessment*, 57(1), 110-119.
- Bylund, C. L., Imes, R. S., & Baxter, L. A. (2005). Accuracy of parents' perceptions of their college student children's health and health risk behaviors. *Journal of American College Health*, *54*(1), 31-37.
- Carlise, E., Stanley, L., & Kemple, K. M. (2006). Opening doors: Understanding school and family influences on family involvement. *Early Childhood Education Journal*, 33(3), 155-162.
- Center for Cognitive Therapy. (1978). *The Beck Inventory*. Philadelphia: Room 602, 133 South 36th Street, 19104. Presently published by the Psychology Corporation, Harcourt, Brace, Jovanovich.
- Chandler, L. A., & Gallagher, R. P. (1996). Developing a taxonomy for problems seen at

- a university counseling center. *Measurement and Evaluation in Counseling and Development*, 29, 4-12.
- Chao, R. K. (2002). Cultural explanations for the role of parenting in the school success of Asian-American childrearing beliefs. In R. Taylor (Ed.), Resilience across contexts: Family, work, culture, and community. Mahway: Lawrence Erlbaum.
- Chavkin, N. F. & Williams, D. L. (1993). Minority parents and the elementary school:

 Attitudes and practices. In N. F. Chavkin (Ed.), *Families and schools in a*pluralistic society)pp. 73-83). Albany: SUNY Press.
- Choi, Y. E., Bempechat, J., & Pinsburg, H. P. (1994). Educational socialization in Korean American children: A longitudinal study. *Journal of Applied Developmental Psychology*, *15*, 313-318.
- Chorpita, B. F., & Barlow, D. H. (1998). The development of anxiety: The role of control in the early environment. *Psychological Bulletin*, *124*(1), 3-21.
- Christenson, S. L. (1999). Families and schools: Rights, responsibilities, resources, and relationships. In R. C. Pianta & M. J. Cox (Eds.), *The transition to kindergarten*. Baltimore, MD: Brookes Publishing.
- Christie, N. G., & Dinham, S. M. (1991). Instintutional and external influences on social integration in the freshman year. *Journal of Higher Education*, 62, 412-436.
- Coburn, K. L. (2003, 4th ed.). *Letting go: A parents' guide to understanding the college years*. Harper Collins Publishers: New York, NY.
- Coburn, K. L. (2006). Organizing a ground crew for today's helicopter parents. *About Campus*, 11(3), 9-16.
- Colavecchio-Van Sickler, S. (2006, June 18). Mommy, tell my professor he's not nice! St.

- Petersburg Times, p. 1.
- Coleman, M. & Churchill, S. (1997). Challenges to family involvement. *Childhood Education*, 73(3), 144-148.
- College Board. (2005). Education pays update: A supplement to education pays 2004 –

 The benefits of higher education for individuals and society. Retrieved April 10,

 2006 from, http://www.collegeboard.com/prod_downloads/press/cost05/

 education pays 05.pdf
- Coleman, J. S. (1998). Social capital in the creation of human capital. *American Journal of Sociology*, 94,95-120.
- Comer, J. (2001). Schools that develop children. *The American Prospect*, 126(7), 3-12.
- Conneely, J. F., Good, C., & Perryman, K. (2001). Balancing the role of parents in the residential community. *New Directions for Student Service*, *94*, 51-61.
- Coombs, R. H., Paulson, M. J., & Richardson, M. A. (1991). Peer vs. parental influence
 In substance use among Hispanic and Anglo children and adolescents. *Journal of Youth and Adolescence*, 20(1), 73-88.
- Coopersmith, S. (1967). *The antecedents of self-esteem*. Palo Alto, CA: Consulting Psychologists Press.
- Coopersmith, S. (1981). Coopersmith Invetory Self-Esteem Inventory (SEI). Palo Alto, CA: Consulting Psychologists Press.
- Daniel, B. B., Evans, S. C., & Scott, B. R. (2001). Understanding family involvement in the college experience today. *New Directions for Student Services*, *94*, 3-13.
- Davidson, M. (2008, March 22). Stop hovering! Just let him go When children start university, parents must learn a degree of separation, says Max Davidson. *The*

- Daily Telegraph, p. 011.
- David, M. E., Ball, S. J., Davies, J., & Reay, D. (2003). Gender issues in parental involvement in student choices of higher education. *Gender & Education*, 15(1), 21.
- Davies, P. T., & Windle, M. (1997). Gender-specific pathways between maternal depressive symptoms, family discord, and adolescent adjustment. *Developmental Psychology*, 33(4), 657-68.
- Desidorato, L. L., & Crawford, H. J. (1995). Risky sexual behavior in college students: Relationships between number of sexual partners, disclosure of previous risky behavior, and alcohol use. *Journal of Youth and Adolescence*, 24(1), 55-68.
- Desimore, L. (1999). Linking parent involvement with student achievement: Do race and income matter? *Journal of Educational Research*, *93*, 11-30.
- Diener, E. (2003). What is positive about positive psychology: The curmudgeon and Pollyanna. *Psychological Inquiry*, *14*, 115-120.
- Dominguez, M. M. & Carton, J. S. (1997). The relationship between self-actualization and parenting style. *Journal of Social Behavior and Personality*, 12(4), 1093-1100.
- Dornbusch, S. M. & Glasgow, K. L. (1996). The structural context of family-school relations. In A. Booth & J. F. Dunn (Eds.), *family-school links: How do they affect educational outcomes?* (pp. 35-44). Mahway, NJ: Erlbaum.
- Dornbusch, S. M., Ritter, P., Liederman, H., Roberts, D., & Fraleigh, M. (1987). The relation of parenting style to adolescent school performance. *Child Development*, 56, 1244-1257.

- Dyson, R. & Renk, K. (2006). Freshmen adaptation to university life. *Journal of Clinical Psychology*, 62(10), 1231-1244.
- Ellen, E.F. (2005). Suicide prevention on campus. *Psychiatric Times*, *XIX*. Retrieved August 23, 2005, from www.psychiatrictimes.com/p021001a.html.
- Elkind, D. (1994). The new family imbalance. Cambrindge, MA: Harvard University.
- Elkins, I. J., McGue, M., & Iacono, W. G. (1997). Genetic and environmental influences on parent—son relationships: Evidence for increasing genetic influence during adolescence. *Developmental Psychology*, *33*, 351–363.
- Engels, R. C., Finkenauer, C., Meeus, W., & Dekovic, M. (2001). Parental attachment and adolescents' emotional adjustment: The associations with social skills and relational competence. *Journal of Counseling Psychology*, 48(4), 428-439.
- Epel, E. S., Blackburn, E. H., Lin, J., Dhabhar, F. S., Adler, N. E., Morrow, J. D., & Cawthon, R. M. (2004). Accelerated telomere shortening in response to life stress. *Proceedings of the National Academy of Sciences, USA, 101*, 17312-17315.
- Epstein, J. L. & Sanders, M. (2002). Family, school, and community partnerships. In M. Bornstein (Ed.), *Handbook of parenting*, *Vol. 5: Practical issues in parenting* (2nd ed.) (pp. 407-437). Mahwah, NJ: Erlbaum.
- Erikson, E. H. (1959). *Identity and the life cycle: Selected papers*. New York, NY: International Universities Press.
- Erikson, E. H. (1982). The life cycle completed. New York, NY: Norton.
- Ettinger, L. K. (2006, August 13). Helicopter parents should let kids soar: Mom's To-Do

- List. *The Oregonian*, p. 4.
- Evans, N.J., Forney, D.S. & Guido-DiBrito, F. (1998). *Student development in college:*Theory, research, and practice. San Francisco: Jossey-Bass.
- Falbo, T., Lein, L., & Amador, N. (2001). Parental involvement during the transition to high school. *Journal of Adolescent Research*, *16*, 511-529.
- Fan, X. & Chen, M. (2001). Parental involvement and students' academic achievement:

 A meta-analysis. *Education Psychology Review*, 13(1), 1-22.
- Fehrmann, P. G., Keith, T. Z., & Reimers, T. M. (1987). Home influences on school learning: Direct and indirect effects of parental involvement on high school grades. *Journal of Educational Research*, 80, 330-337.
- Ferrara, M. & Farrara, P. (2005). Parents as partners: Raising awareness as a teacher preparation program. *The Clearing House, Nov/Dec.*
- Flett, G. L., Hewitt, P. L., & Singer, A. (1995). Perfectionism and parental authority styles. *Individual Psychology*, *51*(1), 50-60.
- Forbes, K.J. (2001). Students and their parents: Where do campuses fit in? *About Campus*, 6(4), 11-17.
- Franz, C. E., McClelland, D. C., & Weinberger, J. (1991). Childhood antecedents of conventional social accomplishment in midlife adults: A 36-year prospective study. *Journal of Personality and Social Psychology*, 60(4), 586-595.
- Freedman, S. G. (2004, September 15). Weaning Parents From Children As They Head Off to College. *The New York Times*, p. 9.
- Fromme, K., Kruse, M., & Corbin, W. (2008). Behavioral risks during the transition from high school to college. *Development Psychology*, *44*(5), 1497-1505.

- Frost, P.O, Lahart, C. M., & Rosenblate, R. (1991). The development of perfectionism: A study of daughters and their parents. *Cognitive Therapy and Research*, 15, 469-490.
- Gall, M.D., Borg, W.R., & Gall, J.P. (1996). Educational research: An introduction. (6th ed.) White Plains, NY: Longman Publishers.
- Gallagher, R. P., Gill, A. M., & Goldstrohm, S. L. (1998). *National survey of counseling center directors*. Alexandrea, VA: International Association of Counseling Services.
- Gallimore, R., Goldenberg, C., & Weisner, T. (1993). The social construction and subjective reality of activity settings: Implications for community psychology.

 *American Journal of Community Psychology, 21, 537-559.
- Gecas, V., & Schwalbe, M. L. (1986). Parental behavior and adolescent self-esteem. *Journal of Marriage and the Family*, 48, 37-46.
- Gerdes, E. P. (2004). Valuing parents. Change, September/October, 60-61.
- Ginsbug, G.S., & Bronstein, P. (1993). Family factors related to children's intrinsic/extrinsic motivational orientation and academic performance. *Child Development*, 64, 1461-1474.
- Gonzalez, A., Greenwood, G., & WenHsu, J. (2001). Undergraduate students' goal orientations and their relationship to perceived parenting styles. *College Student Journal*, 35(2), 182-192.
- Gonzalez, L., Lewinsohn, P. M., & Clarke, G. (1985). Longitudinal follow-up of unipolar depressives: An investigation of predictors of relapse. *Journal of Consulting and Clinical Psychology*, 53, 461-469.

- Gribble, J. N., & Preston, S. H. (Eds.). (1993). *The epidemiological transition: Policy*and planning implications for developing countries. Washington, DC: National Academies Press.
- Griffith, J. (1998). The relation of school structure and social environment to parent involvement in elementary schools. *Elementary School Journal*, 99(1), 53-80.
- Grolnick, W.S. (2003). The psychology of parental control: How well-meant parenting backfires. Mahwah, NJ: Erlbaum.
- Guldner, G. T. (1996). Long-distance romantic relationships: Prevalence and separation-related symptoms in college students. *Journal of College Student Development*, 37, 289-296.
- Gutman, L. M. & McLoyd, V. C. (2000). Parents' management of their children's education within the home, at school, and in the community: An examination of African-American families living in poverty. *Urban Review*, 32, 1-24.
- Hair, E. C., Jager, J., & Garrett, S. B. (2002). Background for community-level work on social competency in adolescence: Reviewing the literature on contributing factors. Washington, DC: Child Trends.
- Hair, E. C., Moore, K. A., Garrett, S. B., Kinukawa, A., Lippman, L., & Michelson, E. (in press). The parent-adolescent relationship scale. In L. Lippman (Ed.),Conceptualizing and measuring indicators of positive development: What do children need to flourish? New York, NY: Kluwer Academic/Plenum Press.
- Halsey, P. (2004). Nurturing parent involvement: Two middle level teachers share their secrets. *The Clearing House, March/April*, 153-137.
- Harvey, J.H. & Pauwels, B.G. (2003). The ironies of positive psychology. *Psychological*

- *Inquiry, 14*, 125-128.
- Herman, M. R., Dornbusch, S. M., Herron, M. C., & Herting, J. R. (1997). The influence of family regulation, connection, and psychological autonomy on six measures of adolescent functioning. *Journal of Adolescent Research*, 12(1), 34-67.
- Hertzler, A. A., & Frary, R. (1992). Dietary status and eating out practices of college students. *Journal of the American Dietary Association*, 92, 867-869.
- Hewitt, P. L., & Flett, G. L. (1993). Perfectionism in the self and social contexts:

 Conceptualization, assessment, and association with psychology. *Journal of Personality and Social Psychology*, 60, 456-470.
- Higher Education Research Institute [HERI]. (2008). *The americanfreshman national norms for fall 2007.* Los Angeles, CA: Author.
- Hill, N. E. & Taylor, L. C. (2004). Parental school involvement and children's academic achievement: Pragmatics and issues. *Current Direction in Psychological Science*, 13, 161-164.
- Hilliard, D. & Pelo, A. (2001). Changing our attitudes and actions in working with families. *Child Cre Information Exchange*, *3*(138), 48-51.
- Hingson, R., Heeren, T., Winter, M., & Wechsler, H. (2005). Magnitude of alcohol-related 2001. *Annual Review of Public Health*, 26, 259–279.
- Hinkelman, J. M., & Luzzo, D. A. (2001, August). *Mental health and careerdevelopment of college students*. Paper presented at the meeting of the AmericanPsychological Association, San Francisco, CA.
- Holahan, C.J. & Moos, R.H. (1981). Social support and psychological distress: A longitudinal analysis. *Journal of Abnormal Psychology*, 90, 365-370.

- Hood, A.B. (1984). *Student development: Does participation affect growth?*Bloomington: Association of College Unions International.
- Hopkins, H. R., & Klein, H. A. (1993). Multidimensional self-perception: Linkages to parental nurturance. *The Journal of Genetic Psychology*, *154*, 465-473.
- Horn, L., & West, J. (1992). *National education longitudinal study of 1998: A profile of parents of eight graders*. Washington, DC: U.S. Government Printing Office.
- Hossler, D., & Vesper, N. (1993). An exploratory study of the factors associated with parental saving for potsecondary education. *The Journal of Higher Education*, 64(2), 140-165.
- Hundleby, J. D., & Mercer, G. W. (1987). Family and friends as social environments and their relationship to young adolescents' use of alcohol, tobacco, and marijuana.

 **Journal of Marriage and the Family, 49, 151-164.
- Hur, Y.-M., & Bouchard, T. J., Jr. (1995). Genetic influences on perceptions of childhood family environment: A reared apart twin study. *Child Development*, 66, 330–345.
- Hurt, J. A. (2000). Create a parent place: Make the invitation for family involvement real. *Young Children*, 55(5), 88-93.
- Huston, A. (1983). Sec typing. In E. M. Hetherington (Ed.), *Handbook of child*psychology: Vol. 4, Socialization, personality, and social development (pp. 387-467). New York: Wiley.
- Jaccard, J., & Becker, M. A. (1997). Statistics for the behavioral sciences. Pacific Grove, CA: Brooks/Cole Publishing Company.
- Jackson, S., Bijstra, J., Oostra, L., & Bosma, H. (1998). Adolescents' perceptions of

- communication with parents relative to specific aspects of relationships with parents and personal development. *Journal of Adolescence*, *21*, 305-322.
- Jacobson, J. (2003, August 8, 2003). Whose College Application Is It, Anyway? *Chronicle of Higher Education*, p. B16.
- Jones, A. & Crandall, R. (1986). Validation of a short-index of self-actulaization.

 *Personality and Social Psychology Bulletin, 12, 63-73.
- Juang, L.P., & Silbereisen, R.K. (1999). Supportive parenting and adolescent adjustment across time in former East and West Germany. *Journal of Adolescence*, 22, 719-736.
- Kaczmarek, P., acklunch, B., & Blemer, P. (1990). The dynamics of ending a romantic relationship: An empirical assessment of grief in college students. *Journal of College Student Development*, 31, 319-324.
- Kadaba, L. S. (2004, January 24). Off to college for parent and child; These parents hover, stay involved. They want it that way. So do the children. *The Philadelphia Inquirer*, p.12
- Kadison, R., & DiGeronimo, T.F. (2004). *College of the overwhelmed: The campus mental health crisis and what to do about it.* San Francisco: Jossey-Bass.
- Katainen, S., Raikkonen, K., Keskivaara, P., & Keltikangas-Jarvinen, L. (1999). Maternal child-rearing attitudes and role satisfaction and children's temperament as antecedents of adolescent depressive tendencies: Follow-up study of 6- to 15-year-olds. *Journal of Youth and Adolescence*, 28(2), 139-163.
- Keith, Z. (2006). Multiple regression and beyond. Austin, TX: Allyn & Bacon.
- Kennedy, M. (2007). Crisis on campus. American School & University, 79(10), 18-25.

- Kenny, M. E., & Donaldson, G. A. (1991). Contributions of parental attachment and family structure to the social and psychological functioning of first-year college students. *Journal of Counseling Psychology*, 38, 135-146.
- Kerns, K. A., & Stevens, A. C. (1996). Parent-child attachment in late adolescence: Links to social relations and personality. *Journal of Youth and Adolescence*, 25(3), 323-342.
- Keyes, C.L.M (2002). The mental health continuum: From languishing to flourishing in life. *Journal of Health and Social Behavior*, 43, 207-222.
- Keyes, C.L.M. (2003). Complete mental health: An agenda for the 21st century. In Keyes and Haidt (Eds.) *Flourishing: Positive psychology and the life well-lived* (pp. 293-312). Washington, D.C.: American Psychological Association.
- Keyes, C.L.M. (2005). Mental illness and/or complete mental health? Investigating axioms of e complete state model of health. *Journal of Consulting and Clinical Psychology*, 73, 539-548.
- Keyes, C. L. M. (2007). Promoting and protecting mental health as flourishing. *American Psychologist*, 62(2), 95-108.
- Keyes, C.L.M. & Haidt, J., (Eds.). (2003). Flourishing: Positive psychology and the life well-lived. Washington, D.C.: American Psychological Association.
- Keyes, C.L.M., & Lopez, S.J. (2002). Toward a science of mental health. In C.R. Snyder and S.J. Lopez (Eds.) *Handbook of Positive Psychology* (pp.45-59). New York: Oxford University Press.
- Keyser, J. (2001). Creating partnerships with families: Problem-solving through communication. *Child Care Information Exchange*, 44(138), 44-47.

- Kim, E. (2002). The relationship between parental involvement and children's educational achievement in the Korean immigrant family. *Journal of Comparative Family Studies*, *33*, 529-563.
- King, L.A. (2003). Some truth behind the trombones. *Psychological Inquiry*, *14*, 128-131.
- Klein, H. A., O'Bryant, K., & Hopkins, H. R. (1996). Recalled parental authority style and self-percetion in college men and woman. *The Journal of Genetic Psychology*, *157*(1), 5-17.
- Kohl, G. O., Weissberg, R. P., Reynolds, A. J., & Kasprow, W. J. (1994, August).

 Teacher perceptions of parent involvement in urban elementary schools:

 Sociedemographic and school adjustment correlates. Paper presented at the annual meeting of the American Psychological Association, Los Angeles, CA.
- Kuh, G. D. (2002). The National Survey of Student Engagement: Conceptual Framework and Overview of Psychometric Properties. Retrieved November 5, 2007 from http://www.iub.edu/~nsse/html/psychometric_framework_2002.htm
- Kuh, G.D., Kinzie, J., Schuh, J.H., Whitt, E.J., & Associates (2005). Student success in college: Creating conditions that matter. San Francisco: Jossey-Bass.
- Lane, T. (2006, March 19). Helicopter Parents: They hover over their college-age children, unwilling to let go. *Toledo Blade*. Retrieved June 30, 2008, from http://toledoblade.com/apps/pbcs.dll/article?AID=/20060319/ART16/60319010/-1/ART.
- Lange, D. K., & Stone, M. E. (2001). Parental Involvement in Admissions and Financial Aid. *New Directions for Student Services* 2001(94), 15-27.

- Lapsley, D. K., Rice, K. G., & Shadid, G. E. (1989). Psychological separation and adjustment to college. *Journal of Counseling Psychology*, *36*, 286-294.
- Larson, L. E. (1974). An examination of the salience hierarchy during adolescence: The influence of the family. *Adolescence*, *9*, 317-332.
- Larson, R. W. (1983). Adolescents' daily experience with family and friends:Contrasting opportunity systems. *Journal of Marriage and the Family*, 45, 739-750.
- Lazarus, R.S. & Folkman, S. (1984). *Stress, appraisal, and coping*. New York: Springer Publishing.
- Lee, S. A. (1993). Family structure effects on student outcomes. In B. Schneider & J. S. Coleman (Eds.), *Parents, their children, and school* (pp. 43-75). Boulder, CO: West-view Press.
- Lee, G. & Manning, L. (2001). Working with Asian parents and families. *Multicultural Education*, 9, 23-25.
- Levine A. & Cureton, J. S. (1998). When hope and fear collide: A portrait of today's college student. San Francisco, CA: Jossey-Basee.
- Lewinsohn, P. M., Hoverman, H. M., & Rosenbaum, M. (1988). A prospective study of risk factors for unipolar depression. *Journal of Abnormal Psychology*, 97, 251-264.
- Lin, N. (2001a). Building a network theory of social capital. In N. Lin, K. Cook, & R. S. Burt (Eds.), *Social capital: Theory and research* (pp. 3-29). New York, NY: Aldine De Gruyter.
- Lin, N. (2001b). Social capital: A theory of social structure and action. New York, NY:

- Cambridge University Press.
- Lins-Dyer, T. (2005). Mexican adolexcents' perceptions of parental control and academic achievement: A social domain approach. Unpublished doctoral dissertation,

 University of Illinois at Chicago.
- Linver, M. R. & Silverberg, S. B. (1997). Maternal predictors of early adolescent achievement-related outcomes: Adolescent gender as moderator. *Journal of Early Adolescence*, 17, 294-318.
- Lipka, S. (2008). Stress Takes a Toll on C ollege Students, Survey Finds. *The Chronicle of Higher Education*. Retrieved March 20, 2008 from http://chronicle.com/temp/email2.php?id=jmmmmYSmFjmFnvxzXfDFDgfncpS Nyb5h.
- Lum, L. (2006). Handling 'Helicopter Parents'. *Diverse: Issues in Higher Education*, 23(20), 40-42.
- Mansfield, P. M., & Warwick, J. (2005). Gender differences in students' and parents' evaluative criteria when selecting a college, *Journal of Marketing for Higher Education*, 15(2), 47-80.
- Martinez Aleman, A. M. (1997). Understanding and investigating female friendship's educative value. *Journal of Higher Education*, 68, 119-159.
- Martini, M. (1995). Features of home environments associated with children's school success. *Early Child Development and Care*, 111, 49-68.
- Maslow, A. (1954). *Motivation and personality*. New York: Harper & Row.
- Maslow, A. (1970). *Motivation and personality* (2nd ed.). New York: Harper & Row.
- Mason, C. A., Cauce, A. N., Gonzales, N., & Hiraga, Y. (1996). Neither too sweet nor

- too sour: Problem peers, maternal control, and problem behavior in African American adolescents. *Child Development*, *67*(*5*), 2115-2130.
- Mau, W. (1997). Parental influences on the high school studnets' academic achievement:

 A comparison of Asian immigrants, Asian Americans, and White Americans.

 Psychology in the Schools, 34, 267-277.
- Mayhew, K.P., & Lempers, J.D. (1998). The relation among financial strain, parenting, parent self-esteem, and adolescent self-esteem. *Journal of Early Adolescence*, 18, 145-72.
- McEwen, B. S. (1998). Stress, Adaptation, and disease: Allostatsis and allostatic load. *Annals of the New York Academy of Sciences*, 840, 33-44.
- McGinty, S. M. (2002). Issues of Access: the College Application Essay. *The Journal of College Admission*, 2002(177), 26-30.
- McGue, M., Elkins, I., Walden, B., & Iacono, W. G. (2005). Perceptions of the parent–adolescent relationship: A longitudinal investigation. *Developmental Psychology*, *41*, 971–984.
- McNall, S., Eisenberg, N., & Harris, J. D. (1991). Consistency and change in maternal child-rearing practices: A longitudinal study. *Child Development*, 62, 190-198.
- McNeal, R. B. (1999). Parental involvement as a social capital: Differential effectiveness on science, achievement, truancy, and dropping out. *Social Forces*, 78, 114-144.
- Mearns, J. (1991). Coping with a breakup: Negative modd regulation expectancies and depression following the end of a romantic relationship. *Journal of Personality and Social Psychology*, 60, 327-324.
- Merriman, L. S. (2007). It's Your Child's Education, Not Yours. Chronicle of Higher

- Education, 54(13), B20.
- Miller, B. C, (1998). Families matter: *A research synthesis of family influences on adolescent pregnancy*. Washington, DC: The National Campaign to Prevent Teen Pregnancy.
- Moles, O. (1993). Collaboration between schools and disadvantaged parents: Obstacles and opportunities. In N. Chavkin (Ed.), *Families and schools in a pluralistic society* (pp. 21-49). Albany, NY: SUNY Press.
- Moore, K. A. & Zaff, J. F. (2002). Building a better teenager: A summary of "what works" in adolescent development. Report prepared for John s. and James L. Knight Foundation. Washington, DC: Child Trends.
- Moore, K. A., Guzman, L., Hair, E., Lippman, L. & Garrett, S. (2004). Parent-teen relationships and interactions: Far more positive than not. *Child Trends*, December 2004.
- Moos, R. M., Insel, P. M., & Humphrey, B. (1974). *Combined preliminary manual for the family, work and group encironment scales*. Palo Alto, CA: Consulting Psychologists Press.
- Morrow, R. D. (1989). Southeast-Asian parental involvement: Can it be a reality? *Elementary School Guidance & Counseling*, 23, 289-297.
- Muller, C. (1993). Parent involvement and academic achievement: An analysis of family resources available to the child. In B. Schneider & J. S. Coleman (Eds.). *Parents, their children, and school* (pp. 77-113). Boulder, CO: Westview Press.
- Marano, H.E. (2002, May). Crisis on the campus. [Electronic version]. *Psychology Today*. Retrieved November 8, 2007 from

- http://www.psychologytoday.com/htdocs/prod/ptoarticle/pto-20030501-000005.asp
- Murray, C. J. L., & Lopez, A. D. (1996). The global burden of disease: A comprehensive assessment of mortality and disability from diseases, injuries, and risk factors in 1990 and projected to 2020. Cambridge, MA: Harvard School of Public Health.
- Murray, C. J. L., & Lopez, A. D. (1997). Global mortality, disability, and the contribution of risk factors: Global burden of disease study. *Lancet*, *349*, 1436-1442.
- Nord, C. W., Lennon, J., Liu, B., & Chandler, K. (1999). *Home literacy activities and signs of children's emerging literacy, 1993 and 1999* (NCES Publication No. 2000-026). Wahsington, DC: National Center for Education Statistics.
- Nord, C. W. & West, J. (2001). Fathers' and mothers' involvement in their children's schools by family type and resident status. *Education Statistics Quarterly*, 3(2), 40-43.
- Nucci, L. (2001). *Education in the moral domain*. Cambridge: Cambridge University Press.
- Nucci, L., Camino, C., & Milnitsky-Sapiro, C. (1996). Social class effects on Northeastern Brazilian children's conceptions of areas of personal choice and social regulation. *Child Development*, 67(3), 1223-1242.
- Nucci, L., Hasabe, Y., & Lins-Dyer, M. T. (2005). Adolescent psychological well0being and parent control of the personal. *New Directions for Child and Adolescent Development*, 108, 17-30.

- Nystul, M. (1984). Positive parenting leads to self-actualizing children. *Individual Psychology: Journal of Adlerian Theory. Research and Practice, 40,* 177-183.
- O'Connor, T. G., McGuire, S., Reiss, D., Hetherington, E. M., & Plomin, R. (1998). Co-occurrence of depressive symptoms and antisocial behavior in adolescence: A common genetic liability. *Journal of Abnormal Psychology*, 107(1), 27-37.
- Oliver, B. (n.d.). Parent involvement and its impact on undergraduate students at four year public universities. Unpublished doctoral dissertation, University of Alabama, Alabama.
- Oliver, J. M. & Paul, J. C. (1995). Self-esteem and self-efficacy: Perceived parenting and family climate; and depression in university students. *Journal of Clinical Psychology*, *51*(4), 467-481.
- O'Malley, P., & Johnston, L. D. (2002). Epidemiology of alcohol and other drug use among American college students. *Journal of Studies on Alcohol*, 14, 23–39.
- Padilla, R., Trevino, J., Gonzalez, K., & Trevino, J. (1997). Developing local models of minority student success in college. *Journal of College Student Development*, 38, 125-135.
- Parsons, J. E., Adler, T. F., & Kaczala, C. M. (1982). Socialization of achievement attitudes and beliefs: Parental influences. *Child Development*, *53*, 310-321.
- Pascarella, E.T. & Terenzini, P.T. (2005) *How college affects students: A third decade of research*. San Francisco: Jossey-Bass.
- Pattenaude, R. L. (2000). Administering the modern university. In J. Losco & B. L. Fife (Eds.), *Higher Education in Transition*. New York, NY: Bergin & Garvey.
- Paulson, S. E. (1994). Relations of parenting style and parental involvement with ninth-

- grade students' achievement, Journal of Early Adolescence, 14, 250-267.
- Paulussen-Hoogeboom, M. C., Stams, G. J. J. M., Hermanns, J. M. A., & Peetsma, T. T.
 D. (2007). Child negative emotionality and parenting from infancy to preschool:
 A meta-analytic review. *Developmental Psychology*, 43, 438–453.
- Perna, L. W. & Titus, M. A. (2005). The relationship between parental involvement as social capital and college enrollment: An examination of racial/ethnic group differences. *The Journal of Higher Education*, 76(5), 485-518.
- Peterson, B. E. (2006). Generativity and Successful Parenting: An analysis of young adult outcomes. *Journal of Personality*, 74(3), 847-869.
- Peterson, C. & Seligman, M.P. (2004). *Character strengths and virtues: A handbook and classification*. New York: Oxford University Press.
- Piaget, J. (1965). Moral judgement and the child. New York, NY: Free Press.
- Piccinelli, M., & Wilkinson, G. (1994). Outcome of depression in psychiatric settings. *British Journal of Psychiatry*, 164, 297-304.
- Plomin, R., & Bergeman, C. S. (1991). The nature of nurture: Genetic influence on "environmental" measures. *Behavioral and Brain Sciences*, 14, 373–427.
- Plomin, R., McClearn, G. E., Pedersen, N. L., Nesselroade, J. R., & Bergeman, C. S. (1988). Genetic influence on childhood family environment perceived retrospectively from the last half of the life span. *Developmental Psychology*, 24, 738–745.
- Ratelle, C.F., Larose, S., Guay, F., & Senecal, C. (2005). Perceptions of parental involvement and support as predictors of college students' persistence in a science curriculum. *Journal of Family Psychology*, *19*(2), 286-293.

- Reissman, C. K., Whalen, M. H., Frost, R. O., & Morgenthau, J. E. (1991). Romance and help-seeking among college women: "It hurts so much to care." *Women and Health*, 17(4), 21-47.
- Reynolds, A. J. (1992). Comparing measures of parental involvement and their effects on academic achievement. *Early Childhood Research Quarterly*, 7, 441-462.
- Reynolds, A. J., & Gill, S. (1994). Comparing measures of parental involvement and their effects on academic achievement. *Early Childhood Research Quarterly*, 7, 441-462.
- Riba, M. (2004). APA to launch MH initiative for college students, families.

 *Psychiatric News, 39(23), 3.
- Roberts, G. C., Block, J. H., & Block, J. (1984). Continuity and change in parents' child-rearing practices. *Child Development*, *55*, 586-597.
- Robins, L. N., & Regier, D. A. *Eds.). (1991). *Psychiatric disorders in America: The epidemiological catchment area study*. New York: Free Press.
- Roedel, T.D., Schraw, G., & Plake, B.S. (1994). Validation of a measure of learning and performance goal orientations. *Educational and Psychological Measurement*, 54(4), 1013-1021.
- Rogers, M.J., & Holmbeck, G.N. (1997). Effects of interparental aggression on children's adjustment: The moderating role of cognitive appraisal and coping. *Journal of Family Psychology*, 11, 125-130.
- Rowe, D. C. (1981). Environmental and genetic influences on dimensions of perceived parenting: A twin study. *Developmental Psychology*, *17*, 203–208.
- Rowe, D. C. (1983). A biometrical analysis of perceptions of family environment: A

- study of twin and singleton sibling kinships. Child Development, 54, 416–423.
- Rubin, K. H., Burgess, K. B., Dwyer, K. M., & Hastings, P. D. (2003). Predicting preschoolers' externalizing behaviors from toddler temperament, conflict, and maternal negativity. *Developmental Psychology*, *39*, 164–176.
- Rudolf, R. (1994). Post-adolescence as a new phase in the family cycle? *Innovation:*The European Journal of Social Sciences, 7(1), 63-68.
- Rumberger, R. W., Ghatak, R., Poulos, G., Ritter, P. L., & Dornbusch, S. M. (1990).

 Family influences of dropout behavior in one California high school. *Sociology of Education*, 63, 283-299.
- Ryff, C.D. (2003). Corners of myopia in the positive psychology parade. *Psychological Inquiry*, *14*, 153-159.
- Savage, M. (2007). *Developing a parent program that reflects campus culture*. Twin Cities, MN: University of Minnesota, Office of Parent Programs.
- Sax, L. J. (1996). The dynamics of tokenism: How college students are affected the proportion of women in their major. *Research in Higher Education*, 37(3), 389-426.
- Sax, L.J., Bryant, A.N., & Gilmartin, S.K. (2002). A longitudinal investigation of emotional health among first-year college students: Comparisons of women and men. Los Angeles, CA: Higher Education Research Institute.
- Schaefer, E. W. (1965). Children's reports of parental behavior: An inventory. *Child Development*, *36*, 413-424.
- Schneider, L. D. (1977). Counseling. In W. Packwood (Eds.), *College student personnel services* (pp. 335-357). Springfield, IL: Thomas.

- Schulengerg, J., & Maggs, J. L., (2002). A developmental perspective on alcohol use and heavy drinking during adolescence and the transition to young adulthood.

 **Journal of Studies on Alcohol, Supplement 14: 54-70.
- Schulenberg, J. E., Merline, A. C., Johnston, L. D., O'Malley, P. M., Bachman, J. G., & Laetz, V. B. (2005, Spring). Trajectories of marijuana use during the transition to adulthood: The big picture based on national panel data. *Journal of Drug Issues*, 255–279.
- Scott, B. R., & Daniel, B. V. (2001). Why Parents of Undergraduates Matter to Higher Education. *New Directions for Student Services*, 2001(94), 83-89.
- Seligman, M.E.P. (1998a). President's address from the American Psychological Association's annual report. Retrieved on November 5, 2003 from http://www.psych.upenn.edu/seligman/aparep98.htm.
- Seligman, M.E.P. (1998b). Positive psychology network concept paper. Retrieved on November 5, 2003 from http://www.psych.upenn.edu/seligman/aparep98.htm.
- Seligman, M.E.P. & Pawelski, J.O. (2003). Positive psychology: FAQs. *Psychological Inquiry*, 14, 159-163.
- Servin, L. & Karp, J. (2003). Intergenerational studies of parenting and the transfer of risk from parent to child. *Current Issues in Psychological science*, 2003(12), 138-142.
- Shaw, B. A., Krause, N., Chatters, L. M., Connell, C. M., & Ingersoll-Dayton, B. (2004). Emotional support from parents early in life, aging and health. *Psychology and Aging*, 19(1), 4-12.
- Sheldon, S. B. (2002). Parents' social networks and beliefs as predictors of parent

- involvement. The Elementary School Journal, 102(4), 301-316.
- Shellenbarger, S. (2006). College ward off overinvolved parents. Retrieved on September 24, 2006 from
 - htt://www.careerjournal.com/columnists/workfamily/20050729-workfamily.html
- Sliwka, A. & Istance, D. (2006). Parental and stakeholder 'voice' in schools and systems. European Journal of Education, 41(1), 29-43.
- Slutske, W. S., Hunt-Carter, E. E., Nabors-Oberg, R. E., Sher, K. J., Bucholz, K. K., Madden, P., et al. (2004). Do college students drink more than their non-college-attending peers? Evidence from a population-based longitudinal female twin study. *Journal of Abnormal*Psychology, 113, 530–540.
- Smetana, J. G. (1995). Parenting styles and conceptions of parental authority. *Child Development*, 66(2), 299-316.
- Smetana, J., Crean, H. F., & Campione-Barr, N. (2005). Adolescents' and parents' changing conceptions of parental authority. *New Directions for Child and Adolescent Development*, 108, 31-46.
- Smetana, J. G., Crean, H. F., & Daddis, C. (2002). Family processes and problem behaviors in middle-class African-American adolescents. *Journal of Research on Adolescence*, 12(2), 563-580.
- Smith, T. & Renk, K. (2007). Predictors of academic-related stress in college students:

 An examination of coping, social support, parenting, and anxiety. *NASPA Journal*, *44*(3), 405-431.
- Snyder, M.B. (2004, March). Senior Student Affairs Officer Round Table: Hottest Topics

- in Student Affairs. Session held at the annual meeting of the National Association of Student Personnel Administrators, Denver, Colorado.
- Snyder, C.R. & Lopez, S.J., Eds. (2002). *Handbook of positive psychology*. New York: Oxford University Press.
- South, S. C., Krueger, R. F., Iacono, W. G., & Johnson, W. (2008). Adolescent personality moderates genetic and environmental influences on relationships with parents. *Journal of Personality and Social Psychology*, *94*(5), 899-912.
- Steinberg, L. Dornbusch, S. M., & Brown, B. B. (1992). Ethnic differences in adolescent achievement. *American Psychologist*, 47, 723-729.
- Steinberg, L., Lamborn, S.D., Darling, N., Mounts, N.S., & Dornbusch, S.M. (1994).

 Overtime changes in adjustment and competence among adolescents from authoritative, authoritarian, indulgent, and neglectful families. *Child Development*, 65, 754-770.
- Steinberg, L., Lamborn, S. C., Dornbusch, S. M., & Darling, N. (1992). Impact of parenting practices on adolescent achievement: Authoritative parenting, school involvement, and encouragement to succeed. *Child Development*, 63, 1266-1281.
- Steinberg, L. & Silk, J. S. (2002). Parenting adolescents. In M. H. Bornstein (Ed.), *The handbook of parenting, volume I: Children and parenting*. Mahwah, NJ: Erlbaum.
- Steinberg, L., & Silverberg, S. (1990). The vicissitudes of autonomy in early adolescence. *Child Development*, *57*, 841-851.
- Stiles, P. (2005). *Is the American dream killing you? How the market rules our lives.*New York: Collins.
- Stone, S. (2006). Correlates of change in student reported parent involvement in

- schooling: A new look at the National Education Longitudinal Study of 1988. American Journal of Orthopsychiatry, 76(4), 518-529.
- Stoolmiller, M. (2001). Synergistic interaction of child manageability problems and parent-discipline tactics in predicting future growth in externalizing behavior for boys. *Developmental Psychology*, *37*, 814–825.
- Strage, A. & Brandt, T. S. (1999). Authoritative parenting and college students' academic adjustment and success. *Journal of Educational Psychology*, *91*(1), 146-156.
- Strange, C.C. & Banning, J.H. (2001). Educating by design: Creating campus learning environments that work. San Francisco, CA: Jossey-Bass.
- Strauss, V. (2006, March 21). Putting Parents In Their Place: Outside Class. Too Much Involvement Can Hinder Students' Independence, Experts Say. *Washington Post*, p.A08
- Sui-Chu, E. H., & Willims, J. D. (1996). Effects of parental involvement on eigth-grade achievement. *Sociology of Education*, 69, 126-141.
- Sun, Y. (1998). The academic success of East-Asian-American students: An investment model. *Social Science Research*, 27, 432-456.
- Super, C. M, & Harkness, S. (1986). The development niche: A conceptualization at the interface of child and culture. *International Journal of Behavioral Development*, 9, 545-569.
- Swick, K. (2006). Families and educators together: Raising caring and peaceable children. *Early Childhood Education Journal*, *33*(4), 279-287.
- Sy, S. R., Rowley, S. J., & Schulenberg, J. E. (2007). Predictors of parent involvement

- across contexts in Asian American and European American families. *Journal of Comparative Family Studies*, 38(1), 1-9.
- Thomas, I., & Raj. H. S. (1985). A factor analytical study on the antecedents of self-esteem. *Psychological Studies*, *30*, 97-101.
- Tinto, V. (1993). *Leaving college: Rethinking the causes and cures of student attrition* (2nd ed.). Chicago: University of Chicago Press.
- Tipton, R. W. & Worthington, E. L. (1984). *Self-Efficacy Scale*. (Available from Robert M. Tipton, Department of Psychology, Virginia Commonwealth University, 800 West Franklin Street, Richmond, VA 23284.)
- Tuckman, B. W. (1999). *Conducting educational research* (5th ed.). Fort Worth, TX: Harcourt Brace.
- Turiel, E. (2005). The many faces of parenting. New Directions for Child and Adolescent Development, 108, 79-88.
- Umberson, D. (1992). Relationships between adult children and their parents:

 Psychological consequences for both generations. *Journal of Marriage and the Family*, *54*, 664-674.
- United States Department of Education [USDE]. (2005). Facts and terms every parent should know about NCLB. Retrieved December 1, 2005

 (http://www.ed.gov/nclb/overview/intro/parents/parentfacts.html). In Sy, S. R., Rowley, S. J., & Schulenberg, J. E. (2007). Predictors of parent involvement across contexts in Asian American and European American families. *Journal of Comparative Family Studies*, 38(1), 1-9.
- United States Public Health Service. (1999). Mental Health: A report of the Surgeon

- General. Rockville, MD: Author.
- Wagner, B. M., Cohen, P., & Brook, J. S. (1996). Parent/adolescent relationships:

 Moderators of the effects of stressful life events. *Journal of Adolescent Research*,

 11(3), 347-374.
- Watabe-Dawson, M., & Sasaki, A. (2000). Comparison of nutrient intake profiles between university students in the United States and Japan. *Journal of the*
- National Association of College and University Food Service, 22, Article 0001a.

 Retrieved November 10, 2007 from

 http://www.nacufs.org/services/publications/journal_2000/dawson.asp
- Wechsler, H., Dowdall, G. W., Davenport, A., & Castillo, S. (1995). Correlates of college student binge drinking. *American Journal of Public Health*, 85, 921–926.
- Weiss, L. H., & Schwarz, J. C. (1996). The relationship between parenting types and older adolescents' personality, academic achievement, adjustment, and substance use. *Child Development*, 67, 2101-2114.
- Wetherill, R. R., & Fromme, K. (2007). Perceived awareness and caring influences alcohol use by high school and college students. *Psychology of Addictive Behaviors*, 21, 147–154.
- Wetherill, R. R., Neal, D. J., & Fromme, K. (2008). Perceived awareness and caring influences sexual behavior during the transition from high school to college.

 Manuscript submitted for publication.
- Whitbeck, L. B., Hoyt, D. R., Miller, M., & Kao, M. Y. (1992). Parental support, depressed affect and sexual experience among adolescents. *Youth and Society*, 24(2), 166-177.

- Wong, S. W. & Hughes, J. N. (2006). Ethnicity and language contributions to dimensions
- Zahn-Waxler, C., & Smith, K. D. (1992). The development of prosocial behavior. In V.
 - B. Van Hasselt and M. Hersen (Ed.), *Handbook of social development: A lifespan perspective* (pp. 229-256). New York, NY: Plenum Press.
- Zick, C. D., Bryant, W. K., & Osterbacka, E. (2001). Mothers' employment, parental involvement, and the implications for intermediate child outcomes. *Social Science Research*, *30*, 25-49.

APPENDIX A

Institutional Review Board Research Approval Form



Office of Human Subjects Research 307 Samford Hall Auburn University, AL 36849

Telephone: 334-844-5966 Fax: 334-844-4391 hsubjec@auburn.edu

October 8, 2007

MEMORANDUM TO:

Education Foundation Leadership Technology

PROTOCOL TITLE:

"A Study of the Relationship between Parental Involvement and Student Mental Health"

IRB AUTHORIZATION NO:

07-201 EP 0710

APPROVAL DATE: **EXPIRATION DATE:** October 5, 2007 October 4, 2008

The above referenced protocol was approved by IRB Expedited procedure under 45 CFR 46.110 (#7):

"Research on individual or group characteristics or behavior (including but not limited to, research of perception, cognition, motivation, identity, language, communication, cultural beliefs or practices, and social behavior) or research employing survey, interview, oral history, focus group, program evaluation, human factors evaluation, or quelity assurance methodologies? or quality assurance methodologies.

You should report to the IRB any proposed changes in the protocol or procedures and any unanticipated problems involving risk to subjects or others. Please reference the above authorization number in any future correspondence regarding this project.

If you will be unable to file a Final Report on your project before October 4, 2008, you must submit a request for an extension of approval to the IRB no later than September 20, 2008. If your IRB authorization expires and/or you have not received written notice that a request for an extension has been approved prior to October 4, 2008, you must suspend the project immediately and contact the Office of Human Subjects Research for received written approved prior to October 4, 2008, you must suspend the project immediately and contact the Office of Human Subjects Research for assistance.

A Final Report will be required to close your IRB project file. You are reminded that you must use the stamped, IRB-approved parental consent/assent and information sheet (enclosed) when you consent your participants. Please remember that you must keep signed informed consents for three years after your study is completed.

If you have any questions concerning this Board action, please contact the Office of Human Subjects Research at 844-5966.

Sincerely.

Peter W. Grandjean, Chair Institutional Review Board for the Use of Human

Subjects in Research

Enclosures

cc: Dr. Jose Llanes Dr. James E. Witte

APPENDIX B

Email to Instructors to Allow Researcher to Visit and Administer Surveys

Dear [Instructor],

I am in the process of gathering my data for my dissertation and have received permission from Kathryn to invite the UNIV students to take my survey. I was wondering if I could come to your class next semester and invite your students to participate? I could either leave the surveys for them and come pick them up the following week or I would also love to teach the lesson for that day. My topic is the impact of parental involvement on students' college experience and it fits in nicely with our mission to help students with their transition to college. I have a presentation that I can do with them about transforming your relationship with your parents. Let me know if either of these would fit into your schedule and not disrupt your class. Just wanted to check now while we are all in the syllabus writing process.

Thank you so much and have a wonderful break! Ruthie

APPENDIX C

Administrator Script

Survey Administrator Script

I will begin by thanking the students and instructor for allowing me to come speak to them. Next, I will read the following script to invite them to participate. After questions are answered, students will receive a packet including an information letter, parental consent form, survey instruments, and list of mental health resources. Finally, to contribute to the goals of University courses to educate students about healthy relationships and student development, I will lead a brief activity and discussion about development and the impact students' parents have.

- I would like to invite you to complete a survey being conducted by myself, an Academic Counselor and graduate student of the Education Leadership program here at Auburn University.
- The survey is being conducted to investigate the relationship between parental involvement and student mental health
- The survey will take approximately ten minutes to complete and does not ask for any identifying information.
- It does not ask for any identifying information, and therefore, you will remain anonymous.
- Please read the information letter attached to your survey before completing it. This letter is yours to keep.
- If you are under the age of nineteen you must also have parental consent to participate. There is a parental consent form in your packet if you need one. You need to have your parent sign it and then you may turn in your survey via campus mail in the envelope included in your packet.
- If you would like to participate you may complete the survey at this time and submit it in the collection box, or you may take it with you to complete at a later time. If you choose to take it with you, surveys may be submitted via campus mail in the provided envelope.
- If you choose to participate, it is our hope that this survey will allow you to consider the impact that your parents have on your college experience and reflect on your current mental health status. If this experience raises concerns or questions about your or a friend's mental health, you are encouraged to contact one of the resources listed on the mental health resource list in your packet. Please feel free to contact me if I can answer any questions for you about these or other resources.
- Information collected will be published in my dissertation and my be published in a journal and/or presented at a professional meeting.
- Thank you for considering participating in this research. Please feel free to ask any questions at this time or to direct any questions to the email address listed on the survey. Also, please remember to read the Information Letter/Parental Consent Letter attached to your survey before completing the survey.

APPENDIX D

Alternative Questions

Questions to Think About...

Take a few moments to reflect on the following questions relating to the presentation. It will prepare you to apply the presentation to your own experiences and relationships. You may make notes if you would like. This handout will not be collected.

- How would you describe your experience of transitioning to college life?
- How have you changed since you started college?
- Are you happy with these changes?
- What role did your parents play in your decision to come to college?
- How often do you talk to your parents?
- How many times have you been home since you started college?
- How has your relationship with your parents changed since you started college?

APPENDIX E

Information Letter

Auburn University

Auburn University, Alabama 36849-5221

The Auburn University
Institutional Review Board
has approved this document for use
from 10/5/07 to 10/4/08
Protocol # 07-Z01 EP 0710

FAX: (334) 844-3072

Educational Foundations Leadership and Technology 4036 Haley Center

INFORMATION LETTER
for a Research Study Entitled
A Study of the Relationship between Parental
Involvement and Student Mental Health

NOTE: DO NOT AGREE TO PARTICIPATE UNLESS AN IRB APPROVAL STAMP WITH CURRENT DATES HAS BEEN APPLIED TO THIS DOCUMENT.

Telephone: (334) 844-4460

You are invited to participate in a research study being conducted to explore the relationship between parental involvement and student mental health. This study is being conducted by Ruthanna Payne under the supervision of Dr. James E. Witte, Associate Professor in the Auburn University Department of Education, Foundations, Leadership, and Technology. I hope to learn how parents' involvement in their student's college experience impacts student mental health. You were selected as a possible participant because you are taking a UNIV1000 or UNIV1050 course, and are 19 years of age or older.

If you decide to participate, we will collect your anonymous survey and compile the information with the other surveys to determine if a connection exists between parental involvement and student mental health. The survey will take approximately ten minutes to complete. You have the option to complete the survey and submit it to the collection box at this time or take it with you to complete at a later time. If you choose to complete it later, you have been provided with a pre-addressed envelope to mail in your surveys via campus mail.

No identifying information is collected on the survey, and therefore your responses will remain anonymous. If you choose to not finish the survey you may simply discard it.

It is our hope that this survey along with the class presentation will allow you to reflect on your parent(s)'/guardian(s)' involvement in your college experience and on your mental health status at this stage in your life. If this experience leads to any concerns or questions about your personal mental health, you are encouraged to contact one of the resources listed in the referral list you received with this survey. Please feel free to contact me for additional information about resources at the phone number or email listed below.

Information collected through your participation will be published in my dissertation and possibly published in a journal and/or presented at a professional meeting.

Your decision whether or not to participate and whether or not to withdraw your survey will not jeopardize your future relations with Auburn University, your UNIV1000/UNIV1050 instructor, or the EFLT department.

If you have any questions I invite you to ask them now. If you have questions later I will be happy to answer them: 844-3468, blakerl@auburn.edu.

For more information regarding your rights as a research participant you may contact the Auburn University Office of Human Subjects Research or the Institutional Review Board by phone (334)-844-5966 or e-mail at hsubjec@auburn.edu or IRBChair@auburn.edu .

HAVING READ THE INFORMATION PROVIDED, YOU MUST DECIDE WHETHER TO PARTICIPATE IN THIS RESEARCH PROJECT. IF YOU DECIDE TO PARTICIPATE, THE DATA YOU PROVIDE WILL SERVE AS YOUR AGREEMENT TO DO SO. THIS LETTER IS YOURS TO KEEP.

Durham Pang 9.24.07
Investigator's signature Date

Print Name

Name

Page 1 of 1

APPENDIX F

Parental Permission/Child Assent Form

Auburn University

Auburn University, Alabama 36849-5221

The Auburn University
Institutional Review Board
has approved this document for use
from 10/5/07 to 10/4/08
Protocol # 11-201 E P 0710

Educational Foundations Leadership and Technology 4036 Haley Center Telephone: (334) 844-4460 FAX: (334) 844-3072

Parental Permission/Child Assent for a Research Study Entitled A Study of the Relationship between Parental Involvement and Student Mental Health NOTE: DO NOT AGREE TO PARTICIPATE UNLESS AN IRB APPROVAL STAMP WITH CURRENT DATES HAS BEEN APPLIED TO THIS DOCUMENT.

Your child has been invited to participate in a research study being conducted to explore the relationship between parental involvement and student mental health. This study is being conducted by Ruthanna Payne under the supervision of Dr. James E. Witte, Associate Professor in the Auburn University Department of Education, Foundations, Leadership, and Technology. I hope to learn how parents' involvement in their student's college experience impacts student mental health. Your child was selected as a possible participant because he or she is taking a UNIV1000 or UNIV1050 course. Since your child is age 18 or younger we must have your permissions to include him/her in the study.

If you decide to allow your child to participate, I will collect his/her anonymous survey and compile the information with the other surveys to determine if a connection exists between parental involvement and student mental health. Your child's time commitment will be approximately ten minutes. Your child may submit the surveys if he or she chooses by mailing through campus mail via the provided envelope.

No identifying information is collected on the survey, and therefore responses will remain anonymous. If your child chooses to not finish the survey he or she may simply discard it.

It is my hope that this survey along with the class presentation will allow your child to reflect on his/her parents' involvement and his/her mental health status at this stage in life. If this experience leads to any concerns or questions about personal mental health, your child is encouraged to contact one of the resources listed in the referral list received with this survey. Please feel free to contact me for additional information about resources at the phone number or email listed below.

Information collected through your child's participation will be published in my dissertation and possibly published in a journal and/or presented at a professional meeting.

Your decision to allow your child to participate and whether or not to withdraw his/her survey will not jeopardize future relations with Auburn University, his/her UNIV1000/UNIV1050 instructor, or the EFLT department. If you or your child have any questions I invite you to contact me at 844-3468, blakerl@auburn.edu.

For more information regarding your child's rights as a research participant you may contact the Auburn University Office of Human Subjects Research or the Institutional Review Board by phone (334)-844-5966 or e-mail at hsubjec@auburn.edu or IRBChair@auburn.edu .

| Parent/Guardian Initials | |
|--------------------------|--|
| Participant Initials | |

Page 1 of 2

HAVING READ THE INFORMATION PROVIDED, YOU MUST DECIDE WHETHER OR NOT YOU WISH FOR YOUR CHILD TO PARTICIPATE IN THIS RESEARCH STUDY. YOUR SIGNATURE INDICATES YOUR WILLINGNESS TO ALLOW YOUR CHILD TO PARTICIPATE. YOUR CHILD'S SIGNATURE INDICATES HIS/HER WILLINGNESS TO PARTICIPATE.

| | | Ruena Pay | 9.24.07 |
|----------------------------------|------|--------------------------------|---------|
| Participant's signature | Date | Investigator obtaining consent | Date |
| | | Ruthanna Payne | |
| Print Name | | Print Name | |
| Parent's or Guardian Signature | Date | | |
| raicines or dual diant signature | Date | | |
| | | | |
| Print Name | - | | |

The Auburn University
Institutional Review Board
has approved this document for use
from 10/5/07 to 16/4/08
Protocol # 07-20/ EP 07/10

APPENDIX G

Survey of Parental Involvement and Student Well Being

SURVEY OF PARENTAL INVOLVEMENT & STUDENT WELL BEING

I. Demographics (Circle One) Year in School: Freshman Sophomore Junior Senior Gender: Male Female Alaskan Native Asian/Pacific Islander Race: African American Caucasian Hispanic Native American Other Distance from Home: 0-199 miles 200-399 miles 400-599 miles 600 miles or more Home Town (City & State): Are you the first one in your immediate family to attend college? Yes No The following questions are to be answered on a 4 point scale from Strongly Disagree to Strongly Agree. The choices will be Strongly Disagree-SD, Disagree-D, Agree-A, Strongly Agree-SA, and Not Applicable-NA. SD=1, D=2, A=3, SA=4. II. College Choice SD D SA NA A Parents helped fill out applications SD D SA NA Parents wrote college essays SD Α sANA Parents helped in college decision process SD D SA NA Α SD Parents reminded you of application deadlines D Α SANA Parents pressured you to attend the same college theydid. SD D Α SA NA Parents pressured you to attend college SD D Α SANA 7. Parent involvement in college choice was positive SD D Α SA SD 8. Parents helped college choice process Α SA NA SD D NA III. Social Involvement A SA 9. Parents encourage you to be involved on campus D NA SDΑ SAParents encouraged you to join a fraternity so rority SD D Α SA NA Parents spoke to you about drinking SD D Α SA NA SD D Parents spoke to you about drugs Α SANA SD D Α Parents spoke to you about social pressures you would face. SANA SD D SA Parents encouraged you to live on campus NA SD D Parents made you live on campus Α SA NA Parents helped transition from high social activities to college SD D SANA D Parent involvement in college social life helped transition SD Α SA NA 18. Parent involvement in college social life hindered transition SD SA NA Α 19. Parent involvement in college social life was positive SD Α sANA IV. Academic Involvement NA SD A SA

SD

SD D

D

Α

Α

Parents involved in scheduling your classes

21. Parents called to wake you up for class

NA

NA

SA

SA

| 22. Parents called to remind you of assignments 23. Parents knew when tests were 24. Parents attended freshman orientation 25. Parents ask what your grades are 26. Parents helped transition you from high course load to college 27. Parent involvement in academics was a hindrance 28. Parent involvement in academics was positive 29. Parent involvement in academics improved your GPA 30. I am satisfied with the overall level of my parent's involvement in my college life | SD SD SD SD | D D D D D D D D | A A A A A | SA SA SA SA | |
|---|----------------------|-----------------|-----------------------|----------------------|----|
| What would you change about your parents' involvement in your | college | life?: | | | |
| | | | | | |
| 31. I am satisfied with the institution I attend | SD | D | A | SA | NA |
| 32. How often do you communicate with your parent(s)? Via email per week Via instant messenger per week Via phone per week Via mail per week 33. Who initiated the majority of this contact? Parent St 34. How often do you see you parent(s) per semester? You travel home per semester Parents visit campus per semester In other locations per semester | udent | | | | |
| 35. Who made trips more often? Parent Student Describe your relationship with your parents: | | | | | |
| | | | | | |
| How has your relationship with your parents impacted your colleg | e expe | rience' | ? | | |
| | | | | | |

During the past 30 days, how much of the time did you feel...

| | ALL THE TIME | MOSTOF THE TIME | SOME OF THE TIME | A LITTLE OF THE TIME | NONE OF THE TIME |
|---------------------|-----------------|-----------------------|------------------------|-------------------------------|------------------------|
| acheerful? | A11 | Most | Some | Little | None |
| b in good spirits? | A11 | Most | Some | Little | None |
| cextremely happy? | A11 | Most | Some | Little | None |
| dcalm and peaceful? | A11 | Most | Some | Little | None |
| esatisfied? | A11 | Most | Some | Little | None |
| f full of life? | A11 | Most | Some | Little | None |

Using a scale from 0 to 10 where 0 means "the worst possible life overall" and 10 means "the best possible life overall," how would you rate your life overall these days?

WORST BEST 0 1 2 3 4 5 6 7 8 9 10

Please indicate how strongly you agree or disagree with each of the following statements.

| | AGREE | | | | DISAGREE | | |
|--|----------|--------------|-------------|---------------|-------------|--------------|----------|
| | STRONGLY | SOME WHAT | A LITTLE | DON'T KNOW | A LITTLE | SOME WHAT | STRONGLY |
| I like most parts of my personality | SA | A | AL | N | DL | D | SD |
| When I look at the story of my e, I am pleased with how things ve turned out so far. | SA | A | AL | N | DL | D | SD |
| Some people wander aimlessly rough life, but I am not one of em | SA | A | AL | N | DL | D | SD |
| The demands of everyday life ten get me down | SA | A | AL | N | DL | D | SD |
| In many ways I feel disappointed out my achievements in life | SA | A | AL | N | DL | D | \$D |

| | | AGREE | | | | DISAGR | EE |
|---|----------|--------------|-------------|---------------|-------------|--------------|----------|
| | STRONGLY | SOME WHAT | A LITTLE | DON'T KNOW | A LITTLE | SOME WHAT | STRONGLY |
| Maintaining close relationships s been difficult and frustrating for | SA | A | AL | N | DL | D | \$D |
| I live life one day at a time and n't really think about the future | SA | A | AL | N | DL | D | SD |
| In general, I feel I am in charge of a situation in which I life | SA | A | AL | N | DL | D | \$D |
| I am good at managing the sponsibilities of daily life | SA | A | AL | N | DL | D | \$D |
| I sometimes feel as if I've done there is to do in life. | SA | A | AL | N | DL | D | \$D |
| . For me, life has been a ntimous process of learning, anging, and growth | SA | A | AL | N | DL | D | \$D |
| I think it is important to have wexperiences that challenge how I nk about myself and the world | SA | A | AL | N | DL | D | \$D |
| . People would describe me as a ving person, willing to share my ne with others. | SA | A | AL | N | DL | D | \$D |
| . I gave up trying to make big provements or changes in my life ong time ago | SA | A | AL | N | DL | D | \$D |
| . I tend to be influenced by people th strong opinions | SA | A | AL | N | DL | D | \$D |
| I have not experienced many irm and trusting relationships with ners | SA | A | AL | N | DL | D | \$D |
| . I have confidence in my own inions, even if they are different om the way most other people think | SA | A | AL | N | DL | D | \$D |
| I judge myself by what I think is portant, not by the values of what ners think is important. | SA | A | AL | N | DL | D | \$D |

Please answer the following questions are about how you have been feeling and how you have been functioning during the past month. Place a check mark in the box that best represents how often you have experienced or felt the following:

| During the past month, how often did you feel | NEVER | ONCE OR TWICE | ABOUT ONCE A WEEK | ABOUT 2 OR 3 TIMES A WEEK | ALMOST EVERY DAY | EVERY DAY |
|---|-------|---------------------|-------------------------|------------------------------------|------------------------|--------------|
| 1. happy | | | | | | |
| 2. interested in life | | | | | | |
| 3. satisfied | | | | | | |
| that you had something important to contribute to society | | | | | | |
| fnat you belonged to a community (like a social group, or your neighborhood) | | | | | | |
| 6. that our society is becoming a better place for people | | | | | | |
| 7. that people are basically good | | | | | | |
| 8. that the way our society works makes sense to you | | | | | | |
| that you liked most parts of your personality | | | | | | |
| good at managing the responsibilities of your daily life | | | | | | |
| that you had warm and trusting relationships with others | | | | | | |
| 12. that you have experiences that challenge you to grow and become a better person | | | | | | |
| 13. confident to think or express your own ideas and opinions | | | | | | |
| 14. that your life has a sense of direction or meaning to it | | | | | | |

Thank you for completing this survey.

APPENDIX H

Referral List of Auburn-Area Mental Health Service Providers

REFERRAL LIST OF AUBURN-AREA MENTAL HEALTH SERVICE PROVIDER

INDIVIDUAL/AGENCY SERVICES AVAILABLE COST/HOUR Crisis Center Phone Counseling (only) (No Charge) (334)821-8600 **Student Counseling Services** Individual & Group Therapy (No Charge) 400 Lem Morrison Dr. Suite 2086 Auburn University, AL 36849 (334) 844-5123 Auburn University Marriage, Family, and \$25-55 **Psychological Services** Individual Therapy Based on income 1122 Haley Center Auburn, AL 36849-5234 (334)844-4889 East Alabama Mental Health Individual & Group Therapy Sliding Scale 2506 Lambert Dr. Based on income Opelika, Al 36801 (334)742-2700 (334)742-2877 Clinical Psychologists Individual & Group Therapy Initial Appt: \$130 248 E. Glenn Ave. Other Appt: \$120 Auburn, AL 36849

As of June 2007

APPENDIX I

Scoring Information for Keyes' Scales of Subjective Well Being

Scoring Instructions for Subjective Well-Being Scales:

By Corey L. M. Keyes, Emory University and the MIDMAC (Successful Midlife Development)
MacArthur Foundation Network.

Emotional well-being:

Positive Affect scale = Reverse code items EWB1a through EWB1f, then sum items.

Psychological well-being (section PWB items):

(Reverse code the following items: 1, 2, 3, 8, 9, 11, 12, 13, 17, 18)

Self-Acceptance scale = sum items 1, 2, 5. Purpose in Life scale = sum items 3, 7, 10. Environmental Mastery scale = sum items 4, 8, 9. Positive Relations with Others scale = sum items 6, 13, 16. Personal Growth scale = sum items 11, 12, 14. Autonomy scale = sum items 15, 17, 18.

Social well-being (section SWB items):

(Reverse code the following items: 3, 4, 5, 6, 11, 12, 14)

Social Coherence scale = sum items 1, 8, 12. Social Integration scale = sum items 2, 6, 11. Social Acceptance scale = sum items 3, 10, 14. Social Contribution scale = sum items 4, 7, 15. Social Actualization scale = sum items 5, 9, 13.

Subjective Well-Being Scale Item Weights based on Confirmatory Factor Analysis based on Keyes' (2002, 2005) Mental Health Continuum Model

Emotional Well-Being (EWB), Psychological Well-Being (PWB), Social Well-Being (SWB)

| TTEN (| XX7 : 1.4 |
|--------|-----------|
| ITEM | Weight |
| EWB1a | .04 |
| EWB1b | .038 |
| EWB1c | .019 |
| EWB1d | .006 |
| EWB1e | .01 |
| EWB1f | .018 |
| EWB2 | .06 |
| PWB1 | .133 |
| PWB2 | .108 |
| PWB3 | .063 |
| PWB4 | .037 |
| PWB5 | .073 |
| PWB6 | .041 |
| PWB7 | 011 |
| PWB8 | .044 |
| PWB9 | .069 |
| PWB10 | .004 |
| PWB11 | .07 |
| PWB12 | .036 |
| PWB13 | .022 |

| Weight |
|--------|
| .01 |
| .004 |
| .047 |
| .024 |
| .023 |
| .019 |
| .014 |
| .003 |
| .045 |
| .008 |
| .024 |
| .014 |
| .011 |
| .012 |
| .011 |
| .018 |
| .003 |
| .018 |
| .024 |
| .058 |
| |

APPENDIX J

Letter of Approval

From: "Oliver, Bryan" < >
To: "Ruthanna Payne" < >

Date: 9/20/2007 10:28:55 AM

Subject: RE: Dissertation

Ruthanna,

Consider this permission to use any of the material related to my dissertation as long as it is properly cited and follows all the rules of plagiarism of course. If you need anything else please let me know.

Bryan

From: "Corey Keyes Dr" < > To: "Ruthanna Payne" < >

Date: 9/20/2007 8:45:24 AM

Subject: RE: Mental Health Continuum

Dear Ruthanna,

I think your proposed research will be very useful for colleges as well as parents. I really am encouraged by the interest in applying this work in the school arena.

I have attached the questionnaire that was used in the MIDUS study and represents what I have been calling the long form for the mental health continuum. Because the response formats on these questions are the usual likert self descriptive (agree/disagree) kind, the long form presents a challenge in terms of deciding cutpoints (I have used tertiles in our MIDUS national data). So, I have moved toward using the attached short form of the mental health continuum. I like the latter because it represents a single item from each dimensions/scale of psychological and social well-being, and uses response formats that are identical to the response format use to diagnose major depressive episode (e.g., has someone had depressed affect "all of the time" or "most of the time" for a period of 2 weeks or more ...).

Both documents include descriptions for the assessment and diagnosis along with some citations. I can also send you the recent Am. Psychologist article I published earlier this year if you haven't read it, as it summarizes the research up to that point.

Best of luck with your research. By the way, you should check out the website for the "Healthy Minds" study, which include my short form for annual assessment of college students' mental health.

Cheers, Corey